Registration No. 333-

SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM S-3

REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

ALCOA INC.

(Exact name of registrant as specified in its charter)

Pennsylvania

(State or other jurisdiction of incorporation or organization)

25-0317820

(I.R.S. Employer Identification Number)

201 Isabella Street Pittsburgh, Pennsylvania 15212-5858 (412) 553-4545

(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)

Lawrence R. Purtell
Executive Vice President and General Counsel
390 Park Avenue
New York, New York 10022-4608
(212) 836-2650

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Approximate date of commencement of	proposed sale to the	nublic: From time to	time after the effective of	ate of this registration statement

	If the only securities being registered on this Form are	e being offered pursuant to divi	idend or interest reinvestment	plans, please ch	eck the following box.
Г					

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, other than securities offered only in connection with dividend or interest reinvestment plans, check the following box: [X]

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. []

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. []

If delivery of the prospectus is expected to be made pursuant to Rule 434, please check the following box. []

CALCULATION OF REGISTRATION FEE

Title of each class of securities to be registered	Amount to be registered (1)	Proposed maximum offering price per share (2)	Proposed maximum aggregate offering price (2)	Amount of registration fee
Alcoa Inc. Common Stock, \$1.00 par value per share	17,773,540 shares	\$26.485	\$470,732,207	\$38,082.24

- (1) Pursuant to Rule 416(a) under the Securities Act of 1933, as amended, this Registration Statement also covers such additional shares as may be offered by reason of stock splits, stock dividends or similar transactions.
- (2) Estimated solely for the purpose of calculating the registration fee in accordance with Rule 457(c) under the Securities Act of 1933, as amended, based on the average of the high and low prices per share of the Registrant's common stock as reported on the New York Stock Exchange on August 7, 2003.

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933, as amended, or until the Registration Statement shall become effective on such date as the Securities and Exchange Commission, acting pursuant to said Section 8(a), may determine.

The information in this prospectus is not complete and may be changed. The selling shareholders may not sell these securities until the registration statement filed with the Securities and Exchange Commission is effective. This prospectus is not an offer to sell these securities and it is not soliciting an offer to buy these securities in any state where the offer or sale is not permitted.

SUBJECT TO COMPLETION DATED AUGUST 13, 2003

Prospectus



This prospectus relates to the offering for resale of up to 17,773,540 shares of common stock, \$1.00 par value per share, of Alcoa Inc. from time to time by the selling shareholders named in this prospectus. The selling shareholders will receive all of the proceeds from any sales. Alcoa will not receive any of the proceeds.

The selling shareholders may sell the shares at various times and in various types of transactions, including sales in the open market, sales in negotiated transactions and sales by a combination of these methods. Shares may be sold at prevailing market prices, at prices related to prevailing market prices, at negotiated prices or at fixed prices. See "Selling Shareholders" and "Plan of Distribution."

Alcoa's common stock is listed on the New York Stock Exchange under the symbol "AA." On August 12, 2003, the closing price of our common stock on the New York Stock Exchange was \$27.10 per share.

The mailing address of our principal executive offices is 201 Isabella Street, Pittsburgh, Pennsylvania 15212-5858. The telephone number is (412) 553-4545.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities or determined if this prospectus is truthful or complete. Any representation to the contrary is a criminal offense.

The date of this Prospectus is

, 2003.

TABLE OF CONTENTS

	Page
Alcoa Inc.	3
Where You Can Find More Information	5
Incorporation of Certain Documents by Reference	5
Forward-Looking Statements	6
<u>Use of Proceeds</u>	8
Description of Common Stock	8
Selling Shareholders	11
Plan of Distribution	13
<u>Legal Matters</u>	18
Independent Accountants	18

ALCOA INC.

Formed in 1888 under the laws of the Commonwealth of Pennsylvania, Alcoa has its principal office in Pittsburgh, Pennsylvania. As used in this prospectus, "Alcoa" means Alcoa and its consolidated subsidiaries unless the context otherwise requires.

Alcoa is the world's leading producer of primary aluminum, fabricated aluminum and alumina, and is active in all major aspects of the industry: technology, mining, refining, smelting, fabricating and recycling. Alcoa serves the aerospace, automotive, packaging, building and construction, commercial transportation and industrial markets. Aluminum and alumina represent approximately two-thirds of Alcoa's revenues, and the price of aluminum influences the operating results of Alcoa. Nonaluminum products include precision castings, industrial fasteners, vinyl siding, consumer products, food service and flexible packaging products, plastic closures, fiber optic cables, and electrical distribution systems for cars and trucks.

Alcoa's operations consist of five worldwide segments: Alumina and Chemicals, Primary Metals, Flat-Rolled Products, Engineered Products and Packaging and Consumer. Alcoa businesses that are not reported to management as part of one of these five segments are aggregated and reported as "Other."

Alumina and Chemicals—This segment consists of Alcoa's worldwide alumina and chemicals system that includes the mining of bauxite, which is then refined into alumina. Alumina is sold directly to internal and external smelter customers worldwide or is processed into industrial chemical products.

Primary Metals—This segment consists of Alcoa's worldwide smelter system. Primary Metals receives alumina primarily from the Alumina and Chemicals segment and produces aluminum ingot to be used by Alcoa's fabricating businesses, as well as sold to external customers, aluminum traders, and commodity markets. Results from the sale of aluminum powder, scrap and excess power are also included in this segment, as well as the results from aluminum derivative contracts.

Flat-Rolled Products—This segment's principal business is the production and sale of aluminum plate, sheet and foil. This segment includes rigid container sheet, which is used to produce aluminum beverage cans, and sheet and plate used in the transportation and distributor markets.

Engineered Products—This segment includes hard- and soft-alloy extrusions, including architectural extrusions, super-alloy castings, steel and aluminum fasteners, aluminum forgings, and wheels. These products serve the transportation, building and construction, and distributor markets.

Packaging and Consumer—This segment includes consumer products, foodservice, flexible packaging, and packaging graphics design, as well as closures and packaging machinery. The principal products in this segment include aluminum foil; plastic wraps and bags; metal and plastic beverage and food closures; flexible packaging; prepress services; and thermoformed plastic containers and extruded plastic sheet and film. Consumer products are marketed under brands including Reynolds Wrap®, Diamond®, Baco®, and Cut-Rite® wax paper.

Other—This group includes other Alcoa businesses that are not included in the segments previously mentioned. This group includes Alcoa Fujikura Ltd., which produces electrical components for the automotive industry and fiber-optic cable and provides services for the telecommunications industry; the residential building products operations, Alcoa Home Exteriors (formerly Alcoa Building Products); automotive parts businesses; Thiokol, a producer of solid rocket propulsion systems (Thiokol was sold in April 2001.); and Reynolds' metal distribution business, RASCO. (In November 2001, the net assets of RASCO were contributed to a joint venture, Integris Metals, Inc., in which Alcoa retains a 50% equity interest.)

Alcoa is a global company operating in 40 countries. The principal executive offices of Alcoa are located at 201 Isabella Street, Pittsburgh, Pennsylvania 15212-5858 (telephone: (412) 553-4545).

Recent Developments

On August 1, 2003, Alcoa completed the acquisition of the approximately 41% shareholding in Alcoa Alumínio S.A. ("Alcoa Alumínio") and Alcoa Latin American Holdings Corporation ("ALAHC") held by affiliates of Camargo Corrêa S.A. ("Camargo Corrêa"), a leading contractor and industrial conglomerate in Brazil (Camargo Corrêa and its affiliates collectively, the "Camargo Group"). Before the acquisition, Alcoa had owned approximately 59% of these South American operations. Alcoa and a wholly-owned subsidiary exchanged 17,773,540 shares of Alcoa common stock for all of the shares of common stock in Alcoa Alumínio and ALAHC held by the Camargo Group (the "Share Exchange"). Additional cash consideration tied to future performance of Alcoa Alumínio and ALAHC over the next five years may be payable by Alcoa under earn-out and guaranteed minimum payment formulas, but such additional consideration will be limited by notional dividends on and appreciation in the market price of Alcoa's common stock during the same period. In connection with the Share Exchange, Alcoa agreed to register the resale of the 17,773,540 shares of Alcoa common stock by the selling shareholders. See "Selling Shareholders" and "Plan of Distribution."

WHERE YOU CAN FIND MORE INFORMATION

Alcoa files annual, quarterly and special reports, proxy statements and other information with the Securities and Exchange Commission. Its SEC filings are available to the public over the Internet at the SEC's web site at http://www.sec.gov. You may also read and copy any document Alcoa files with the SEC at the SEC's public reference room at 450 Fifth Street, N.W., Room 1024, Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information on the public reference room. You may also inspect and copy reports and other information about Alcoa at the offices of the New York Stock Exchange, 20 Broad Street, New York, New York 10005.

Alcoa has filed a registration statement on Form S-3 with the SEC under the Securities Act of 1933, as amended (the "Securities Act"), that registers the resale by the selling shareholders of the shares offered by this prospectus. As permitted by the rules and regulations of the SEC, this prospectus does not contain all of the information set forth in the registration statement. You should read the registration statement for further information about Alcoa and its common stock.

INCORPORATION OF CERTAIN DOCUMENTS BY REFERENCE

The SEC allows Alcoa to "incorporate by reference" in this prospectus the information in the documents that it files with the SEC, which means that Alcoa can disclose important information to you by referring you to those documents. The information incorporated by reference is considered to be a part of this prospectus, and information in documents that Alcoa files later with the SEC will automatically update and supersede information contained in documents filed earlier with the SEC or contained in this prospectus. Alcoa incorporates by reference in this prospectus the documents listed below and any future filings that it may make with the SEC under Sections 13(a), 13(c), 14 or 15(d) of the Securities Exchange Act of 1934 (the "Exchange Act") until the completion of the offering of the securities described in this prospectus, except that Alcoa is not incorporating by reference any information that is not deemed to be filed under those sections.

- Annual Report on Form 10-K for the fiscal year ended December 31, 2002;
- Quarterly Reports on Form 10-Q for the fiscal quarters ended March 31, 2003 and June 30, 2003; and
- Current Reports on Form 8-K filed on May 6, 2003 and August 1, 2003.

Alcoa will furnish without charge to you, upon written or oral request, a copy of any or all of the documents described above, except for exhibits to those documents, unless the exhibits are specifically incorporated by reference into those documents. Requests for copies should be addressed to:

Alcoa Inc. Attention: Investor Relations 390 Park Avenue New York, New York 10022-4608 Telephone: (212) 836-2674

You should rely only on the information contained or incorporated by reference in this prospectus and in any prospectus supplement. Alcoa has not authorized anyone to provide you with different or additional information. If anyone provides you with different or inconsistent information, you should not rely on it. The shares of common stock are not being offered in any jurisdiction where the offer or sale is not permitted. This Prospectus is dated , 2003. You should not assume that the information in this prospectus is accurate as of any date other than that date. Alcoa's business, financial condition, results of operations and prospects may have changed since that date.

FORWARD-LOOKING STATEMENTS

This prospectus and the documents incorporated by reference in this prospectus include certain statements that may be deemed to be "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. Forward-looking statements may be identified by their use of words like "anticipates," "believes," "estimates," "expects," "hopes," "should," "targets," "will," "will likely," "forecast," "outlook," "projects" and similar expressions, as they relate to Alcoa or the management of Alcoa. All statements that address expectations or projections about the future, including statements about Alcoa's strategy for growth, cost reduction goals, expenditures and financial results, are forward-looking statements. Such statements reflect the views of Alcoa with respect to future events, the outcome of which is subject to certain risks and uncertainties. Should one or more of these risks or uncertainties materialize, or should underlying assumptions prove incorrect, actual results, performance or achievements of Alcoa may vary materially from those expressed or implied in the forward-looking statements. Alcoa disclaims any intention or obligation (other than as required by law) to update or revise any forward-looking statements.

In addition to the factors discussed in Management's Discussion and Analysis of Financial Condition and Results of Operations in Alcoa's Form 10-K for the fiscal year ended December 31, 2002 and its Form 10-Q's for the fiscal quarters ended March 31, 2003 and June 30, 2003, the following are some of the important factors that could cause Alcoa's actual results, performance or achievements to differ materially from those projected in any forward-looking statements:

- Alcoa is a leading global producer of alumina, aluminum ingot and aluminum fabricated products. The aluminum industry is highly cyclical, with
 prices subject to worldwide market forces of supply and demand and other influences. Prices can be volatile. Although Alcoa uses contractual
 arrangements with customers, as well as forward, futures and options contracts, to manage its exposure to the volatility of London Metal Exchangebased prices, and is product and segment diversified, Alcoa's results of operations could be affected by material adverse changes in economic or
 aluminum industry conditions generally or in the markets served by Alcoa, including the transportation, building, construction, distribution, packaging,
 industrial gas turbine, telecommunications and other markets.
- Alcoa consumes substantial amounts of energy in its operations. Although Alcoa generally expects to meet the energy requirements for its alumina refineries and primary aluminum smelters from internal sources or from long-term contracts, the following could affect Alcoa's results of operations:
 - o significant increases in electricity costs rendering smelter operations uneconomic;
 - o the unavailability of electrical power due to droughts;
 - o interruptions in energy supply due to equipment failure or other causes; or
 - o the inability to extend contracts upon expiration on economical terms.
- Alcoa's ability to grow earnings will be affected by increases in the cost of raw materials, including caustic soda, calcined petroleum coke and resins, in addition to energy. Alcoa may not be able to offset fully the effects of higher raw material costs through price increases or productivity improvements.
- As part of its strategy for growth, Alcoa has made and may continue to make acquisitions and divestitures and form strategic alliances. There can be no assurance that these will be completed or beneficial to Alcoa.
- Alcoa has investments and activities in numerous countries outside the U.S. and in emerging markets, including China, Brazil, India, Korea and
 Mexico. Changes in the laws or governmental policies in the countries in which Alcoa operates could affect its business in such countries and Alcoa's
 results of operations. In addition, economic factors, including inflation and fluctuations in foreign currency exchange rates and interest rates, and
 competitive factors in the countries could affect Alcoa's revenues, expenses and results of operations.
- The markets for most aluminum products are highly competitive. In addition, aluminum competes with other materials, such as steel, plastics and glass, among others, for various applications in Alcoa's key markets. The willingness of customers to accept substitutions for the products sold by Alcoa, the ability of

large customers to exert leverage in the marketplace to affect the pricing for fabricated aluminum products or other developments by or affecting Alcoa's competitors or customers could affect Alcoa's results of operations.

- A significant downturn in the business or financial condition of a key customer or customers supplied by Alcoa could affect Alcoa's results of
 operations in a particular period.
- Alcoa has undertaken and may continue to undertake productivity and cost-reduction initiatives to improve performance, including deployment of
 company-wide business process models, such as the Alcoa Business System and the Alcoa Enterprise Business Solution, an initiative designed to build
 a common global infrastructure across Alcoa for data, processes and supporting software. There can be no assurance that these initiatives will be
 completed or beneficial to Alcoa or that any estimated cost savings from such activities will be realized.
- Alcoa is working on new developments in advanced smelting process technologies, including inert anode technology. There can be no assurance that such technologies will be commercially feasible or beneficial to Alcoa.
- Alcoa's operations worldwide are subject to numerous complex and increasingly stringent environmental laws and regulations. The costs of complying
 with such environmental laws and regulations, including participation in assessments and cleanups of sites, as well as internal voluntary programs, are
 significant and will continue to be so for the foreseeable future. Alcoa's results of operations or liquidity in a particular period could be affected by
 certain environmental matters, including remediation costs and damages related to several sites.
- Alcoa's results of operations or liquidity in a particular period could be affected by significant legal proceedings or investigations adverse to Alcoa, including product liability, safety and health and other claims.
- Alcoa's estimates of liabilities and expenses for pensions and other postretirement benefits incorporate significant assumptions including the rate used
 to discount the future estimated liability, the long-term rate of return on plan assets and several assumptions relating to the employee workforce (salary
 increases, medical costs, retirement age and mortality). Alcoa's results of operations, liquidity or shareholders' equity in a particular period could be
 affected by a decline in the rate of return on plan assets, the rate used to discount the future estimated liability or changes in employee workforce
 assumptions.
- Alcoa has projected a likely range of proceeds from the divestiture of the businesses and operations announced in the fourth quarter of 2002. There can be no assurance that Alcoa will realize the projected amount of proceeds from such divestitures.
- War or terrorist activities may increase the cost of doing business or otherwise impact Alcoa's financial performance.

The above list of important factors is not all-inclusive or necessarily in order of importance.

USE OF PROCEEDS

All shares of common stock offered by this prospectus will be sold by the selling shareholders. Alcoa will not receive any of the proceeds from such sales.

DESCRIPTION OF COMMON STOCK

Alcoa's common stock is registered under the Exchange Act and listed on the New York Stock Exchange. The following description of the common stock contains summaries of, and is subject to the detailed provisions of, Alcoa's Articles of Incorporation and By-Laws and to the relevant provisions of the Pennsylvania Business Corporation Law ("PBCL").

Alcoa is authorized to issue 1,800,000,000 shares of common stock, par value \$1.00 per share. As of August 4, 2003, there were 864,339,241 shares of Alcoa common stock outstanding. In addition, as of the same date, there were 60,235,297 shares of Alcoa common stock issued and held in Alcoa's treasury, and 105,469,823 shares of Alcoa common stock reserved for issuance under various incentive plans.

Dividend Rights

Holders of Alcoa common stock will receive dividends when and as declared by Alcoa's board of directors. However, no dividend will be declared or paid on the common stock if any Alcoa preferred stock is outstanding, unless all dividends accrued on all classes of Alcoa preferred stock and the current quarter yearly dividend on Alcoa's \$3.75 Cumulative Serial Preferred Stock have been paid or declared and a sum sufficient for payment has been set apart.

Voting Rights

Holders of Alcoa common stock have one vote per share.

Liquidation Rights

Upon any liquidation, dissolution or winding up of Alcoa, whether voluntary or involuntary, after payments to holders of Alcoa preferred stock in an amount determined by the board of directors of Alcoa, plus any accrued dividends, Alcoa's remaining assets will belong to and will be divided among the holders of Alcoa common stock. Under Alcoa's Articles of Incorporation, neither the consolidation or merger of Alcoa with or into one or more corporations or any share exchange or division involving Alcoa under applicable law will be deemed a liquidation, dissolution or winding up of Alcoa.

Preemptive or Other Subscription Rights

Holders of Alcoa common stock have no preemptive right to subscribe for any securities of Alcoa.

Conversion and Other Rights

No conversion, redemption or sinking fund provisions apply to Alcoa common stock, and Alcoa common stock is not liable to further call or assessment by Alcoa. All issued and outstanding shares of Alcoa common stock are fully paid and non-assessable.

Other Matters

Alcoa's Articles of Incorporation provide for the following:

- a classified board of directors with staggered three-year terms;
- · special shareholder voting requirements to remove directors; and

• certain procedures relating to the nomination of directors, filling of vacancies and the vote required to amend or repeal any of these provisions.

Alcoa's Articles of Incorporation also prohibit Alcoa's payment of "green-mail," that is, payment of a premium in purchasing shares of its common stock from a present or recent holder of 5% or more of the common stock, except with the approval of a majority of the disinterested shareholders. This provision and the classified board provision may be amended or repealed only with the affirmative vote of at least 80% of the common stock. In addition, the Articles of Incorporation limit or eliminate to the fullest extent permitted by Pennsylvania law, as from time to time in effect, the personal liability of Alcoa's directors for monetary damages, and authorize Alcoa, except as prohibited by law, to indemnify directors, officers, employees and others against liabilities and expenses incurred by them in connection with the performance of their duties to Alcoa. The classified board article provision and the anti-"greenmail" provision may have certain anti-takeover effects.

Alcoa is governed by certain "anti-takeover" provisions in the PBCL. Chapter 25 of the PBCL contains several anti-takeover provisions that apply to registered corporations such as Alcoa. Section 2538 of the PBCL requires shareholder approval for certain transactions between a registered corporation and an interested shareholder (generally, a shareholder who owns 20% of the stock entitled to vote in an election of directors). Section 2538 applies if an interested shareholder (together with anyone acting jointly with such shareholder and any affiliates of such shareholder):

- is to be a party to a merger or consolidation, a share exchange or certain sales of assets involving such corporation or one of its subsidiaries;
- is to receive a disproportionate amount of any of the securities of any corporation which survives or results from a division of the corporation;
- is to be treated differently from others holding shares of the same class in a voluntary dissolution of such corporation; or
- is to have his or her percentage of voting or economic share interest in such corporation materially increased relative to substantially all other shareholders in a reclassification.

In such a case, the proposed transaction must be approved by the affirmative vote of the holders of shares representing at least a majority of the votes that all shareholders are entitled to cast with respect to such transaction. Shares held by the interested shareholder are not included in calculating the number of shares entitled to be cast, and the interested shareholder is not entitled to vote on the transaction. This special voting requirement does not apply if the proposed transaction has been approved in a prescribed manner by the corporation's board of directors or if certain other conditions, including the amount of consideration to be paid to certain shareholders, are satisfied or the transaction involves certain subsidiaries.

Section 2555 of the PBCL may also apply to a transaction between a registered corporation and an interested shareholder, even if Section 2538 also applies. Section 2555 prohibits a corporation from engaging in a business combination with an interested shareholder unless one of the following conditions is met:

- the board of directors has previously approved either the proposed transaction or the interested shareholder's acquisition of shares;
- the interested shareholder owns at least 80% of the stock entitled to vote in an election of directors and, no earlier than three months after the interested shareholder reaches the 80% level,
 - o the majority of the remaining shareholders approve the proposed transaction;
 - shareholders receive a minimum "fair price" for their shares in the transaction; and
 - the other conditions of Section 2556 of the PBCL are met;

- · holders of all outstanding common stock approve the transaction;
- no earlier than 5 years after the interested shareholder acquired the 20%, a majority of the remaining shares entitled to vote in an election of directors approve the transaction; or
- no earlier than 5 years after the interested shareholder acquired the 20%, a majority of all the shares approve the transaction, all shareholders receive a minimum fair price for their shares, and certain other conditions are met.

Under the PBCL, a person or group of persons acting in concert who hold 20% of the shares of a registered corporation entitled to vote in the election of directors constitutes a control group. On the occurrence of the transaction that makes the group a control group, any other shareholder of the registered corporation who objects can, under procedures set forth under the PBCL, require the control group to purchase his or her shares at "fair value," as defined in the PBCL.

The PBCL also contains certain provisions applicable to a registered corporation such as Alcoa which, under certain circumstances, permit a corporation to:

- redeem "control shares," as defined in the PBCL;
- · remove the voting rights of control shares; and
- require the disgorgement of profits by a "controlling person," as defined in the PBCL.

The transfer agent and registrar for Alcoa common stock is Equiserve Trust Company, N.A.

SELLING SHAREHOLDERS

The selling shareholders acquired the shares of Alcoa common stock offered by this prospectus in a private placement pursuant to the Share Exchange under a principal agreement dated as of July 25, 2003 among Alcoa, Camargo Corrêa and others, filed as an exhibit to the registration statement of which this prospectus is a part. See "Alcoa Inc.—Recent Developments." The selling shareholders, Allpar Limited ("Allpar") and Trelawney Inc. ("Trelawney"), are whollyowned direct or indirect subsidiaries of Camargo Corrêa. As used in this prospectus, "selling shareholders" includes the donees, assignees, pledgees and other transferees who may later hold the selling shareholders' respective interests in the shares of Alcoa common stock acquired in the Share Exchange and who have the benefits of a registration rights agreement with Alcoa dated as of July 25, 2003, filed as an exhibit to the registration statement of which this prospectus is a part.

The following table provides information regarding the beneficial ownership of Alcoa's common stock by the selling shareholders as of August 12, 2003 and the number of shares to be offered by them pursuant to this prospectus from time to time. The information in the table has been provided by the selling shareholders. Information about the selling shareholders may change over time. Alcoa cannot estimate the number of shares of Alcoa common stock that will be held by the selling shareholders upon the termination of the offering since it is possible that they may not sell any or all of the shares covered by this prospectus or may acquire or dispose of shares of Alcoa common stock not included in this prospectus.

Name of Selling Shareholder	Number of Shares Beneficially Owned Prior to the Offering	Number of Shares Being Offered
Allpar Limited (1)	10,366,471(2)	10,366,471(2)
Trelawney Inc. (3)	7,407,069(4)	7,407,069(4)

- (1) Allpar is a wholly owned subsidiary of Construções e Comércio Camargo Corrêa S.A., which in turn is a wholly owned subsidiary of Camargo Corrêa S.A., which in turn is a majority owned subsidiary of Participações Morro Vermelho S.A. Participações Morro Vermelho S.A. is owned, directly and indirectly, in equal proportions by three descendants of the founder of the Camargo Group.
- (2) Represents 1.2% of Alcoa's outstanding common stock as of August 4, 2003.
- (3) Trelawney is a wholly owned subsidiary of International Engineering Holding Ltd., which in turn is a wholly owned subsidiary of Camargo Corrêa S.A., which in turn is a majority owned subsidiary of Participações Morro Vermelho S.A. Participações Morro Vermelho S.A. is owned, directly and indirectly, in equal proportions by three descendants of the founder of the Camargo Group.
- (4) Represents 0.9% of Alcoa's outstanding common stock as of August 4, 2003.

Relationship with Alcoa

The Camargo Group became the principal minority shareholder of Alcoa Alumínio in 1984. Immediately before the Share Exchange, Allpar owned approximately 29% of the common stock of Alcoa Alumínio and Trelawney owned approximately 12% of the common stock of Alcoa Alumínio and approximately 41% of the common stock of ALAHC. The aggregate ownership of Allpar and Trelawney in each of Alcoa Alumínio and ALAHC was approximately 41%. Over the years, senior Alcoa executives had ongoing contacts with the Camargo Group and met annually with representatives of the Camargo Group to review Alcoa Alumínio results, issues, plans for the future and strategic direction. Alcoa and/or one or more of its subsidiaries were also parties to shareholder agreements with the Camargo Group relating to Alcoa Alumínio and ALAHC. In connection with the Share Exchange, the shareholder agreements were terminated and Camargo Group representatives serving as directors and/or executive council members of the council of administration of Alcoa Alumínio resigned their positions. During the five-year earn-out payment period after the closing of the Share Exchange (until December 31, 2007), Alcoa has agreed that the

Camargo Group may have a non-voting observer attend meetings of the board of directors of Alcoa Alumínio for so long as such body is maintained and retains its current scope of authority. Alcoa has reserved the right to change or eliminate such body, but if the board of directors is replaced by a consulting board or other body having similar functions, the Camargo Group will be entitled to have an observer attend the meetings of such board or other body. Unless Alcoa and the Camargo Group agree otherwise, the Camargo Group's observer will continue to be provided with, with respect to the period covered by the earn-out period, certain Alcoa Latin America financial information specified in the principal agreement, prepared and circulated monthly, and a reconciliation between certain financial statements and certain numbers included in the calculation of the earn-out (whether or not provided to the board of directors, consulting board or other similar body), as specified in the principal agreement. If the board of directors is eliminated without being replaced by a consulting board or other body having similar functions, the Camargo Group's observer will be provided with, with respect to the period covered by the earn-out period, and in addition to the documents required in the previous sentence, reports prepared by each business unit for review and discussion with the Alcoa senior executives as part of the quarterly review process and access on a bi-monthly basis to the Alcoa Alumìnio chief financial officer. In any event, the observer will not participate in management or in the formulation, determination or direction of basic business decisions but will receive as to content and timing the same information and material relating to the business of Alcoa Alumínio as is provided to the other members of such board or body during such five-year period.

In addition to the above-described relationships, Alcoa has worked and is continuing to work with the Camargo Group on hydroelectric and other projects in Brazil and throughout South America.

PLAN OF DISTRIBUTION

Alcoa is registering the resale of the shares of common stock acquired by the selling shareholders in the Share Exchange. See "Alcoa Inc.—Recent Developments" and "Selling Shareholders." Alcoa will not receive any of the proceeds from the sale of the shares by the selling shareholders.

Under the principal agreement and the registration rights agreement, Alcoa agreed to register the resale of the shares by the selling shareholders. Alcoa agreed to use its reasonable best efforts to cause the registration statement of which this prospectus is a part to remain effective for the sale of the shares for a period of two years after August 1, 2003, or such shorter period as will terminate when all of the shares have been disposed of by the selling shareholders or may be disposed of without regard to the volume limitations imposed under Rule 144 under the Securities Act (but in any event not before the expiration of any longer period required under the Securities Act). Alcoa will pay all costs and expenses incidental to the registration of the shares, except that the selling shareholders will pay underwriting discounts and commissions and transfer taxes, if any, attributable to the sale of the shares and all fees and expenses of the selling shareholders' counsel. Alcoa also agreed to indemnify each of the selling shareholders, each underwriter of any underwritten transaction and their respective officers, directors and affiliates for certain liabilities, including liabilities under the Securities Act.

Subject to the transfer restrictions and right of first refusal of Alcoa under the principal agreement (see "Plan of Distribution—Transfer Restrictions and Right of First Refusal" below), the selling shareholders may sell the common stock offered by this prospectus in one or more of the following ways at various times:

- · on the New York Stock Exchange or any other national securities exchange on which Alcoa common stock is traded;
- in the over-the-counter market;
- · in privately-negotiated transactions; or
- in a combination of such methods of sale.

The selling shareholders may sell their shares at:

- market prices prevailing at the time of the sale;
- · prices related to the then current market price;
- · negotiated prices; or
- fixed prices.

In addition to selling the shares under this prospectus, the selling shareholders may:

- · sell the shares under Rule 144 of the Securities Act rather than under this prospectus, if the transaction meets the requirements of Rule 144; or
- · transfer the shares in other ways not involving market makers or established trading markets, including directly by gift, distribution or other transfer; or
- sell the shares by any other legally available means.

The selling shareholders may offer and sell their shares in any manner permitted by law, including through underwriters, brokers, dealers or agents, and directly to purchasers. The shares of common stock may also be sold in the following types of transactions:

- sales to underwriters who will acquire the shares for their own account and resell them in one or more transactions at fixed prices or at varying prices determined at the time of sale;
- block transactions in which the broker or dealer engaged will attempt to sell the shares as an agent but may position and resell a portion of the block as a principal to facilitate the transaction;
- purchases by a broker or dealer as principal and resale by such broker or dealer for its account;
- ordinary brokerage transactions and transactions in which a broker solicits purchasers;
- an exchange distribution in accordance with the rules of any such exchange;
- transactions between sellers and purchasers without a broker-dealer;
- · sales directly to institutional investors; or
- sales through agents to the public or to institutional investors.

In effecting sales, brokers or dealers engaged by the selling shareholders may arrange for other brokers or dealers to participate in the resales.

In connection with the distribution of the shares, a selling shareholder may enter into hedging transactions with broker-dealers. In connection with such transactions, broker-dealers may engage in short sales of the shares in the course of hedging the positions they assume with the selling shareholder. A selling shareholder may also sell the shares short and redeliver the shares to close out the short positions. A selling shareholder may also enter into option or other transactions with broker-dealers, which require the delivery of the shares to the broker-dealer or an affiliate thereof. A selling shareholder may also loan or pledge the shares to a financial institution, broker-dealer or an affiliate thereof, who may sell the loaned shares or upon a default in the case of a pledge, sell the pledged shares. In addition to the foregoing, a selling shareholder may enter into, from time to time, other types of hedging transactions.

Certain persons participating in the distribution of the shares of common stock may engage in transactions that stabilize the price of the common stock. Under the rules and regulations under the Exchange Act, any person engaged in a distribution of the shares offered pursuant to this prospectus may be limited in its ability to engage in market activities with respect to those shares. The selling shareholders will be subject to the provisions of the Exchange Act and the rules and regulations under the Exchange Act, including Regulation M. Those rules and regulations may limit the timing of purchases and sales of any shares offered by the selling shareholders pursuant to this prospectus.

Brokers, dealers or agents may receive compensation in the form of commissions, underwriting discounts or concessions from the selling shareholders and/or the purchasers in amounts to be negotiated in connection with the sale. The selling shareholders and any brokers or dealers or agents and any other participating brokers or dealers may be deemed to be "underwriters" within the meaning of the Securities Act, and any commission, discount or concession, and any profit on the sale of the shares by the selling shareholders, may be deemed to be underwriting discounts or commissions under the Securities Act. The selling shareholders may also agree to indemnify any underwriter, broker-dealer or agent against certain liabilities related to the selling of the shares, including liabilities arising under the Securities Act.

Alcoa has not been advised by the selling shareholders of any selling arrangement as of the date of this prospectus between any selling shareholder and any underwriter, broker-dealer or agent.

At the time a particular offering of shares is made, to the extent required under the Securities Act, a prospectus supplement will be distributed, which will set forth the number of shares being offered and the terms of the offering, including the names of any underwriters, any discounts, commissions and other items constituting

compensation to the underwriters, broker-dealers or agents, the public offering price and any discounts, commissions or concessions allowed or paid by underwriters to dealers.

Transfer Restrictions and Right of First Refusal

The following summary of certain provisions of the principal agreement relating to the Share Exchange is not meant to be complete. For more information, you should refer to the full text of the principal agreement filed as an exhibit to the registration statement of which this prospectus is a part, including the definitions of terms used and not defined in this prospectus.

Agreed Volume Limitation. Under the principal agreement, the Camargo Group and its affiliates may not, before June 1, 2008, effect any sale, assignment, pledge, transfer or other disposition (each, a "Transfer"), other than exempt Transfers as described below, of the shares of Alcoa common stock acquired in the Share Exchange in excess of the following agreed volume limitation:

- from August 1, 2003 until August 1, 2005, 900,000 shares per day; and
- from August 1, 2005 until June 1, 2008, the greater of:
 - o an amount of shares per day equal to 30% of the average daily trading volume of Alcoa common stock on the New York Stock Exchange during the 30 day period before the applicable day; and
 - o 900,000 shares per day.

If there is a change in the number of outstanding shares of Alcoa common stock as a result of a stock split, reverse stock split or similar Alcoa corporate action, the number 900,000 will be adjusted so as to maintain the relationship between such number and the number of outstanding shares of Alcoa common stock immediately before such action.

The agreed volume limitation also applies to the sale of shares of Alcoa common stock by a hedging counterparty to establish its net initial hedge position with respect to a forward sale or other derivative transaction between the Camargo Group or its affiliates and such hedging counterparty involving shares of Alcoa common stock acquired in the Share Exchange (a "Monetization Arrangement"). As used in this prospectus, "initial hedge" means, with respect to any Monetization Arrangement, the sale of shares of Alcoa common stock by a hedging counterparty to establish its net initial hedge position with respect to such Monetization Arrangement in an amount not exceeding the aggregate number of shares of Alcoa common stock underlying such Monetization Arrangement multiplied by the initial delta (as determined by the hedging counterparty in a commercially reasonable manner) for such Monetization Arrangement. For the avoidance of doubt, the initial hedge will not include any sales of shares of Alcoa common stock made to dynamically adjust the hedging counterparty's net initial hedge position with respect to any Monetization Arrangement.

Right of First Refusal of Alcoa. Under the principal agreement, Alcoa has a right of first refusal in connection with:

- any Transfer of the shares of Alcoa common stock acquired in the Share Exchange by the Camargo Group or its affiliates that occurs on or before June 1, 2008, other than exempt Transfers as described below; and
- any Transfer of Alcoa common stock occurring on or before June 1, 2008 that is made by a hedging counterparty to establish its net initial hedge
 position with respect to a Monetization Arrangement.

The purchase price for any shares of Alcoa common stock purchased by Alcoa in the exercise of its right of first refusal will equal the volume weighted average of the sales prices per share of Alcoa common stock on the New York Stock Exchange on the trading day on which the Camargo Group's proposed sale notice becomes effective pursuant to the terms of the principal agreement, if Alcoa's exercise notice is delivered to the Camargo Group at or before 1:00 p.m., New York City time, on such day, and otherwise on the subsequent trading day. The Camargo Group must also specify a minimum price per share at which the Camargo Group is willing to Transfer the shares, and a sale of the shares will not be consummated if the applicable price described in the foregoing sentence is below such minimum price. Such minimum price will be no greater than the closing price per share of Alcoa common stock on the New York Stock Exchange on the trading day immediately before the trading day on which the Camargo Group delivered its proposed sale notice to Alcoa in accordance with the right of first refusal provisions.

Exempt Transfers. Under the principal agreement, the following Transfers of the shares of Alcoa common stock acquired in the Share Exchange by the Camargo Group and its affiliates are not subject to the agreed volume limitation or right of first refusal of Alcoa described above:

- (a) **Transfers to Camargo Corrêa Affiliates**—any Transfer of such shares to Camargo Corrêa or any of its affiliates who execute an agreement confirming that any subsequent Transfer of such shares by such Camargo Corrêa affiliate before June 1, 2008 will be subject to the restrictions and obligations set forth in the principal agreement and registration rights agreement;
- (b) **Transfers to Alcoa**—any Transfer of such shares to Alcoa or any of its affiliates;
- (c) **Alcoa Business Combinations**—any Transfer of such shares in a tender offer, merger or other Alcoa business combination transaction involving Alcoa common stock;
- (d) **Alcoa-Approved Limited Transfers**—any Transfer (other than or in relation to a Monetization Arrangement) of such shares to a person (excluding the Camargo Group and its affiliates) who:
 - o together with its affiliates, will not own, directly or indirectly, after giving effect to such Transfer, more than 1,800,000 shares of Alcoa common stock; and
 - o has been approved by Alcoa. Alcoa will have the right to withhold such approval only for good reason. Good reason means a determination made by Alcoa in good faith and upon a reasonable basis that such exempt Transfer, when considered together with all other prior exempt Transfers, would reasonably be expected to materially adversely affect the market price for Alcoa common stock within a period of 15 trading days from the time that such proposed exempt Transfer would be effected, if approved, or reasonably could result in public sales of Alcoa common stock on a single trading day in excess of the agreed volume limitation described above;
- (e) Alcoa-Agreed Restricted Transfers—any Transfer (other than or in relation to a Monetization Arrangement) of such shares to a person (excluding the Camargo Group and its affiliates) who executes an agreement with Alcoa before such Transfer confirming that any subsequent Transfer by such person or its affiliates that occurs before June 1, 2008 will be subject to the agreed volume limitation and right of first refusal of Alcoa described above. The Camargo Group and its affiliates may not make more than two of these Transfers before June 1, 2008;
- (f) **Long Term Alcoa Investor**—any Transfer of such shares to a long term Alcoa investor who is an "accredited investor" within the meaning of Rule 501 of Regulation D under the Securities Act and who represents and warrants to Alcoa that:
 - o it is acquiring the shares for investment purposes and not with a view to distribution thereof;
 - o it has been a beneficial owner of Alcoa common stock at all times during the 180 day period before the date of the proposed Transfer, and is, as of such date, the beneficial owner of at least 1% of Alcoa's then outstanding common stock;
 - o during the 180 day period before the Transfer, it and its affiliates have not sold in the aggregate more than 30% of the number of shares of Alcoa common stock of which it and its affiliates were the beneficial owners as of the beginning of such 180 day period; and
 - o it and its affiliates do not have any present intention to sell any of such shares or other shares of Alcoa common stock of which it or its affiliates are the beneficial owners;
- (g) **Monetization Arrangements** any Transfer of such shares, directly or indirectly, to a hedging counterparty in connection with a Monetization Arrangement. Before making such Transfer, the Camargo Group or its affiliates must enter into an agreement with such hedging counterparty assuring that:

- o the hedging counterparty will not enter into any Monetization Arrangements with other financial institutions or other persons in connection with establishing its initial hedge;
- o before establishing its initial hedge, the hedging counterparty will grant Alcoa a right of first refusal with respect to the Alcoa common stock to be sold under such initial hedge on the same terms as the right of first refusal of Alcoa described above;
- o the hedging counterparty will not effect any initial hedge Transfers, or any Transfers in relation to the early termination of a Monetization Arrangement, on any day which, when aggregated with (i) all initial hedge Transfers, or any Transfers in relation to the early termination of a Monetization Arrangement, made or to be made on such day by all other hedging counterparties to which a Transfer has been made under this provision and (ii) any Transfers (other than exempt Transfers) made, or to be made, by the Camargo Group or its affiliates on such day, exceed the agreed volume limitation described above;
- the maturity date of such Monetization Arrangement is such that after taking into account the maturity dates of all other Monetization Arrangements previously effected under this provision, there will not mature on any single day Monetization Arrangements that in the aggregate could result in sales of Alcoa common stock on any day in excess of the agreed volume limitation described above; and
- o the Monetization Arrangement will not mature before August 1, 2004; and
- (h) Pledges—any Transfer (other than or in relation to a Monetization Arrangement) of such shares to a bank or other financial institution as security for a loan or other financing, or upon foreclosure, so long as such bank or financial institution executes an agreement confirming that any subsequent Transfer of such shares by it or its affiliates that occurs before June 1, 2008 will be subject to the agreed volume limitation and right of first refusal of Alcoa described above.

Restrictions on Transferees. Persons who acquire shares of Alcoa common stock from the Camargo Group or its affiliates in exempt Transfers described in clauses (a), (e), (g) and (h) above will, in turn, be subject to the following restrictions under the principal agreement on the subsequent Transfer of such shares before June 1, 2008:

- if Camargo Corrêa or any of its affiliates obtain the shares in an exempt Transfer under clause (a) above (Transfers to Camargo Corrêa Affiliates), it (and any subsequent affiliate acquiring the shares in an exempt Transfer under such clause) will be subject to the restrictions in the principal agreement, including the agreed volume limitation and right of first refusal of Alcoa described above;
- if the transferee obtains the shares in an exempt Transfer under clause (e) above (Alcoa-Agreed Restricted Transfers), it (and any subsequent transferee acquiring the shares in an exempt Transfer under such clause) will be subject to the restrictions in the principal agreement, including the agreed volume limitation and right of first refusal of Alcoa described above;
- if the transferee is a hedging counterparty that obtains the shares in an exempt Transfer under clause (g) above (Monetization Arrangements), the following restrictions will apply:
 - o it and its affiliates will be subject to the restrictions in the principal agreement, including the agreed volume limitation and right of first refusal of Alcoa described above, only with respect to initial hedge transfers (but not with respect to other transactions involving Alcoa common stock to manage its exposure under the Monetization Arrangement); and
 - o it and its affiliates will be subject to the agreed volume limitation (but not the right of first refusal of Alcoa) described above with respect to sales of the shares of Alcoa common stock acquired in the Share Exchange by the Camargo Group or sales of other Alcoa common stock made at maturity or earlier termination of the Monetization Arrangement; and
- if the transferee obtains the shares in an exempt Transfer under clause (h) above (Pledges), whether as collateral or upon foreclosure, it (and any subsequent transferee acquiring the shares in an exempt Transfer under such clause) will be subject to the restrictions in the principal agreement, including the agreed volume limitation and right of first refusal of Alcoa described above.

LEGAL MATTERS

The validity of the shares of common stock offered by this prospectus has been passed upon for Alcoa by Denis A. Demblowski, Assistant General Counsel, of Alcoa. Mr. Demblowski is paid a salary by Alcoa, is a participant in various benefit plans offered by Alcoa to employees of Alcoa generally and beneficially owns, or has rights to acquire, an aggregate of less than 1% of Alcoa's outstanding common stock.

INDEPENDENT ACCOUNTANTS

The consolidated financial statements and related schedule of Alcoa and its subsidiaries incorporated by reference in Alcoa's Annual Report on Form 10-K for the fiscal year ended December 31, 2002, have been so incorporated in reliance on the reports of PricewaterhouseCoopers LLP, independent accountants, given on the authority of said firm as experts in auditing and accounting.

With respect to the unaudited financial information of Alcoa for the three-month periods ended March 31, 2003 and 2002 and the three-month and sixmonth periods ended June 30, 2003 and 2002, incorporated by reference in this prospectus, PricewaterhouseCoopers LLP reported that they have applied limited procedures in accordance with professional standards for a review of such information. However, their separate reports dated April 4, 2003 and July 8, 2003, except for Note Q, as to which the date is July 18, 2003, incorporated by reference herein, state that they did not audit and they do not express an opinion on that unaudited financial information. Accordingly, the degree of reliance on their reports on such information should be restricted in light of the limited nature of the review procedures applied. PricewaterhouseCoopers LLP is not subject to the liability provisions of Section 11 of the Securities Act for their reports on the unaudited financial information because those reports are not "reports" or "parts" of the registration statement prepared or certified by PricewaterhouseCoopers LLP within the meaning of Sections 7 and 11 of the Securities Act.

PART II. INFORMATION NOT REQUIRED IN PROSPECTUS

Item 14. Other Expenses of Issuance and Distribution.

The following table sets forth the fees and expenses in connection with the offering and distribution of the shares of common stock registered under this registration statement, other than underwriting compensation. Alcoa will pay all of the costs identified below. All such expenses, other than the SEC registration fee, are estimates.

SEC registration fee	\$38,082.24
Printing expenses	25,000.00
Accountants' fees and expenses	20,000.00
Legal fees and expenses	10,000.00
Miscellaneous expenses	5,000.00
m . 1	400,000,00
Total	\$98.082.24

Item 15. Indemnification of Directors and Officers.

Article V of the By-Laws of Alcoa provides that Alcoa shall indemnify, under specified circumstances, persons who were or are directors, officers or employees of Alcoa or who served or serve other business entities at the request of Alcoa. Under these By-Law provisions, a person who is wholly successful in defending a claim will be indemnified for any reasonable expenses. To the extent a person is not successful in defending a claim, reasonable expenses of the defense and any liability incurred are to be indemnified under these provisions only where independent legal counsel or another disinterested person selected by the board of directors determines that such person acted in good faith and in a manner such person reasonably believed to be in, or not opposed to, the best interests of Alcoa, and in addition with respect to any criminal action or proceeding, had no reasonable cause to believe the conduct of such person was unlawful. Any expense incurred with respect to any claim may be advanced by Alcoa if the recipient agrees to repay such amount if it is ultimately determined that such recipient is not to be indemnified pursuant to Article V.

The foregoing By-Law provisions generally parallel Sections 1741 and 1745 of the Pennsylvania Business Corporation Law ("PBCL"). Section 1746 and the By-Laws both also provide that the indemnification provided for therein shall not be deemed exclusive of any other rights to which those seeking indemnification may otherwise be entitled.

Section 1746 of the PBCL and the By-Laws provide for increased indemnification protections for directors, officers and others. Indemnification may be provided by Pennsylvania corporations in any case except where the act or failure to act giving rise to the claim for indemnification is determined by a court to have constituted willful misconduct or recklessness.

Section 1713 of the PBCL also sets forth a framework whereby Pennsylvania corporations, with the approval of the shareholders, may limit the personal liability of directors for monetary damages except where the act or omission giving rise to a claim constitutes self-dealing, willful misconduct or recklessness. The section does not apply to a director's responsibility or liability under a criminal or tax statute and may not apply to liability under Federal statutes, such as the Federal securities laws.

Alcoa's Articles of Incorporation and By-Laws were amended by the shareholders to implement the increased protections made available to directors under the PBCL as described in the preceding paragraph. Article VIII of the By-Laws provides that, except as prohibited by law, every director of Alcoa shall be entitled as of right to be indemnified by Alcoa for expenses and any and all liability paid or incurred by such person by reason of such person being or having been a director of Alcoa. Expenses incurred with respect to any claim may be advanced by Alcoa, subject to certain exceptions. The shareholders have also approved a form of indemnity agreement. Alcoa has entered into such an indemnity agreement with each of its current directors.

Alcoa has purchased a three-year liability insurance policy with an aggregate limit of \$150 million, with certain specified deductible amounts for liability of directors and officers and reimbursement to Alcoa for indemnification provided to directors and officers. The policy has an expiration date of October 1, 2003 and provides liability insurance and reimbursement coverage for Alcoa, and its directors and officers, which is permitted by the laws of Pennsylvania referred to above.

The Articles of Incorporation provide that except as prohibited by law, Alcoa may indemnify any person who is or was a director, officer, employee or agent of Alcoa or is or was serving at the request of Alcoa as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, including, without limitation, any employee benefit plan, and may take such steps as may be deemed appropriate by the board of directors, including purchasing and maintaining insurance, entering into contracts, including, without limitation, contracts of indemnification between Alcoa and its directors and officers, creating a trust fund, granting security interests or using other means, including, without limitation, a letter of credit to ensure the payment of such amounts as may be necessary to effect such indemnification. The By-Laws provide for indemnification of such persons to the fullest extent permitted by law.

The Articles of Incorporation also provide that to the fullest extent that the laws of the Commonwealth of Pennsylvania permit elimination or limitation of the liability of directors, no director of Alcoa shall be personally liable for monetary damages for any action taken, or any failure to take any action.

Under the registration rights agreement relating to the shares of Alcoa common stock registered under this registration statement, each of Alcoa and the selling shareholders have agreed under certain circumstances to indemnify each other, each underwriter of any underwritten transaction, and their respective directors, officers and affiliates for certain liabilities, including liabilities incurred under the Securities Act, in connection with the registration of the shares.

Item 16. Exhibits.

Exhibit Number	Description
4(a)	Articles of Incorporation of Alcoa, as amended (incorporated by reference to Exhibit 3(a) to Alcoa's Quarterly Report on Form 10-Q for the quarter ended June 30, 2000).
4(b)	By-Laws of Alcoa, as amended (incorporated by reference to Exhibit 3 to Alcoa's Quarterly Report on Form 10-Q for the quarter ended June 30, 2002).
4(c)	Form of certificate for shares of Alcoa common stock (incorporated by reference to Exhibit 4(i) to Registration Statement No. 33-49997 on Form S-3).
4(d)	Principal Agreement dated as of July 25, 2003 among Alcoa, Alcoa Luxembourg S.à.r.l., Camargo Corrêa, Allpar and Trelawney.
4(e)	Registration Rights Agreement dated as of July 25, 2003 by and among Alcoa, Camargo Corrêa, Allpar and Trelawney.
5	Opinion of Denis A. Demblowski, Esq., counsel to Alcoa.
15	Independent Accountants' letter regarding unaudited financial information.
23(a)	Consent of Denis A. Demblowski, Esq., counsel to Alcoa (included in Exhibit 5).
23(b)	Consent of Independent Accountants.
24	Powers of Attorney of certain directors and officers of Alcoa.
	XI 0

Item 17. Undertakings.

The undersigned Registrant hereby undertakes:

- (a) (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
 - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act of 1933;
 - (ii) To reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent posteffective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth
 in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total
 dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the
 estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b)
 if, in the aggregate, the changes in volume and price represent no more than a 20 percent change in the maximum aggregate
 offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement; and
 - (iii) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement;

provided, *however*, that paragraphs (a)(1)(i) and (a)(1)(ii) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports filed with or furnished to the Commission by the Registrant pursuant to Section 13 or Section 15(d) of the Securities Exchange Act of 1934 that are incorporated by reference in the registration statement.

- (2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) That, for purposes of determining any liability under the Securities Act of 1933, each filing of the Registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.

Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions described in Item 15 above, the Registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended, Alcoa Inc. certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-3 and has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of New York, State of New York, on August 13, 2003.

ALCOA INC.

By: /S/ ALAIN J. P. BELDA

Name: Alain J. P. Belda

Title: Chairman of the Board and Chief Executive Officer

Pursuant to the requirements of the Securities Act of 1933, as amended, this registration statement has been signed below by the following persons in the capacities and on the dates indicated.

Signature	Title	Date
/S/ ALAIN J. P. BELDA Alain J. P. Belda	Chairman of the Board and Chief Executive Officer and Director (Principal Executive Officer)	August 13, 2003
* RICHARD B. KELSON Richard B. Kelson	Executive Vice President and Chief Financial Officer (Principal Financial Officer)	August 13, 2003
/S/ CHARLES D. MCLANE, JR. Charles D. McLane, Jr.	Vice President—Corporate Controller (Principal Accounting Officer)	August 13, 2003

^{*} Richard B. Kelson, Executive Vice President and Chief Financial Officer (Principal Financial Officer); and Kathyrn S. Fuller, Carlos Ghosn, Joseph T. Gorman, Judith M. Gueron, Sir Ronald Hampel, John P. Mulroney, Henry B. Schacht, Franklin A. Thomas and Ernesto Zedillo, each as a Director, on August 13, 2003, by William B. Plummer, their attorney-in-fact.

* By: <u>/s/ WILLIAM B. PLUMMER</u>
William B. Plummer
Attorney-in-Fact

EXHIBIT INDEX

Exhibit Number	Description
4(a)	Articles of Incorporation of Alcoa, as amended (incorporated by reference to Exhibit 3(a) to Alcoa's Quarterly Report on Form 10-Q for the quarter ended June 30, 2000).
4(b)	By-Laws of Alcoa, as amended (incorporated by reference to Exhibit 3 to Alcoa's Quarterly Report on Form 10-Q for the quarter ended June 30, 2002).
4(c)	Form of certificate for shares of Alcoa common stock (incorporated by reference to Exhibit 4(i) to Registration Statement No. 33-49997 on Form S-3).
4(d)	Principal Agreement dated as of July 25, 2003 among Alcoa, Alcoa Luxembourg S.à.r.l., Camargo Corrêa, Allpar and Trelawney.
4(e)	Registration Rights Agreement dated as of July 25, 2003 by and among Alcoa, Camargo Corrêa, Allpar and Trelawney.
5	Opinion of Denis A. Demblowski, Esq., counsel to Alcoa.
15	Independent Accountants' letter regarding unaudited financial information.
23(a)	Consent of Denis A. Demblowski, Esq., counsel to Alcoa (included in Exhibit 5).
23(b)	Consent of Independent Accountants.
24	Powers of Attorney of certain directors and officers of Alcoa.

PRINCIPAL AGREEMENT

THIS PRINCIPAL AGREEMENT (this "Agreement") is entered into as of July 25, 2003, among ALCOA INC., a corporation organized and existing under the laws of the Commonwealth of Pennsylvania, United States of America ("Alcoa"), ALCOA LUXEMBOURG S.à.r.l., a société à responsabilité limitée (private limited liability company) organized and existing under the laws of Luxembourg ("AL"), CAMARGO CORRÊA S.A., a corporation organized and existing under the laws of the Federative Republic of Brazil ("Camargo Corrêa"), ALLPAR LIMITED, a corporation organized and existing under the laws of the Cayman Islands ("Allpar"), and TRELAWNEY INC., a corporation organized and existing under the laws of The Bahamas ("Trelawney"). The entities listed above are collectively referred to herein as the "parties" and are referred to individually as a "party".

Recital

Alcoa, Allpar and Trelawney own, directly or indirectly, virtually all of the capital stock of Alcoa Alumínio S.A. ("Alcoa Alumínio"), a Brazilian corporation, and Alcoa Latin American Holdings Corporation ("ALAHC"), a British Virgin Islands company. Alcoa owns, directly or indirectly, approximately 59% of the capital stock of Alcoa Alumínio and approximately 59% of the capital stock of ALAHC. Allpar owns approximately 29% of the capital stock of Alcoa Alumínio. Trelawney owns approximately 12% of the capital stock of Alcoa Alumínio and approximately 41% of the capital stock of ALAHC. The aggregate ownership of Allpar and Trelawney in each of Alcoa Alumínio and ALAHC is approximately 41%.

Pursuant to the Agreement in Principle entered into as of May 2, 2003 (the "Agreement in Principle") among Alcoa, Alcoa Brazil Holdings Company, Construções e Comércio Camargo Corrêa S.A., Allpar and Trelawney, the parties have agreed in principle that Allpar and Trelawney would exchange all of their shares in Alcoa Alumínio and ALAHC with Alcoa, AL and/or their designated U.S. Person affiliate(s) for shares of common stock in Alcoa and other consideration as described further herein, and that the parties would negotiate and execute mutually acceptable definitive documents.

In consideration of the mutual agreements, covenants, representations and warranties contained herein and intending to be legally bound, the parties hereby agree as follows:

ARTICLE I DEFINITIONS

As used herein, the terms below shall have the following meanings:

"Affiliate" means any person or entity which directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such person or entity. The term "control" (including the terms "controlled by" and "under common control with") as used with respect to any person or entity means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such person or entity, whether through the ownership of voting securities, by contract or otherwise.

- "Alcoa Alumínio Stock" means all of the shares of common stock of Alcoa Alumínio held by Allpar, Trelawney and the individuals set forth on **Exhibit A**, as set forth on **Exhibit A**, attached hereto and made a part hereof.
 - "Alcoa Exercise Notice" has the meaning set forth in Section 2.4.2.
 - "Alcoa Group" means the following members: Alcoa and AL.
 - "ALAHC Stock" means all of the shares of common stock of ALAHC held by Trelawney, as set forth on Exhibit A.
- "Average Price" means the volume weighted average of the sales prices per share of the Common Stock on the applicable Trading Day on the New York Stock Exchange as reported by Bloomberg Financial Markets (or, if not reported thereby, by another authoritative source as may be agreed by the parties in writing).
 - "BACEN" means the Central Bank of Brazil.
 - "CADE" means the Conselho Administrativo de Defesa Econômica, an agency of the Federative Republic of Brazil.
 - "Camargo Group" means the following members: Camargo Corrêa, Allpar and Trelawney.
 - "Common Stock" means the common stock of Alcoa, par value \$1.00 per share.
- "Closing" means the satisfactory completion of the actions described in Article III of this Agreement, including, without limitation, satisfaction or waiver (other than Section 5.2.5, which may not be waived) of the conditions set forth in Sections 3.4 and 5.2.5 of this Agreement.
- "Closing Date" means August 1, 2003, provided the conditions set forth in Section 5.1 and Section 5.2 have been satisfied or waived on such date, and, otherwise, as soon as practicable after the satisfaction or waiver of such conditions, or such other date as may be mutually agreed by the parties in writing.
 - "Consideration" has the meaning set forth in Section 2.1.
 - "Delay Period" has the meaning set forth in Section 4.2(g) of the Registration Rights Agreement.
 - "Delay Period Notice" has the meaning set forth in Section 4.2(g) of the Registration Rights Agreement.
 - "Delay Period Notice Deadline" has the meaning set forth in Section 4.2(g) of the Registration Rights Agreement.
 - "Delay Period Termination Date" has the meaning set forth in Section 2.7.2.
 - "Dollars" or "U.S.\$" means the lawful currency of the United States of America.
- "Earn-Out Payments" means the payments required to be made, if any, by Alcoa to Trelawney and Allpar pursuant to Section A of **Exhibit C**, attached hereto and made a part hereof.
 - "Earn-Out Period" has the meaning set forth in $\mathbf{Exhibit}\;\mathbf{C}.$
 - "Effectiveness Period" has the meaning set forth in Section 2.3.
 - "Exchange" means the exchange of the Alcoa Alumínio Stock and the ALAHC Stock for the Shares in accordance with this Agreement.
 - "Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended.
 - "Exempt Transfer Notice" has the meaning set forth in Section 2.4.4.
 - "Exercise Deadline" has the meaning set forth in Section 2.4.2.
 - "GAAP" means U.S. generally accepted accounting principles.
- "Good Reason" means, with respect to any Exempt Transfer as to which Alcoa has an approval right, a determination made by Alcoa, in good faith and upon a reasonable basis, that such Exempt Transfer, when considered together with all other prior Exempt Transfers, would reasonably be expected to materially adversely affect the market price for the Common Stock within a period of 15 Trading Days from the time that such proposed Exempt Transfer would be

effected, if approved, or reasonably could result in public sales of the Common Stock on a single Trading Day in excess of the Volume Limitation.

"Guaranteed Minimum Payment" means the payment required to be made, if any, by Alcoa and/or its designated affiliate(s) to Trelawney and Allpar pursuant to Section B of Exhibit C.

"Hedging Counterparty" means any financial institution that is a counterparty to a Monetization Arrangement.

"Initial Hedge" means, with respect to any Monetization Arrangement, the sale of shares of Common Stock by a Hedging Counterparty to establish its net initial hedge position with respect to such Monetization Arrangement in an amount not exceeding the aggregate number of shares of Common Stock underlying such Monetization Arrangement multiplied by the initial delta (as determined by the Hedging Counterparty in a commercially reasonable manner) for such Monetization Arrangement. For the avoidance of doubt, the Initial Hedge shall not include any sales of shares of Common Stock made to dynamically adjust the Hedging Counterparty's net initial hedge position with respect to any Monetization Arrangement.

"Liens" has the meaning set forth in Section 4.1.2.

"Long Term Alcoa Investor" means an "accredited investor" within the meaning of Rule 501 of Regulation D under the Securities Act who represents and warrants to Alcoa that: (i) it is acquiring the Shares for investment purposes and not with a view to distribution thereof, (ii) it has been a beneficial owner (as defined in Rule 13d-3 promulgated by the SEC) of Common Stock at all times during the 180 day period prior to the date of the proposed Transfer and is, as of such date, the beneficial owner of shares of Common Stock representing at least 1% of the then outstanding Common Stock determined based on the most recent filing by Alcoa with the SEC, (iii) during the 180 day period prior to the Transfer, it and its Affiliates have not effected sales of Common Stock that, when aggregated involve more than 30% of the number of shares of which such investor and its Affiliates were the beneficial owners as of the beginning of such 180 day period and (iv) that it and its Affiliates do not have any present intention to effect any sales of the Shares or of other shares of Common Stock of which it or its Affiliates are the beneficial owners.

"Material Adverse Effect" means, with respect to the Alcoa Group, a material adverse effect on the business, assets, operations or financial condition of Alcoa and its subsidiaries, taken as a whole, or a material impairment of the Alcoa Group to perform its obligations under this Agreement or the Registration Rights Agreement; provided, however, that none of the following, individually or in the aggregate, will be deemed to have a Material Adverse Effect on the Alcoa Group: (a) fluctuations in the market price of the Common Stock, (b) changes in economic, political or regulatory conditions (including acts of terrorism or war), or (c) changes or conditions generally affecting the industries in which Alcoa and its subsidiaries operate (including metals or aluminum pricing). "Material Adverse Effect" means, with respect to the Camargo Group, a material adverse effect on the business, assets, operations or financial condition of Camargo Corrêa and its subsidiaries, taken as a whole, or a material impairment of the Camargo Group to perform its obligations under this Agreement or the Registration Rights Agreement; provided, however, that none of the following, individually or in the aggregate, will be deemed to have a Material Adverse Effect on the Camargo Group: (a) changes in economic, political or regulatory conditions (including acts of terrorism or war) or (b) changes or conditions generally affecting the industries in which Camargo Corrêa and its subsidiaries operate.

"Monetization Arrangement" has the meaning set forth in Section 2.4.3(g).

"New York City Time" means the local time of New York, New York, United States of America.

- "Permitted Encumbrances" means the agreed upon restrictions and rights of first refusal set forth in or established pursuant to Sections 2.2 and 2.4 of this Agreement or in the Registration Rights Agreement and applicable restrictions under U.S. federal and/or state securities laws.
 - "Person" means any individual, corporation, limited liability company, partnership or any other private, public or government entity.
 - "Procedure I" has the meaning set forth in Section 2.7.
 - "Procedure I Rescission Deadline" has the meaning set forth in Section 2.7.1.
 - "Procedure II" has the meaning set forth in Section 2.7.
 - "Procedure II Rescission Deadline" has the meaning set forth in Section 2.7.2.
 - "Proposed Sale Notice" has the meaning set forth in Section 2.4.2.
 - "Reais" and "R\$" means the lawful currency of the Federative Republic of Brazil.
 - "Registration Rights Agreement" has the meaning set forth in Section 2.2.
 - "Restriction Termination Date" means June 1, 2008.
 - "SEC" means the U.S. Securities and Exchange Commission.
 - "Securities Act" means the U.S. Securities Act of 1933, as amended.
 - "Shares" means the 17,773,540 shares of Common Stock to be delivered in the Exchange in accordance with this Agreement.
 - "Shelf Notice of Sale" has the meaning set forth in Section 4.2(g) of the Registration Rights Agreement.
 - "Shelf Registration Statement" has the meaning set forth in Section 2.3.
 - "Special Option Expiration Date" has the meaning set forth in Section 2.7.3.
 - "Trade Date" has the meaning set forth in Section 2.4.2.
 - "Trading Day" means a day on which the New York Stock Exchange and commercial banks in New York City are open for the transaction of business.
 - "Transfer" has the meaning set forth in Section 2.4.2.
 - "Transferee" has the meaning set forth in Section 2.4.5.
- "Volume Limitation" means (i) for the period from the Closing Date until the second anniversary thereof, 900,000 Shares per day; or (ii) for the period from the second anniversary of the Closing Date until June 1, 2008, the greater of (x) an amount of Shares per day equal to 30% of the average daily trading volume of the Common Stock on the New York Stock Exchange during the 30 day period prior to the applicable day and (y) 900,000 Shares per day. If there is a change in the number of outstanding shares of Common Stock as a result of a stock split, reverse stock split or similar Alcoa corporate action, the number 900,000 shall be adjusted so as to maintain the relationship between such number and the number of outstanding Common Stock immediately prior to such action.
 - "U.S." means United States of America.

ARTICLE II EXCHANGE AND OTHER MATTERS

2.1 Exchange and Consideration. The parties hereby agree, on the terms and subject to the conditions set forth herein, to exchange (i) the Alcoa Alumínio Stock and the ALAHC Stock for (ii) the Shares and any Earn-Out Payment(s) and any Guaranteed Minimum Payment (the "Consideration"). The number of Shares (17,773,540) was determined by dividing (i) U.S.\$410 million, which was the amount set forth in the Agreement in Principle, by (ii) \$23.068, representing the average of the closing prices of the Common Stock on the New York Stock Exchange for the five Trading Days beginning two Trading Days prior to and ending two

Trading Days after May 5, 2003, being the first date of public announcement of the transaction. On the Closing Date and provided that the transfer of foreign investment registrations with BACEN have been effected and evidenced as provided in Section 5.2.5, the Camargo Group will deliver, or cause to be delivered, to Alcoa and AL the Alcoa Alumínio Stock and the ALAHC Stock, free and clear of all Liens, and the Alcoa Group will deliver, or cause to be delivered, to Trelawney and Allpar the Shares, free and clear of all Liens other than the Permitted Encumbrances. Specifically, the transfers to be effected at Closing to complete the Exchange are as follows:

Transferor:	Transfer of:	Transferee:
Allpar	1,706,022 Alcoa Alumínio shares	AL
AL	10,366,471 Alcoa shares	Allpar
Trelawney	750,988 Alcoa Alumínio shares	AL
AL	4,563,303 Alcoa shares	Trelawney
Trelawney	363,992 ALAHC shares	Alcoa
Alcoa	2,843,766 Alcoa shares	Trelawney
Carlos Pires Oliveira Dias	28 Alcoa Alumínio shares	AL
Fernando de Arruda Botelho	28 Alcoa Alumínio shares	AL
Luiz Roberto Ortiz Nascimento	28 Alcoa Alumínio shares	AL
Fernando Tigre de Barros Rodrigues	1 Alcoa Alumínio share	AL
Daniel Antonio Biondo Bastos	1 Alcoa Alumínio share	AL
Edy Luiz Kogut	1 Alcoa Alumínio share	AL

In accordance with the above table, the Consideration will be allocated to Allpar and Trelawney (with no separate allocations to the individuals) as follows:

Allocation of Consideration between:	Allocation percentage	Allocation of the Shares
Allpar	58%	10,366,471 shares
Trelawney	42%	7,407,069 shares
Total	100%	17,773,540 shares

The transfers described in this Section are based on the following agreed upon allocation of the Consideration between the Alcoa Alumínio Stock and the ALAHC Stock:

Allocation of Consideration between:	Allocation percentage	Allocation of the Shares
Alcoa Alumínio Stock	84%	14,929,774 shares
ALAHC Stock	16%	2,843,766 shares
Total	100%	17,773,540 shares

^{2.2 &}lt;u>Acquisition of the Shares</u>. The Shares will be acquired initially by the Camargo Group from the Alcoa Group in a private placement transaction under section 4(2) of the Securities Act. The Shares will be registered for resale by the Shelf Registration Statement, as more fully described in Section 2.3 and in the Registration Rights Agreement, attached hereto as **Exhibit E** and made a part hereof (the "Registration Rights Agreement"). If the Shares are delivered in certificated form at the Closing, then as soon as practicable following the Closing, the Camargo Group agrees to establish an account at Alcoa's transfer agent, Equiserve Trust Company, N.A.

("Equiserve"), to hold the Shares in book-entry form through the direct registration system to eliminate the need for physical certificates.

2.2.1 The certificate(s) representing and/or the book entry account(s) evidencing ownership of the Shares by the Camargo Group shall be endorsed with legends substantially to the following effect:

"The Shares represented by [this certificate] [this book-entry account] have not been registered under the Securities Act of 1933 and may not be transferred in the absence of such registration or an exemption therefrom under such Act. In addition, the Shares are subject to certain rights of first refusal and restrictions on transfer as provided in the Principal Agreement, dated as of July 25, 2003 among Alcoa Inc. and certain other signatories thereto and the Registration Rights Agreement, dated as of July 25, 2003 among Alcoa Inc. and certain other signatories of which are on file at the principal office of Alcoa Inc."

Insofar as such legends (a) relate to the Securities Act or any U.S. securities laws, such legends shall be removed as and when required pursuant to the terms of the Registration Rights Agreement, and (b) relate to the rights of first refusal and other contractual restrictions on transfer set forth in this Agreement, such legends shall, upon the request of the holder of such Shares, be removed and Alcoa shall issue a certificate without such legend or create a book-entry account without such legend, as the case may be, if such Shares are no longer subject to the rights of first refusal and other contractual restrictions on transfer as set forth herein.

2.3 <u>Shelf Registration</u>. Alcoa shall (a) no later than 30 days after the Closing Date, prepare and file with the SEC a registration statement for the resale of the Shares in an offering to be made on a delayed or continuous basis pursuant to Rule 415 under the Securities Act (or any similar rule that may be adopted by the SEC) (the "Shelf Registration Statement"), (b) use its reasonable best efforts to cause such Shelf Registration Statement to become effective as soon as practicable after such filing, but in all events within 120 days after the Closing Date, (c) use its reasonable best efforts to cause the Shelf Registration Statement to remain effective at all times thereafter until the second anniversary of the Closing Date or such shorter period as will terminate when all of the Shares have been disposed of by the Camargo Group and its Affiliates or may be disposed of without regard to the volume limitations imposed under Rule 144 (but in any event not before the expiration of any longer period required under the Securities Act) (the "Effectiveness Period"), (d) prepare and file with the SEC such amendments and supplements to the Shelf Registration Statement and the prospectus used in connection therewith as may be necessary to keep such registration statement effective for the period specified in this sentence above, and (e) when requested by Camargo Corrêa, keep Camargo Corrêa reasonably apprised of the status of the Shelf Registration Statement prior to the expiration of the Effectiveness Period, all as more fully set forth in the Registration Rights Agreement. Alcoa acknowledges that the foregoing covenants have been determinative to the Camargo Group's decision to consummate the Exchange.

2.4 Transfer Restrictions and Right of First Refusal.

2.4.1 Volume Limitation. Prior to the Restriction Termination Date, the Camargo Group and its Affiliates shall not effect any Transfer of Shares, other than Exempt Transfers made in accordance with Section 2.4.3, in excess of the Volume Limitation. The parties agree

that this provision shall apply regardless of whether, at the time of the proposed Transfer, the volume limitations imposed under Rule 144 under the Securities Act are applicable.

2.4.2 Right of First Refusal. The parties agree that, subject to Section 2.4.3, Alcoa will have a right of first refusal, on the terms set forth below, (x) in connection with any sale, assignment, pledge, transfer or other disposition (each, a "Transfer") of Shares by the Camargo Group or its Affiliates that occurs on or before the Restriction Termination Date and (y) in connection with any Transfer of Common Stock occurring on or before the Restriction Termination Date that is made by a Hedging Counterparty to establish the Initial Hedge (the "Initial Hedge Transfer"). Until the Restriction Termination Date, prior to the Camargo Group or its Affiliates making any Transfer of Shares, Camargo Corrêa will deliver to Alcoa no later than 9:00 a.m., New York City Time, on a Trading Day, a written notice (the "Proposed Sale Notice") specifying the number of Shares proposed to be Transferred (which must be in compliance with the Volume Limitation), the name of the transferor, a description of how and when the proposed Transfer will take place, the date and time of the Exercise Deadline determined in accordance with the provisions of this Section 2.4.2 and, other than in connection with a sale to be effected directly by the Camargo Group or its Affiliates on a national securities exchange or in an underwritten public offering, identifying the price, the name of the proposed Transferee and all other material terms of the proposed Transfer. Until the Restriction Termination Date, prior to permitting a Hedging Counterparty to make an Initial Hedge Transfer, Camargo Corrêa will cause such Hedging Counterparty to deliver a Proposed Sale Notice to Alcoa at the same time and containing the information specified in the preceding sentence. A Proposed Sale Notice given (x) on or before 9:00 a.m., New York City Time, on a day that is a Trading Day shall be effective at 9:00 a.m. on such Trading Day, and (y) on a day that is not a Trading Day or after 9:00 a.m., New York City Time, on a Trading Day will be deemed to be effective at 9:00 a.m., New York City Time, on the next succeeding Trading Day. Such notice shall be in writing, shall be sent to Alcoa's group president—Latin America and to its treasurer and general counsel as set forth in Section 7.6 of this Agreement. Alcoa shall have the right to acquire, or cause one or more of its designated Affiliate(s) to acquire all, but not less than all, of the Shares (or, in the case of an Initial Hedge Transfer, the Common Stock) identified in the Proposed Sale Notice for a price, payable in cash, equal to (x) the Average Price for the Trading Day on which the Proposed Sale Notice was effective if Alcoa delivers the applicable Alcoa Exercise Notice at or before 1:00 p.m. New York City Time on such Trading Day or (y) the Average Price for the Trading Day subsequent to the Trading Day on which the Proposed Sale Notice was effective if the Alcoa Exercise Notice is delivered prior to the Exercise Deadline but after 1:00 p.m. New York City Time on the day the Proposed Sale Notice was effective. The Trading Day on which the price for the Shares was determined in accordance with the foregoing sentence is referred to herein as the "Trade Date". The Camargo Group shall (but this provision is personal to the Camargo Group and its Affiliates and shall not apply to any other Person) specify in its Proposed Sale Notice a minimum per Share price at which it is willing to Transfer the Shares (which shall be no greater than an amount equal to the closing price per share of Common Stock on the New York Stock Exchange (as reported by Bloomberg Financial Markets, or, if not reported thereby, by another authoritative source) on the Trading Day immediately prior to the Trading Day on which the Camargo Group delivered its Proposed Sale Notice and, if no minimum price is specified, shall be deemed to be an amount equal to the closing price per share of Common Stock on the New York Stock Exchange (as reported by Bloomberg Financial Markets, or, if not reported thereby, by another authoritative source) on the Trading Day immediately prior to the Trading Day on which the Camargo Group delivered its Proposed Sale Notice), and if the applicable Average Price determined in accordance with this Section is less

than such minimum price, then such Proposed Sale Notice shall be deemed null and void immediately and automatically without further action or notice required by either party. To exercise its right of first refusal, Alcoa shall deliver to Camargo Corrêa (or, in the case of an Initial Hedge Transfer, to the Hedging Counterparty that delivered the Proposed Sale Notice) no later than 9:00 a.m., New York City Time, on the Trading Day next succeeding the effective date of the Proposed Sale Notice (the "Exercise Deadline"), a written notice (the "Alcoa Exercise Notice") confirming its election to purchase the Shares (or, in the case of an Initial Hedge Transfer, the Common Stock) described in the Proposed Sale Notice and identifying the purchaser or purchasers of such Shares (or, in the case of an Initial Hedge Transfer, the Common Stock). Alcoa's notice shall be sent to Camargo Corrêa's president and treasurer as set forth in Section 7.6 of this Agreement (or, in the case of an Initial Hedge Transfer, to the Hedging Counterparty delivering a Proposed Sale Notice at the address specified in such Proposed Sale Notice). Provided that Camargo Corrêa properly made (or, in the case of an Initial Hedge Transfer, caused the Hedging Counterparty to properly make) a Proposed Sale Notice in accordance with the provisions set forth herein, if Alcoa fails to deliver an Alcoa Exercise Notice to Camargo Corrêa (or, in the case of an Initial Hedge Transfer, to the Hedging Counterparty) by the Exercise Deadline (or if Alcoa notifies Camargo Corrêa, or, in the case of an Initial Hedge Transfer, the Hedging Counterparty, that it has elected not to purchase such Shares), then Alcoa will be deemed to have waived its right of first refusal with respect to the Transfer or Initial Hedge Transfer described in the Proposed Sale Notice and, subject to Section 4.2(g) of the Registration Rights Agreement, if the Transfer is proposed to be made under the Shelf Registration Statement, and subject to applicable law, the Camargo Group or its Affiliate shall be permitted to effect the Transfer (or, in the case of an Initial Hedge Transfer, permit consummation of the Initial Hedge Transfer) described in the Proposed Sale Notice by the end of the Trading Day on which the Exercise Deadline falls. Subject to the terms of this Agreement, delivery by Alcoa of an Alcoa Exercise Notice shall create a legally binding obligation on the part of Alcoa to purchase the Shares (or, in the case of an Initial Hedge Transfer, the Common Stock) and, subject to the provisions of Section 2.7, if applicable, a legally binding obligation on the part of the Camargo Group and/or its Affiliates to sell the Shares described in the Proposed Sale Notice on and subject to the terms set forth herein (or, in the case of an Initial Hedge Transfer, a legally binding obligation on the part of the Hedging Counterparty delivering a Proposed Sale Notice in connection with an Initial Hedge Transfer, to sell the Common Stock described in the Proposed Sale Notice on and subject to the terms set forth herein). Payment of the purchase price for, and delivery of the Shares purchased under this Section 2.4.2, shall be made on the third Trading Day after the Trade Date. Payment shall be made in U.S. Dollars upon confirmation of delivery of the Shares in the manner described below, by wire transfer of immediately available funds to the account specified in wire transfer instructions furnished to Alcoa by the president or treasurer of Camargo Corrêa (or, in the case of an Initial Hedge Transfer, by a duly authorized representative of the Hedging Counterparty that delivered the applicable Proposed Sale Notice). Delivery of the Shares (or in the case of an Initial Hedge Transfer, the Common Stock) shall be made as follows: (i) in the case of certificated Shares (or in the case of an Initial Hedge Transfer, certificated Common Stock), Camargo Corrêa (or in the case of an Initial Hedge Transfer, the Hedging Counterparty) shall physically deliver or cause to be delivered the Share certificates (or in the case of an Initial Hedge Transfer, the Common Stock certificates) representing the purchased Shares (or in the case of an Initial Hedge Transfer, the Common Stock), free and clear of all Liens, duly endorsed in blank for transfer, with signature(s) guaranteed, to Alcoa's transfer agent, addressed to: Ms. Randi Galan, Equiserve, 525 Washington Boulevard, 9th Floor, Mail Suite, 4690, Jersey City, New Jersey, USA 07310; and (ii) in the case of Shares (or in the case of an Initial Hedge Transfer, Common Stock) held in

book-entry form, Camargo Corrêa (or in the case of an Initial Hedge Transfer, the Hedging Counterparty) shall duly issue or cause to be duly issued instructions to Equiserve to cause the purchased Shares (or in the case of an Initial Hedge Transfer, Common Stock) to be electronically transferred, free and clear of all Liens, in the direct registration system and posted to Alcoa's or its designated Affiliate's account at Equiserve.

- **2.4.3 Exempt Transfers.** Notwithstanding Section 2.4.2, the Camargo Group and its Affiliates will be entitled to make any of the following Transfers (collectively, the "Exempt Transfers"); provided that in the case of any Exempt Transfer (other than an Exempt Transfer pursuant to Section 2.4.3(b) or 2.4.3(c) (collectively, "Business Combination Transfers") for which no notice is required) Camargo Corrêa shall give Alcoa written notice (an "Exempt Transfer Notice") of a proposed Exempt Transfer at least five Trading Days prior to such Transfer (or in the case of any Exempt Transfer pursuant to Section 2.4.3(g), at least two Trading Days prior to such Transfer, except for Exempt Transfers in relation to the early termination of a Monetization Arrangement, in which case notice shall be given no later than one Trading Day following the occurrence of such Transfer). For the avoidance of doubt, none of the foregoing shall limit the applicability of the final sentence of Section 2.4.3(g). The effective time of an Exempt Transfer Notice shall be determined in accordance with the provisions of Section 2.4.2 relating to the effectiveness of a Proposed Sale Notice. The Exempt Transfers are as follows:
- **2.4.3(a) Transfers to Camargo Corrêa Affiliates** any Transfer of Shares to Camargo Corrêa or to any Affiliate thereof who executes an agreement with the Transferring member of the Camargo Group confirming that any subsequent Transfer of such Shares by such Affiliate that occurs prior to the Restriction Termination Date will be subject to the restrictions and obligations set forth in this Agreement and the Registration Rights Agreement. The Camargo Group shall be responsible for any breaches by its Affiliates. Such agreement(s) shall provide that Alcoa is an express beneficiary thereof and entitled to enforce the provisions thereof, and true and correct copies of such executed agreements shall be provided by Camargo Corrêa to Alcoa prior to the Transfer.
 - **2.4.3(b) Transfers to Alcoa** any Transfer of Shares to Alcoa or an Affiliate thereof, whether pursuant to a tender offer or otherwise.
- **2.4.3(c) Alcoa Business Combinations** any Transfer of Shares in a tender offer, merger or other Alcoa business combination transaction involving the Common Stock.
- 2.4.3(d) Alcoa-Approved Limited Transfers any Transfer (other than, or in relation to, a Monetization Arrangement) of Shares to a Person (excluding the Camargo Group and its Affiliates) who (i) together with its Affiliates, will not own, directly or indirectly, after giving effect to the Transfer, more than 1,800,000 shares of Common Stock and (ii) has been approved by Alcoa; provided that Alcoa shall have the right to withhold such approval only for Good Reason. Not later than 6:00 p.m. New York City Time on the fifth Trading Day after the effective date of an Exempt Transfer Notice in respect of this Section 2.4.3(d), Alcoa shall deliver to the Camargo Group a written notice indicating whether or not it approves of the proposed Exempt Transfer and, if not, specifying with reasonable particularity, the reasons for its position. If such notice is not given by such time, Alcoa shall be deemed to have given the required approval.
- **2.4.3(e) Alcoa-Agreed Restricted Transfers** any Transfer (other than, or in relation to, a Monetization Arrangement) of Shares to a Person (excluding the Camargo Group and its Affiliates) who executes an agreement with Alcoa prior to such Transfer confirming that any subsequent Transfer(s) by such Person or its Affiliates that occur prior to the Restriction Termination Date (including, without limitation, any subsequent Exempt Transfer other than (x) a Business Combination Transfer and (y) any Transfer by such Person to an Affiliate of such

Person who executes a similar agreement with Alcoa) will be subject to the Volume Limitation and the rights of first refusal of Alcoa described in this Section 2.4; provided that, prior to the Restriction Termination Date, the Camargo Group and its Affiliates shall not make more than two Transfers pursuant to this Section 2.4.3(e).

2.4.3(f) Long Term Alcoa Investor – any Transfer of Shares to a Long Term Alcoa Investor (excluding the Camargo Group and its Affiliates).

2.4.3(g) Monetization Arrangements – any Transfer of Shares, directly or indirectly, to a Hedging Counterparty in connection with a forward sale or other derivative transaction between a member or Affiliate of the Camargo Group and such Hedging Counterparty (a "Monetization Arrangement"); provided that, prior to Transferring Shares to any such Hedging Counterparty the member or Affiliate of the Camargo Group shall enter into a contract (an "Assurance Contract") with such Hedging Counterparty assuring that (i) the Hedging Counterparty will not enter into any Monetization Arrangements with other financial institutions or other Persons in connection with establishing its Initial Hedge; (ii) prior to establishing its Initial Hedge, the Hedging Counterparty will grant Alcoa a right of first refusal with respect to the Common Stock to be sold under such Initial Hedge on terms as described in this Section 2.4, (iii) the Hedging Counterparty will not effect any Initial Hedge Transfers, or any Transfers in relation to the early termination of a Monetization Arrangement, on any day which, when aggregated with (x) all Initial Hedge Transfers, or any Transfers in relation to the early termination of a Monetization Arrangement, made or to be made on such day by all other Hedging Counterparties to which a Transfer has been made pursuant to this Section 2.4.3(g) and (y) any Transfers (other than Exempt Transfers) made, or to be made, by any members of the Camargo Group and their Affiliates on such day, exceed the Volume Limitation; (iv) the maturity date of such Monetization Arrangement is such that, after taking into account the maturity dates of all other Monetization Arrangements previously effected pursuant to this Section 2.4.3(g), there shall not mature, on any single day, Monetization Arrangements that, in the aggregate could result in sales of Common Stock on any day in excess of the Volume Limitation; and (v) the Monetization Arrangement will not mature prior to the first anniversary of the Closing Date. The Camargo Group shall: (i) ensure that such Assurance Contract assures compliance with clauses (i) through (v) of the foregoing sentence, (ii) ensure that such Assurance Contract provides that Alcoa is an express beneficiary thereof and is entitled to enforce the provisions thereof, (iii) provide true and correct copies of the Assurance Contract to Alcoa before the execution thereof and at least two Trading Days prior to the Transfer, (iv) provide true and correct copies of such executed Assurance Contract and, if requested, of the executed documentation with respect to the Monetization Arrangement to Alcoa prior to the Transfer, and (v) deliver to Alcoa prior to the Transfer a certificate of an officer of Camargo Corrêa certifying that the Assurance Contract is in accordance with the applicable provisions of this Agreement.

2.4.3(h) Pledges – any Transfer (other than, or in relation to, a Monetization Arrangement) of Shares to a bank or other financial institution as security for a loan or other financing or any Transfer of Shares to such bank or financial institution upon foreclosure, so long as such bank or financial institution executes an agreement with the Transferring member or Affiliate of the Camargo Group confirming that any subsequent Transfer by it or its Affiliates that occurs prior to the Restriction Termination Date (including, without limitation, any subsequent Exempt Transfer other than a Business Combination Transfer but excluding any Transfer by such bank or other financial institution to an Affiliate of such bank or financial institution that executes a similar agreement with the Transferring member or Affiliate of the Camargo Group) of such Shares will be subject to the Volume Limitation and to the rights of first refusal of Alcoa on terms as described in this Section 2.4. The Camargo Group shall: (i) ensure that such contract is in accordance with the applicable provisions of this Agreement,

(ii) ensure that such contract provides that Alcoa is an express beneficiary thereof and is entitled to enforce the provisions thereof, (iii) provide true and correct copies of the contract to Alcoa before the execution thereof and at least two Trading Days prior to the Transfer, and (iv) provide true and correct copies of such executed contract to Alcoa prior to the Transfer.

- **2.4.4** Use of Shelf Registration Statement by Transferees; Persons Eligible to be Permitted Sellers. Other than assignments to Affiliates of the Camargo Group, the Camargo Group and its Affiliates agree that the rights and obligations under the Registration Rights Agreement may only be assigned to a Transferee acquiring Shares in a Transfer if such Transfer cannot reasonably be accomplished under the Shelf Registration Statement or under Rule 144 of the Securities Act by the proposed closing date of the proposed Transfer. Furthermore, the parties agree that the number of Transferees who may be assignees of the Registration Rights Agreement and thus eligible to be Permitted Sellers (as such term is defined in the Registration Rights Agreement) is subject to the following limitations:
 - (a) no more than two (2) Transferees pursuant to each of Sections 2.4.3(d), 2.4.3(e), 2.4.3(f), 2.4.3(g), and 2.4.3(h); and
 - (b) no more than eight (8) Transferees (other than the Camargo Group and its Affiliates) in the aggregate at any time.

Furthermore, each such Transferee (other than the Camargo Group and its Affiliates) shall be eligible to reassign in whole its rights and obligations under the Registration Rights Agreement only once to one subsequent Transferee of the Shares (who shall have no further assignment rights) if such subsequent Transfer cannot reasonably be accomplished under the Shelf Registration Statement or under Rule 144 of the Securities Act by the proposed closing date of the proposed Transfer and provided that under all circumstances the total number of Transferees (other than the Camargo Group and its Affiliates) eligible at any given time to be Permitted Sellers under the Registration Rights Agreements is subject to the limitation set forth in (b) above. The Camargo Group and its Affiliates shall: (i) give prior written notice to Alcoa in accordance with Section 7.6 of any assignment by it of the rights and obligations of the Registration Rights Agreement hereunder (which notice may be included as part of the Proposed Sale Notice); (ii) ensure that the assignee agrees to all restrictions and obligations imposed under the Registration Rights Agreement; (iii) ensure that the assignee agrees to the provisions of this Section 2.4.4, including those relating to any reassignment by it, and agrees to give prior written notice to Alcoa in accordance with the provisions of Section 7.6 of any such reassignment. Any such assignment shall not be effective unless the proposed Transferee executes and delivers a written instrument reasonably satisfactory to Alcoa whereby such proposed Transferee agrees to be bound by the restrictions and obligations under the Registration Rights Agreement and the limitations on assignment of the Registration Rights Agreement contained in this Section 2.4.4.

2.4.5 Restrictions on Transferees. Upon consummation of any Transfer of Shares prior to the Restriction Termination Date, the Person to whom the Transfer is made (the "Transferee") will be subject to, in addition to any restrictions or obligations imposed under the Registration Rights Agreement, if applicable to such Transferee, and the Securities Act, the following restrictions on the subsequent Transfer of such Shares:

Type of Transferee

Restrictions on subsequent Transfer

2.4.5(a) If the Transferee obtains the Shares pursuant to Section 2.4.3(a) (Transfers to Camargo Corrêa Affiliates), then, whether or not the Transfer of Shares has been registered under the Shelf Registration Statement, the

the restrictions set forth in this Agreement, including the Volume Limitation and rights of first refusal described in Section 2.4.

Transferee (and any subsequent Transferee acquiring Shares under such section) will be subject to:

2.4.5(b) If the Transferee obtains the Shares pursuant to Section 2.4.3(b) (Transfers to Alcoa), the Transferee will be subject to:

2.4.5(c) If the Transferee obtains the Shares pursuant to Section 2.4.3(c) (Alcoa Business Combinations), the Transferee will be subject to:

2.4.5(d) If the Transferee obtains the Shares pursuant to Section 2.4.3(d) (Alcoa-Approved Limited Transfers), the Transferee will be subject to:

2.4.5(e) If the Transferee obtains the Shares pursuant to Section 2.4.3(e) (Alcoa-Agreed Restricted Transfers), then the Transferee (and any subsequent Transferee acquiring Shares under such section) will be subject to:

2.4.5(f) If the Transferee obtains the Shares pursuant to Section 2.4.3(f) (Alcoa Long Term Investor), the Transferee will be subject to:

2.4.5(g) If the Transferee is a Hedging Counterparty that obtains the Shares pursuant to Section 2.4.3(g) (Monetization Arrangements), then:

2.4.5(h) If the Transferee obtains the Shares pursuant to Section 2.4.3(h) (Pledges), whether as collateral or upon foreclosure, then, the Transferee (and any subsequent Transferee acquiring Shares under such section) will be subject to:

2.4.5(i) Provided that the Transfer was not in violation of any provision of this Agreement or the Registration Rights Agreement, if the Transferee acquires the Shares in a transaction properly effected on a national securities exchange (whether effected pursuant to Rule 144 under the Securities Act or otherwise) or in a registered public offering, the Transferee will be subject to:

2.4.5(j) Provided that the Transfer was not in any violation of any provision of this Agreement or the Registration Rights

no transfer restrictions.

no transfer restrictions.

no transfer restrictions.

the restrictions set forth in this Agreement, including the Volume Limitation and rights of first refusal described in Section 2.4.

no transfer restrictions.

(i) the Transferee and its Affiliates will be subject to the restrictions set forth in this Agreement, including the Volume Limitation and rights of first refusal described in Section 2.4, with respect to, and only with respect to, Initial Hedge Transfers;

(ii) the Transferee and its Affiliates will be subject only to the Volume Limitation (and will not be subject to the rights of first refusal) described in Section 2.4 with respect to sales of the Shares or Common Stock made at the maturity or earlier termination of the Monetization Arrangement; and (iii) the Transferee and its Affiliates will not be subject to the Volume Limitation and the rights of first refusal described in Section 2.4 in connection with any other transactions involving the Shares or Common Stock, including on-going sales of the Shares or Common Stock made by the Hedging Counterparty to manage its exposure under the Monetization Arrangement.

the restrictions set forth in this Agreement, including the Volume Limitation and rights of first refusal described in Section 2.4.

no transfer restrictions.

no transfer restrictions. For the avoidance of doubt, any such Transferee shall not be eligible to be a Permitted Seller, as such term is defined in the Registration Rights Agreement, if the Transferee acquires Shares in a transaction that is not an Exempt Transfer after Alcoa waived its right of first refusal with respect to such Shares as provided in Section 2.4.2, the Transferee will be subject to:

Agreement.

2.5 Certain Alcoa Alumínio and ALAHC Matters. The parties agree that the total amount of the Camargo Group's dividend distributions from Alcoa Alumínio and ALAHC relating to 2002 and a portion of 2003 will be R\$ 31,688,230 from Alcoa Alumínio and U.S.\$1,146,068 from ALAHC. The remittance of the dividend distributions: (a) from Alcoa Alumínio shall be made prior to the transfer to the Alcoa Group of the BACEN certificates of registration as set forth in Section 3.3.3 of this Agreement; and (b) from ALAHC shall be made prior to Closing. Following the Closing, the Camargo Group shall not be entitled to receive any additional dividend distributions from Alcoa Alumínio and ALAHC. During the Earn-Out Period after the Closing, the Alcoa Group agrees that the Camargo Group may have one observer attend meetings of the board of directors (i.e. the Conselho de Administração) of Alcoa Alumínio for so long as such body is maintained and retains its current scope of authority. The Alcoa Group reserves the right to change or eliminate such body; provided that, in the event that the board of directors is replaced by a consulting board or other body having similar functions, the Camargo Group will be entitled to have an observer attend the meetings of such board or other body. Unless the parties agree otherwise, the Camargo Group's observer shall continue to be provided with, with respect to the period covered by the Earn-Out Period: (i) the Alcoa Latin America Financial Report ("Blue Book") prepared and circulated monthly; and (ii) a reconciliation between financial statements included in each December edition of the Blue Book with the numbers included in the calculation of the EBITDA for the year (whether or not provided to the board of directors, consulting board or other similar body). If the board of directors is eliminated without being replaced by a consulting board or other body having similar functions, the Camargo Group's observer shall be provided with, with respect to the period covered by the Earn-Out Period, and in addition to the documents specified in the previous sentence: (i) reports prepared by each business unit for review and discussion with the Alcoa Inc. senior executives as part of the quarterly review process; and (ii) access on a bi-monthly basis to the Alcoa Alumínio chief financial officer. In any event, the observer shall be a non-voting observer and shall not participate in management or in the formulation, determination, or direction of basic business decisions, but the observer shall receive as to content and timing the same information and material relating to the business of Alcoa Alumínio as is provided to the other members of such board or body. The information provided to the Camargo Group's observer shall be subject to the confidentiality provisions of Section 7.9. The Camargo Group's initial observer is Francisco Caprino Neto. Further, the Camargo Group represents that it has no present intention of participating in the formulation, determination or direction of the basic business decisions of Alcoa and does not intend to nominate a candidate for Alcoa's board of directors, propose corporate action requiring shareholder approval, or solicit proxies. Notwithstanding the above, the Camargo Group may vote its shares in Alcoa.

2.6 <u>Adjustments to Number of Shares</u>. If, during the period between Closing and the Restriction Termination Date, there is a pro-rata change in the number of outstanding shares of Common Stock as a result of a stock split, reverse stock split, or similar Alcoa corporate action that results in a pro rata adjustment of the market price of the shares of Common Stock, then Alcoa will adjust accordingly the number of shares and per share prices used in the calculations

in **Exhibit C** (e.g., the Threshold Share Price) and the number of Shares set forth in Section 2.7.3(ii). If such an event occurs during the period between the date of this Agreement and Closing, then in addition to the adjustments described in the foregoing sentence, the parties shall also adjust the number of Shares accordingly and shall amend this Agreement before Closing to reflect the adjusted amounts. Notwithstanding anything in this Agreement, for the avoidance of doubt the parties hereby confirm that the number of Shares delivered hereunder and the Earn-Out Payments calculations set forth in **Exhibit C** shall not be adjusted to reflect changes in the value of Alcoa Alumínio Stock, the ALAHC Stock or the Common Stock after the date hereof, whether due to (i) any change or effect that is materially adverse to business, operations, property, assets, liabilities or the financial or operational condition of Alcoa Alumínio, ALAHC or Alcoa, (ii) changes or conditions affecting the industries in which Alcoa Alumínio, ALAHC or Alcoa operate (including metals or aluminum pricing), (iii) changes in economic, regulatory or political conditions generally (including acts of terrorism or war) or (iv) otherwise.

2.7 Special Option During First Year After the Closing Date. Subject to the limitations set forth in Section 2.7.3 below, if Camargo Corrêa delivers, at any time before the first anniversary of the Closing a Shelf Notice of Sale pursuant to Section 4.2(g) of the Registration Rights Agreement and corresponding Proposed Sale Notice pursuant to Section 2.4.2 of this Agreement and a Delay Period is in effect with respect to the Transfer described in such notices, Alcoa and the Camargo Group shall comply with procedure I below ("Procedure I"), unless Alcoa includes a statement in its Delay Period Notice given pursuant to Section 4.2(g) of the Registration Rights Agreement that procedure II below ("Procedure II") shall apply in respect of such Delay Period instead of Procedure I.

2.7.1 Procedure I.

- (i) **Disclosure of Confidential Information**. Alcoa shall provide to Camargo Corrêa by the Delay Period Notice Deadline a description in reasonable detail of the non-public information that is the reason for the Delay Period (the "Confidential Information"). The receipt of the Confidential Information by Camargo Corrêa shall be subject to the confidentiality agreement set forth in Section 7.9 of this Agreement. In addition, and without limiting the foregoing, each member of the Camargo Group agrees during the Delay Period to maintain the Confidential Information in strict confidence and not to disclose it to any third party or trade in any Alcoa securities.
- (ii) **Rescission Right.** Upon receipt of the Confidential Information in accordance with the immediately preceding clause (i), Camargo Corrêa shall have a right to rescind the previously delivered Shelf Notice of Sale and corresponding Proposed Sale Notice by so notifying Alcoa in writing no later than 9:00 a.m., New York City Time, on the Trading Day next succeeding the Delay Period Notice Deadline (the "Procedure I Rescission Deadline"). If Camargo Corrêa rescinds its Shelf Notice of Sale and corresponding Proposed Sale Notice, then the Camargo Group and its Affiliates may not deliver additional Shelf Notices of Sale or corresponding Proposed Sale Notices until such Delay Period is no longer in effect. If Camargo Corrêa does not rescind its Shelf Notice of Sale, then it may deliver additional notice(s) during such Delay Period but the Camargo Group shall not have rescission rights with respect to such additional notice(s) unless there is an additional disclosure of new or different Confidential Information pursuant to such additional notice, in which case the provisions of this subsection (ii) shall apply.
- (iii) **Purchase of Shares and Settlement Procedures.** Unless the Shelf Notice of Sale and corresponding Proposed Sale Notice are rescinded by Camargo Corrêa by the

Procedure I Rescission Deadline, the Camargo Group shall be deemed to have waived its rescission right with respect to such Shelf Notice of Sale and corresponding Proposed Sale Notice and Alcoa shall be obligated to purchase the Shares indicated in such Shelf Notice of Sale and corresponding Proposed Sale Notice, subject to the limitations set forth in Section 2.7.3 below, at the Average Price for the applicable Trading Day as determined in accordance with Section 2.4.2 of this Agreement. The settlement procedures applicable to the purchase of such Shares shall be as provided in Section 2.4.2 of this Agreement.

2.7.2 Procedure II.

- (i) **Deferred Disclosure of Confidential Information**. Alcoa may defer disclosing the Confidential Information to Camargo Corrêa beyond the Delay Period Notice Deadline until (x) the date that such Confidential Information is publicly disclosed (the "Delay Period Termination Date") or (y) such other date during the Delay Period as Alcoa is able or deems it appropriate to disclose such information to Camargo Corrêa. If Alcoa discloses the Confidential Information to Camargo Corrêa before the information is publicly disclosed, Camargo Corrêa's receipt of such Confidential Information shall be subject to the confidentiality agreement set forth in Section 7.9 of this Agreement. In addition, without limiting the foregoing, each member of the Camargo Group agrees to maintain the Confidential Information in strict confidence and not to disclose it to any third party or trade in any Alcoa securities.
- (ii) **Rescission Right.** Once the Confidential Information is disclosed to Camargo Corrêa in the manner described in the immediately preceding clause (i), Camargo Corrêa shall have a right to rescind all (but not less than all) Shelf Notices of Sale and corresponding Proposed Sale Notices theretofore delivered by it during such Delay Period by notifying Alcoa in writing no later than 9:00 a.m., New York City Time, on the earlier of (x) the Trading Day next succeeding the Delay Period Termination Date or (y) the Trading Day next succeeding Camargo Corrêa's receipt of the Confidential Information (the "Procedure II Rescission Deadline").
- (iii) **Purchase of Shares and Settlement Procedures.** Unless all such Shelf Notices of Sale and corresponding Proposed Sale Notices are rescinded by Camargo Corrêa by the applicable Procedure II Rescission Deadline as provided in the immediately preceding clause (ii), the Camargo Group shall be deemed to have waived its rescission right with respect to all such Shelf Notices of Sale and corresponding Proposed Sale Notices and Alcoa shall be obligated to do the following after the Delay Period Termination Date, as Alcoa elects for each Shelf Notice of Sale and corresponding Proposed Sale Notice in its sole discretion: (x) to purchase all of the Shares indicated in each such Shelf Notice of Sale and corresponding Proposed Sale Notice, subject to the limitations set forth in Section 2.7.3 below, at the Average Price for the applicable Trading Day as determined in accordance with Section 2.4.2 of this Agreement, or (y) subject to compliance with applicable law, to arrange for the purchase of all of the Shares indicated in each Shelf Notice of Sale and corresponding Proposed Sale Notice by a third party on such terms as the parties shall agree, provided that Alcoa shall pay Camargo Corrêa the difference, if any, between the price obtained for the Shares under such arrangement and the price of such Shares as determined by reference to the applicable Average Price per Share calculated under the formula set forth in the immediately preceding sub-clause (x). If Alcoa purchases the Shares under sub-clause (x), the settlement procedures applicable to the purchase of such Shares shall be as provided in Section 2.4.2 of this Agreement, except that the payment for and delivery of the Shares shall be as agreed by the parties but in no event shall payment for and delivery of the Shares be made later than on the tenth Trading Day after the Delay Period

Termination Date. For clarification, if more than one Shelf Notice of Sale and corresponding Proposed Sale Notice are outstanding, Alcoa may elect subclause (x) with respect to some Shelf Notices of Sale and corresponding Proposed Sale Notices and subclause (y) with respect to others.

- 2.7.3 Limitations on Special Option. Notwithstanding anything to the contrary contained in this Agreement, the parties agree that:
- (i) Section 2.7 is applicable only with respect to Shelf Notices of Sale (and corresponding Proposed Sale Notices) that become effective during the first year after the Closing Date, and Alcoa shall have no obligation to purchase (or to cause the purchase of) any Shares covered by such notices that become effective on or after the first anniversary date of the Closing Date (the "Special Option Expiration Date").
- (ii) Alcoa shall not be required to purchase (or to cause the purchase of) more than 900,000 Shares pursuant to any single Shelf Notice of Sale and corresponding Proposed Sale Notice nor more than 8,886,770 Shares in the aggregate pursuant to all Shelf Notices of Sale and corresponding Proposed Sale Notices that become effective before the Special Option Expiration Date.
- (iii) The rights and obligations of the Camargo Group under this Section 2.7 are personal to the Camargo Group and its Affiliates and may not be assigned or otherwise transferred to any other Person. For the avoidance of doubt, no Permitted Seller (as defined in the Registration Rights Agreement) or other Person who is not an Affiliate of the Camargo Group shall have any right or benefit under such subsections 2.7.1 or 2.7.2.
- 2.8 **Counsel Opinions for Transfers Other Than Under the Shelf Registration.** Until the second anniversary of the Closing Date, prior to making any Transfer of Shares to any Person, other than a Transfer registered under the effective Shelf Registration Statement, Camargo Corrêa will give written notice to Alcoa of the intention by a member or Affiliate of the Camargo Group to effect such Transfer. Each such notice shall describe the manner and circumstances of the proposed Transfer. Unless the proposed Transfer is a Business Combination Transfer as set forth in Sections 2.4.3(b) or 2.4.3(c), if within 5 Trading Days (or 2 Trading Days in the case of Transfers pursuant to Section 2.4.3(g)) after receipt by Alcoa of such notice, Alcoa requests an opinion of counsel for such member or Affiliate of the Camargo Group that the proposed Transfer may be effected without registration of the Transfer under the Securities Act (such counsel opinion to be reasonably satisfactory to legal counsel for Alcoa), then Alcoa shall not be required to cause its share transfer agent to record such Transfer, and the Camargo Group and its Affiliates shall not be entitled to effect such Transfer, unless and until Alcoa receives such a counsel opinion reasonably satisfactory to Alcoa.

ARTICLE III CLOSING

- 3.1 <u>Time and Location</u>. Subject to the satisfaction or waiver of the conditions contained herein, the Closing shall be held on the Closing Date at 9:00 a.m., New York City Time, at the offices of Alcoa at 390 Park Avenue, New York, New York, United States of America, or such other time or place as the parties mutually agree in writing.
- 3.2 <u>Documents to be Delivered or Actions to be Taken by the Alcoa Group</u>. At the Closing, the Alcoa Group shall deliver or cause to be delivered:

- 3.2.1 A share certificate representing 10,366,471 shares of Alcoa Common Stock, duly registered in the name of Allpar and share certificate(s) representing 7,407,069 shares of Alcoa Common Stock duly registered in the name of Trelawney, in each case, free and clear of all Liens other than the Permitted Encumbrances; provided that, if so requested by the Camargo Group at least five business days before the Closing, in lieu of delivery of share certificates, the Alcoa Group will make delivery by means of registering the transfer, free and clear of all Liens other than the Permitted Encumbrances, of 10,366,471 shares of Alcoa Common Stock to Allpar and 7,407,069 shares of Alcoa Common Stock to Trelawney in book-entry form registered on the books of Equiserve.
- 3.2.2 Executed copies of the Termination of Shareholders Agreement documents substantially in the forms set forth on **Exhibits D-1 and D-2** as provided for in Section 3.5.
- 3.2.3 A favorable opinion or opinions of counsel, reasonably satisfactory to the Camargo Group, as to the matters set forth in Sections 4.1.1 and 4.1.3(a) and (c) hereof.
- 3.2.4 Such other documents, such as certificate(s) of incumbency and copies of articles of incorporation and by-laws or similar corporate documents, as the Camargo Group may reasonably request.
- 3.3 <u>Documents to be Delivered or Actions to be Taken by the Camargo Group</u>. At the Closing, the Camargo Group shall deliver or cause to be delivered:
- 3.3.1 The Alcoa Alumínio Stock, free and clear of all Liens, for transfer to AL from Allpar, Trelawney and the individuals as set forth in **Exhibit A** and Section 2.1 above. Such transfer shall be accomplished by (i) the authorized representative(s) or lawful attorneys-in-fact of Allpar and Trelawney and (ii) the individuals set forth on **Exhibit A** and Section 2.1, or the lawful attorneys-in-fact of such individuals, signing the corresponding entries in Alcoa Alumínio's book of transfer of nominative shares to register and execute the transfer of title of such shares on the Closing Date and delivering at the Closing (x) executed copies of such corresponding entries together with copies of the entries in Alcoa Alumínio's registry book of nominative shares based on the transfer of title of such shares, (y) certificates of incumbency in the form agreed by the Camargo Group and the Alcoa Group for the authorized representatives of Allpar and Trelawney, and (z) for any attorneys-in-fact signing on behalf of Allpar, Trelawney, or any of the individuals set forth on **Exhibit A** and Section 2.1, duly legalized powers of attorney in the form set forth on **Exhibit B-2** or such other form agreed by the Camargo Group and the Alcoa Group.
- 3.3.2 The share certificates representing the ALAHC Stock, free and clear of all Liens, duly endorsed by Trelawney in blank for transfer, together with a duly executed deed of transfer of shares in substantially the form set forth on **Exhibit B-1**.
- 3.3.3 The Consolidated Statement of Foreign Capital Investment in Brazil ("Extrato Consolidado de Investimento Externo Direto no Brasil") in connection with the corresponding Electronic Declaratory Registration ("Registro Declaratório Eletrônico") RDE issued by SISBACEN ("Sistema Eletrônico do Banco Central") prior to the transfer with BACEN mentioned in Section 3.3.4, indicating that Allpar, in its capacity of foreign investor, owns 1,705,985 common shares representative of 28.38% of the paid-in capital of Alcoa Alumínio in the total amount of U.S.\$163,725,371.50, which constitutes a portion of the paid-in capital of

Alcoa Alumínio in the amount of R\$197,316,041.62 and the Consolidated Statement of Foreign Capital Investment in Brazil ("Extrato Consolidado de Investimento Externo Direto no Brasil") in connection with the corresponding Electronic Declaratory Registration ("Registro Declaratório Eletrônico")—RDE—issued by SISBACEN ("Sistema Eletrônico do Banco Central") prior to the transfer with BACEN mentioned in Section 3.3.4, indicating that Trelawney, in its capacity of foreign investor, owns 745,014 common shares representative of 12.39% of the paid-in capital of Alcoa Alumínio in the total amounts of U.S.\$78,172,116.70, Swiss Francs 4,158,031.86 and Panama Balboa 512,861.55, which constitutes a portion of the paid-in capital of Alcoa Alumínio in the amount of R\$86,169.112.53.

- 3.3.4 The Consolidated Statements of Foreign Capital Investment in Brazil ("Extratos Consolidado de Investimento Externo Direto no Brasil") in connection with the corresponding Electronic Declaratory Registration ("Registro Declaratório Eletrônico")—RDE—issued by SISBACEN ("Sistema Eletrônico do Banco Central") evidencing the transfer on the Closing Date of the registrations with BACEN of the Alcoa Alumínio shares set forth in Section 3.3.3 above to Alcoa, AL and/or their designated U.S. Person Affiliate(s). The Alcoa Group will confirm to the Camargo Group at least five days prior to Closing the names of its designated transferee(s) if such transferee(s) are other than AL.
- 3.3.5 Executed copies of the Termination of Shareholders Agreement documents substantially in the forms set forth on **Exhibits D-1 and D-2** as provided for in Section 3.5.
- 3.3.6 Written resignations of Fernando Tigre de Barros Rodrigues, Edy Luiz Kogut and Daniel Antonio Biondo Bastos from their respective positions as directors (i.e. members of the council of administration) of Alcoa Alumínio.
- 3.3.7 Written resignations of Fernando de Arruda Botelho, Carlos Pires Oliveira Dias and Luiz Roberto Ortiz Nascimento from their respective positions as directors and executive council members of ALAHC.
- 3.3.8 A favorable opinion or opinions of counsel, reasonably satisfactory to the Alcoa Group, as to the matters set forth in Sections 4.2.1, 4.2.3(a) and (c), and 4.2.5 hereof, including, under Section 4.2.3(c), the enforceability in Brazil of Section 7.16.
- 3.3.9 Such other documents, such as certificate(s) of incumbency and copies of articles of incorporation and by-laws or similar corporate documents, as the Alcoa Group may reasonably request.
- 3.4 <u>Update of the RDE at SISBACEN</u>. Notwithstanding delivery of the documents set forth in Sections 3.2 and 3.3 above, the Closing shall only be deemed effective and title to the Shares pass to and vest in the Camargo Group upon the Alcoa Group's receipt at the Closing of the documents set forth in Section 3.3.4 showing foreign registered capital in Alcoa Alumínio in the amounts set forth in Section 3.3.3 above and as represented in Section 4.2.5. The Camargo Group acknowledges that the transfer of foreign investment registrations with BACEN, as described in Sections 3.3.3 and 3.3.4 and as represented in Section 4.2.5, to Alcoa, AL and/or their designated U.S. Person Affiliate(s) is a condition for the Alcoa Group to consummate the transactions contemplated hereby. Accordingly, the Camargo Group shall take all actions required under regulations issued by BACEN to be taken by a foreign investor in order to cause

the transfer of foreign investment registrations with BACEN, as described in Sections 3.3.3 and 3.3.4, to Alcoa, AL and/or their designated U.S. Person Affiliate(s). The Alcoa Group shall take and shall cause Alcoa Alumínio to take all actions, along with the Camargo Group, necessary to be taken by a foreign investor and/or by a Brazilian company having foreign shareholders in order to supply all the information that may be required by BACEN with a view of transferring the foreign investment registrations with BACEN, as described in Sections 3.3.3 and 3.3.4, to Alcoa, AL and/or their designated U.S. Person Affiliate(s).

- 3.5 Termination of Shareholders Agreements. Upon the Closing, the following shareholder agreements shall terminate without liability or claims:
- 3.5.1 the Shareholders Agreement among Alcoa Brazil Holdings Company, Construções e Comércio Camargo Corrêa S.A., Trelawney and Allpar dated March 28, 1995, as amended, regarding Alcoa Alumínio; and
- 3.5.2 the ALAHC Shareholders Agreement among Alcoa, Construções e Comércio Camargo Corrêa S.A., Trelawney and ALAHC dated March 15, 1995, as amended, regarding ALAHC.

ARTICLE IV REPRESENTATIONS AND WARRANTIES

- 4.1 <u>Representations and Warranties of the Alcoa Group</u>. Each member of the Alcoa Group, jointly and severally, represents and warrants to each member of the Camargo Group that:
- 4.1.1. **Organization; Authorization.** Each member of the Alcoa Group is duly organized, validly existing and in good standing under the laws of its place of incorporation or organization, and has full corporate power and authority to enter into this Agreement and the Registration Rights Agreement, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. Each member of the Alcoa Group has taken all corporate action necessary to authorize the execution and delivery by it of this Agreement and the Registration Rights Agreement, the performance by it of its obligations hereunder and thereunder, and the consummation by it of the transactions contemplated hereby and thereby. This Agreement and the Registration Rights Agreement have been duly executed and delivered by each member of the Alcoa Group and constitute the legal, valid and binding obligations of each member of the Alcoa Group enforceable against such member of the Alcoa Group in accordance with their respective terms, except as limited by applicable bankruptcy, insolvency, reorganization, moratorium or other laws of general application relating to or affecting enforcement of creditors' rights and rules of law concerning equitable remedies.
- 4.1.2 **Shares.** Upon delivery at the Closing in accordance with the terms of this Agreement, the Shares will be validly issued and outstanding, fully paid and non-assessable and will be free and clear of any claim, lien, pledge, option, charge, security interest, encumbrance or any other rights of third parties ("Liens") other than the Permitted Encumbrances.
- 4.1.3 **No Violation.** Neither the execution and delivery by any member of the Alcoa Group of this Agreement or the Registration Rights Agreement nor the consummation of the

transactions contemplated hereby or thereby will result in: (a) a violation of or a conflict with any provision of the articles of incorporation or bylaws (or other comparable corporate charter documents) of any member of the Alcoa Group; (b) a breach of, or a default (or give rise to any right of termination, cancellation or acceleration) under, any term or provision of any contract, agreement, indebtedness, lease, commitment, license, franchise, permit, authorization or concession to which it is a party or by which it is bound; or (c) a violation of any law, statute, rule, regulation, ordinance, code, order, judgment, writ, injunction, decree or award applicable to it, excluding in the foregoing clauses (b) and (c) such violations, breaches or defaults which are not reasonably likely to have a Material Adverse Effect. There is no pending or, to the knowledge of any member of the Alcoa Group, threatened proceeding by or before any court, arbitrator or governmental authority against any member of the Alcoa Group which in any manner challenges or seeks to prevent, enjoin, alter or materially delay the transactions contemplated by this Agreement or the Registration Rights Agreement, and, to the knowledge of any member of the Alcoa Group, there exists no basis for any such proceeding.

- 4.1.4 **Governmental Consent, etc.** Except for obtaining the approval of CADE and except for compliance with the requirements of the Securities Act in connection with the Shelf Registration Statement and compliance with applicable notice requirements of state securities laws, no consent, approval or authorization of, or declaration, filing or registration with, or any action by, any governmental or regulatory authority, whether domestic or foreign, or any other Person, is required to be made or obtained by any member of the Alcoa Group in connection with the execution, delivery and performance by it of this Agreement and the Registration Rights Agreement and, except for the transfer of foreign investment registrations as described in Sections 3.4 and 5.2.5, the consummation of the transactions contemplated hereby or thereby.
 - 4.1.5 **Status of AL.** AL is a wholly-owned indirect subsidiary of Alcoa.
- 4.1.6 Capitalization. The authorized capital stock of Alcoa consists of 1,800,000,000 shares of Common Stock, 557,740 shares of \$3.75 Cumulative Preferred Stock, par value \$100.00 per share ("Alcoa Serial Preferred Stock") and 10,000,000 shares of Class B Serial Preferred stock, par value \$1.00 per share ("Alcoa Class B Serial Preferred Stock"). As of June 30, 2003, (i) 846,051,542 shares of Common Stock were issued and outstanding; (ii) 92,469,104 shares of Common Stock were subject to outstanding options issued pursuant to Alcoa's existing stock incentive plans (the "Alcoa Incentive Plans"), and 105,990,141 shares of Common Stock were reserved for issuance under Alcoa's employees savings plans or fee continuation plan for non-employee directors; (iv) no shares of Common Stock were reserved for issuance under Alcoa's incentive compensation plan; (v) 78,552,996 shares of Common Stock were issued and held in the treasury of Alcoa; (vi) 546,024 shares of Alcoa Serial Preferred Stock were issued and outstanding; and (vii) no shares of Alcoa Class B Serial Preferred Stock were issued and outstanding.
- 4.1.7 **Offering of Securities.** No member of the Alcoa Group nor any Person acting on their behalf has offered the Shares to, or solicited any offers to buy any of the Shares from, or otherwise approached or negotiated with respect thereto with, any Person or Persons by any form of general solicitation or general advertising or in any other manner as would subject the offering, issuance, delivery or exchange of any of the Shares to the provisions of Section 5 of the Securities Act. No member of the Alcoa Group nor any Person acting on their behalf has taken or will take any action that would subject the offering, issuance, delivery or exchange of any of

the Shares to the provisions of Section 5 of the Securities Act. Subject to the accuracy of each of the Camargo Group's representations and warranties in Section 4.2 of this Agreement, the offer, issuance, delivery and exchange of the Shares constitute transactions exempt from the registration requirements of Section 5 of the Securities Act and from the registration or qualification requirements of any applicable United States state or jurisdiction (other than notice filings required under applicable law).

- 4.1.8 **Alcoa SEC Reports.** Alcoa has previously furnished or otherwise made available to the Camargo Group true and complete copies of: (i) Alcoa's Annual Reports on Form 10-K filed with the SEC for each of the fiscal years ended December 31, 2001 and December 31, 2002; (ii) Alcoa's Quarterly Reports on Form 10-Q filed with the SEC since January 1, 2002 to the date of this Agreement; (iii) Alcoa's Current Reports on Form 8-K filed with the SEC since January 1, 2002 to the date of this Agreement; (iv) each definitive proxy statement filed by Alcoa with the SEC since January 1, 2002 to the date of this Agreement, except any final prospectus on Form S-8 relating to employee benefit plans. As of their respective dates, such reports, proxy statements and prospectuses (collectively with any amendments, supplements and exhibits thereto, the "Alcoa SEC Reports") complied as to form in all material respects with the applicable requirements of the Securities Act, the Exchange Act and the rules and regulations promulgated thereunder and did not contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading. Since January 1, 2002, Alcoa has timely made all filings required to be made by it with the SEC under the rules and regulations of the SEC.
- 4.1.9 **Financial Statements.** The audited consolidated financial statements and unaudited consolidated interim financial statements of Alcoa included in the Alcoa SEC Reports (including any related notes and schedules) filed with the SEC prior to the date of this Agreement fairly present, in all material respects, the financial position of Alcoa and its consolidated subsidiaries as of the dates thereof and the results of operations and cash flows for the periods or as of the dates then ended (subject, in the case of the unaudited interim financial statements, to normal year-end adjustments), in each case in accordance with GAAP consistently applied during the periods involved (except as otherwise disclosed in the notes thereto).
- 4.1.10 **Absence of Certain Changes or Events.** Except as disclosed in the Alcoa SEC Reports, since the date of Alcoa's latest financial statements included in the Alcoa SEC Reports, (a) Alcoa has conducted its business in the ordinary course and (b) there has not been any event, occurrence, development or state of circumstances or facts that has had, or is reasonably likely to have, a Material Adverse Effect.
- 4.1.11 **No Undisclosed Liabilities.** Except those which are not reasonably likely to have a Material Adverse Effect, there are no liabilities or obligations of any nature required to be set forth in a consolidated balance sheet of Alcoa under GAAP, and there is no existing condition, situation or set of circumstances which could reasonably be expected to result in such a liability or obligation, except (a) liabilities or obligations reflected in the Alcoa SEC Reports filed prior to the date hereof, and (b) liabilities or obligations incurred in the ordinary course of business consistent with past practice since the date of Alcoa's latest financial statements included in the Alcoa SEC Reports.

- 4.1.12 **Litigation.** Except as set forth in the financial statements referred to in Section 4.1.9 or disclosed in the SEC Reports, there is no pending or, to the knowledge of Alcoa, threatened proceeding by or before any court, arbitrator or governmental authority against Alcoa which in the opinion of Alcoa's counsel is likely to result in a Material Adverse Effect.
- 4.1.13 **Tax Returns.** Alcoa and its consolidated subsidiaries have filed or caused to be filed all federal, state, local and foreign tax returns required to have been filed by it and have paid or caused to be paid all taxes shown to be due and payable on such returns or on any assessments received by it, except taxes that are being contested in good faith by appropriate proceedings and for which adequate reserves are maintained in accordance with GAAP and except for matters which are not likely to result in a Material Adverse Effect.
- 4.1.14 **Base EBITDA**. To the knowledge of Alcoa, the Base EBITDA calculation amounts as determined by the methodology described in **Exhibit C** were derived from U.S. Dollar financial information prepared from the books and records of Alcoa Alumínio and ALAHC. Such information was prepared in all material respects in accordance with Alcoa's accounting policies and procedures applied on a consistent basis, which policies and procedures complied in all material respects with GAAP. For the avoidance of doubt, the Base EBITDA calculation amounts in **Exhibit C** have been agreed by the parties and shall not be adjusted absent proof of intentional material misstatement by Alcoa.
- 4.1.15 **No Finder or Broker.** No member of the Alcoa Group has retained any finder or broker in connection with the transactions contemplated by this Agreement or the Registration Rights Agreement.
 - 4.1.16 No Other Representations. The Alcoa Group makes no other representations or warranties other than those expressly set forth herein.
- 4.2 **Representations and Warranties of the Camargo Group.** Each member of the Camargo Group, jointly and severally, represents and warrants to each member of the Alcoa Group that:
- 4.2.1 **Organization; Authorization.** Each member of the Camargo Group is duly organized, validly existing and in good standing under the laws of its place of incorporation or organization, and has full corporate power and authority to enter into this Agreement and the Registration Rights Agreement, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The Camargo Group has taken all corporate action necessary to authorize the execution and delivery by of it of this Agreement and the Registration Rights Agreement, the performance by it of its obligations hereunder and thereunder, and the consummation by it of the transactions contemplated hereby and thereby. This Agreement and the Registration Rights Agreement have been duly executed and delivered by each member of the Camargo Group and constitute the legal, valid and binding obligations of each member of the Camargo Group enforceable against such member of the Camargo Group in accordance with their respective terms, except as limited by applicable bankruptcy, insolvency, reorganization, moratorium or other laws of general application relating to or affecting enforcement of creditors' rights and rule of laws concerning equitable remedies.

- 4.2.2 **Alcoa Alumínio Stock and ALAHC Stock.** The shares in Alcoa Alumínio and ALAHC are owned as stated on **Exhibit A**. Such shares are and will be transferred to Alcoa, AL and/or their designated U.S. Person Affiliate(s) on the Closing Date free and clear of all Liens, and the shares in Alcoa Alumínio qualified as foreign registered capital with BACEN (other than the Tainted Shares) are and will be transferred to Alcoa, AL and/or their designated U.S. Person Affiliate(s) on the Closing Date together with all rights and interests deriving from the registration granted by BACEN to all such shares qualifying them as a foreign capital investment, including for purposes of dividends, capital interests and repatriation.
- 4.2.3 **No Violation.** Neither the execution and delivery by any member of the Camargo Group of this Agreement or the Registration Rights Agreement nor the consummation of the transactions contemplated hereby or thereby will result in: (a) a violation of or a conflict with any provision of the articles of incorporation or bylaws (or other comparable corporate charter documents) of any member of the Camargo Group; (b) a breach of, or a default (or give rise to any right of termination, cancellation or acceleration) under, any term or provision of any contract, agreement, indebtedness, lease, commitment, license, franchise, permit, authorization or concession to which it is a party or by which it is bound; or (c) a violation by the Camargo Group of any law, statute, rule, regulation, ordinance, code, order, judgment, writ, injunction, decree or award applicable to it, excluding in the foregoing clauses (b) and (c) such violations, breaches or defaults which are not reasonably likely to have a Material Adverse Effect. There is no pending or, to the knowledge of any member of the Camargo Group, threatened proceeding by or before any court, arbitrator or governmental authority against any member of the Camargo Group which in any manner challenges or seeks to prevent, enjoin, alter or materially delay the transactions contemplated by this Agreement or the Registration Rights Agreement, and, to the knowledge of any member of the Camargo Group, there exists no basis for any such proceeding.
- 4.2.4 **Governmental Consent, etc.** Except for obtaining the approval of CADE and except for compliance with the requirements of the Securities Act in connection with the Shelf Registration Statement, no consent, approval or authorization of, or declaration, filing or registration with, or any action by, any governmental or regulatory authority, or any other Person, is required to be made or obtained by the Camargo Group in connection with the execution, delivery and performance of this Agreement and, except for the transfer of foreign investment registrations as provided for in Sections 3.4 and 5.2.5, the consummation of the transactions contemplated hereby.
- 4.2.5 **Foreign Registered Capital.** The following shares owned by Allpar and Trelawney in Alcoa Alumínio are properly registered with BACEN as a foreign capital investment: (a) 1,705,985 common shares owned by Allpar, representative of 28.38% of the paid-in capital of Alcoa Alumínio, with a value of U.S.\$ 163,725,371.50, which constitutes a portion of the paid-in capital of Alcoa Alumínio in the amount of R\$197,316,041.62; and (b) 745,014 common shares owned by Trelawney, representative of 12.39% of the paid-in capital of Alcoa Alumínio, with a value of U.S.\$78,172,116.70, Swiss Francs 4,158,031.86 and Panama Balboa 512,861.55, which constitutes a portion of the paid-in capital of Alcoa Alumínio in the amount of R\$86,169,112.53. All such shares are freely transferable to any third party together with all rights and interests deriving from the registration granted by BACEN to all such shares qualifying them as a foreign capital investment, including for purposes of dividends, capital interests and repatriation. Upon the Closing, the title to the registration of all such shares with BACEN as a foreign capital investment shall be transferred, independent of any authorization

granted by BACEN or any other Brazilian authority, to Alcoa, AL and/or their designated U.S. Person Affiliate(s) as the new owner(s) of such shares.

- 4.2.6 **Tainted Shares.** In addition to the shares described in Section 4.2.5 above, Allpar owns 37 common shares and Trelawney owns 5,974 common shares representative of the paid-in capital of Alcoa Alumínio, totaling 6,011 shares which are not registered with BACEN as a foreign capital investment (the "Tainted Shares") and which are included in the total number of shares to be transferred to AL by Allpar and Trelawney, respectively, as described in Section 2.1. The Camargo Group shall cooperate with the Alcoa Group and shall use its best efforts upon the request of the Alcoa Group to prepare and submit to BACEN any application as may be required to obtain the registration of the Tainted Shares as a foreign capital investment.
 - 4.2.7 Status of Allpar and Trelawney. Allpar and Trelawney are both wholly-owned direct or indirect subsidiaries of Camargo Corrêa.
- 4.2.8 **Investment Intent.** Each member of the Camargo Group is acquiring the Shares for its own account, for investment and not with a view to, or for resale in connection with, any distribution or public offering in violation of the Securities Act.
- 4.2.9 **Accredited Investor.** Each member of the Camargo Group (a) is an "accredited investor" as that term is defined in Rule 501 of Regulation D promulgated under the Securities Act, (b) is able to bear the economic risk of the acquisition of the Shares allocated to such member pursuant to the terms of this Agreement, including a complete loss of such member's investment, and (c) is not an entity formed for the specific purpose of consummating the transactions contemplated by this Agreement.
- 4.2.10 **Access to Information.** During the negotiation of the transactions contemplated herein, Alcoa has furnished or otherwise made available to the Camargo Group true and complete copies of the Alcoa SEC Reports and such other information concerning Alcoa as the Camargo Group have deemed necessary in connection with their decision to acquire the Shares.
- 4.2.11 **Due Diligence Investigation.** Each member of the Camargo Group has been solely responsible for its own analysis of the merits and risks of the Exchange and the investment in the Shares, and for its own analysis of the fairness and desirability of the terms of the Exchange and such investment. In taking any action or performing any role relative to the arranging of the proposed investment, each member of the Camargo Group has acted solely in its own interest, and neither it nor any of its agents or employees has acted as an agent of Alcoa. Each member of the Camargo Group has such knowledge and experience in financial and business matters such that it is capable of evaluating the merits and risks of the Exchange and the acquisition of the Shares allocated to such member pursuant to the terms of this Agreement and of protecting its interests in connection therewith.
- 4.2.12 **Unregistered Securities.** Each member of the Camargo Group understands that the Shares have not been registered under the Securities Act or the securities laws of any state in the United States and that the Alcoa Group has no present intention of registering the Shares other than as contemplated by Section 2.3 hereof and the Registration Rights Agreement. Each member of the Camargo Group understands that the Shares are being exchanged and delivered to it pursuant to an exemption from registration contained in the Securities Act, based in part upon the representations contained in this Agreement.

- 4.2.13 **No Finder or Broker.** No member of the Camargo Group has retained any finder or broker in connection with the transactions contemplated by this Agreement or the Registration Rights Agreement.
 - 4.2.14 No Other Representations. The Camargo Group makes no other representations or warranties other than those expressly set forth herein.

ARTICLE V CONDITIONS PRECEDENT

- 5.1 <u>Camargo Group Conditions to Closing</u>. The obligations of the Camargo Group to consummate the transactions contemplated hereby shall be subject to the fulfillment (or waiver by the Camargo Group) on or prior to the Closing Date of the following conditions:
- 5.1.1 the representations and warranties of the Alcoa Group contained in Section 4.1 of this Agreement shall be deemed repeated and shall be true and correct in all material respects on and as of the Closing Date;
- 5.1.2 no statute, rule or regulation shall have been enacted which prohibits or materially restricts the consummation of the transactions contemplated by this Agreement, and no injunction or final judgment prohibiting or restraining the transactions contemplated hereby shall be in effect, and no governmental action shall be threatened or pending which seeks to restrain or prohibit or to obtain damages or other relief in connection with this Agreement or the transactions contemplated hereby;
- 5.1.3 all obligations of the Alcoa Group to be performed hereunder through and including the Closing Date shall have been performed in all material respects, and all those documents required to be delivered pursuant to Section 3.2 hereof shall have been received; and
- 5.1.4 all actions by or in respect of or filings with any governmental body, agency, official or authority required to permit the consummation of the transactions contemplated hereby shall have been taken, made or obtained. For the avoidance of doubt, the parties agree that the Closing shall not depend on the obtaining of CADE's approval.
- 5.2 <u>Alcoa Group Conditions to Closing</u>. The obligations of the Alcoa Group to consummate the transactions contemplated hereby shall be subject to the fulfillment (or waiver by the Alcoa Group, except for Section 5.2.5 which may not be waived) on or prior to the Closing Date of the following conditions:
- 5.2.1 the representations and warranties of the Camargo Group contained in Section 4.2 of this Agreement shall be deemed repeated and shall be true and correct in all material respects on and as of the Closing Date;
- 5.2.2 (a) no statute, rule or regulation shall have been enacted which prohibits or materially restricts the consummation of the transactions contemplated by this Agreement, and no injunction or final judgment prohibiting or restraining the transactions contemplated hereby shall be in effect, and no governmental action shall be threatened or pending which seeks to restrain or prohibit or to obtain damages or other relief in connection with this Agreement or the transactions contemplated hereby; and (b) no suit, action or other proceeding seeking to restrain, prohibit or materially restrict the transactions contemplated hereby, or seeking to obtain material damages or other material relief from Alcoa or any of its Affiliates in connection with this Agreement, shall be pending or threatened to be instituted by or on behalf any Person who

directly or indirectly is a member of the group of Persons with a direct or indirect ownership interest in any member of the Camargo Group;

- 5.2.3 all obligations of the Camargo Group to be performed hereunder through and including the Closing Date shall have been performed in all material respects, and all those documents required to be delivered pursuant to Section 3.3 hereof shall have been received;
- 5.2.4 all actions by or in respect of or filings with any governmental body, agency, official or authority required to permit the consummation of the transactions contemplated hereby shall have been taken, made or obtained. For the avoidance of doubt, the parties agree that the Closing shall not depend on the obtaining of CADE's approval; and
- 5.2.5 the transfer of foreign investment registrations with BACEN to Alcoa, AL and/or their designated U.S. Person Affiliate(s), as set forth in Section 3.4, shall have been effected. The transfer of foreign investment registrations as provided for in this Section 5.2.5 shall be evidenced by the Consolidated Statements of Foreign Capital Investment in Brazil ("Extratos Consolidados de Investimento Externo Direto no Brasil") as set forth in Section 3.3.4.

ARTICLE VI TERMINATION AND SURVIVAL

- 6.1 <u>Termination</u>. This Agreement and the transactions contemplated hereby may be terminated at any time before the Closing by notice given in accordance with Section 7.6:
 - 6.1.1 by the mutual written consent of the Camargo Group and the Alcoa Group; or
- 6.1.2 by either the Camargo Group or the Alcoa Group if the Closing shall not have occurred at or before 11:59 p.m., New York City Time, on December 31, 2003.
- 6.2 <u>Survival and Indemnity</u>. The representations and warranties of the parties shall survive for a period of five years from the Closing Date, except for those in Sections 4.1.8, 4.1.9, 4.1.10, 4.1.11, 4.1.12, 4.1.13 and 4.1.14 which shall survive for a period of eighteen months from the Closing Date. Each party shall indemnify and hold the others harmless from any breach by it of any of the representations and warranties and covenants and agreements made by such party in this Agreement. This Section 6.2 shall not limit any covenant or agreement set forth in this Agreement that by its terms contemplates performance after the Closing Date, which covenants and agreements shall survive the Closing Date.

ARTICLE VII MISCELLANEOUS

7.1 Conduct of Alcoa Alumínio and ALAHC. From the date hereof until the termination of the Earn-Out Period, Alcoa agrees to retain Alcoa Alumínio and ALAHC as consolidated subsidiaries, and each member of the Alcoa Group agrees to cause Alcoa Alumínio and ALAHC: (i) to diligently conduct its businesses in good faith; (ii) not to enter into or agree or commit to enter into any transaction with any of its Affiliates on terms other than those that would prevail between unrelated Persons acting on arms-length terms; and (iii) not to make or agree or commit to make any material change in accounting methods, principles or practices other than changes required by reason of a change in GAAP or changes that would not materially affect the EBITDA calculations provided for in Exhibit C.

- 7.2 Release of Pre-Closing Claims. Each member of the Alcoa Group and each of their respective successors and assigns, and all shareholders, directors, officers, members, agents or employees, hereby remise, release and forever discharge and undertake to hold harmless each member of the Camargo Group and each of their respective successors and assigns, and all shareholders, directors, officers, members, agents or employees, and their respective heirs, executors, administrators, successors and assigns (the "Camargo Released Parties"), from any and all liabilities (including environmental liabilities), whether at law or in equity (including any right of contribution), whether arising under any contract or agreement, by operation of law or otherwise, relating to, or arising out of the conduct of business by Alcoa Alumínio or ALAHC or the actions or omissions of any Camargo Released Party relating to such businesses, whether or not known as of the Closing Date. Each member of the Camargo Group and each of their respective successors and assigns, and all shareholders, directors, officers, members, agents or employees, hereby remise, release and forever discharge each member of the Alcoa Group, Alcoa Alumínio and ALAHC and each of their respective successors and assigns (the "Alcoa Released Parties"), from any and all liabilities, whether at law or in equity (including any right of contribution), whether arising under any contract or agreement, by operation of law or otherwise, relating to, or arising out of the conduct of business by Alcoa Alumínio or ALAHC or the actions or omissions of any Alcoa Released Party relating to such businesses, whether or not known as of the Closing Date. The foregoing provisions shall not operate as a release of any claims or liabilities arising out of a breach by a party hereto of any of its covenants or agreements or representations or warranties set forth in the first sentence of this Section.
- 7.3 <u>Publicity</u>. Unless required by law (including a requirement that, in the reasonable judgment of Alcoa's counsel, disclosure should be included in any filing with, or report required by, the SEC), no party shall issue any press release or make any public statement regarding the transactions contemplated hereby, without the prior approval of the other parties hereto.
- 7.4 <u>CADE</u>. The Alcoa Group and the Camargo Group hereby agree to cooperate and use their best efforts to prepare promptly and submit any supplemental application to, and to respond to any requests for additional information from, CADE as may be required and to use their best efforts to obtain the approval by CADE of the transactions contemplated hereby.
- 7.5 Expenses. Each party will bear its own costs and expenses incurred in connection with the negotiation, preparation, execution and performance of this Agreement and the transactions contemplated herein.
- 7.6 **Notices**. All notices required to be given hereunder shall be in writing and shall be served personally, sent by tested facsimile (with transmission confirmation as evidenced by a completed transmission report generated by the sender's facsimile machine) or delivered by an international courier to the addresses set forth below:

To the Alcoa Group: Alcoa Inc.

390 Park Avenue

New York, New York, U.S.A. 10022

Attention: Group President – Latin America

Facsimile: (212) 836-2816

Attention: Treasurer Facsimile: (212) 836-2823 Attention: General Counsel Facsimile: (212)836-2844

To the Camargo Group:

Camargo Corrêa S.A. Rua Funchal, 160 Sao Paulo, SP Brasil 04551-903

Attention: President

Facsimile: 55-11-3849-7837

Attention: Treasurer Facsimile: 55-11-3841-5658

Each such notice shall be effective (a) if delivered by international courier, when delivered to the recipient's address provided above, or, if such recipient shall have given effective notice hereunder of a new address, when delivered at such new address, (b) if sent by facsimile, when received by the recipient at his address provided above, or, if such recipient shall have given effective notice hereunder of a new address, when received at such new address, or (c) if delivered in person by hand, upon the delivery to the intended recipient.

7.7 <u>Assurances and Cooperation</u>. On and after the Closing Date, the parties will take all appropriate action and execute all documents, instruments or conveyances which may reasonably be necessary or advisable to carry out any of the provisions hereof. Each party shall coordinate and cooperate with each other in supplying such reasonable assistance as may be reasonably requested by the other in connection with the actions contemplated by this Agreement.

7.8 <u>Annual Reviews</u>. Alcoa and Camargo Corrêa will jointly engage an independent audit firm of international reputation, which shall not be the regular auditor of either party unless they agree otherwise, to conduct agreed upon procedures on an annual basis to review the amount, if any, of the Earn-Out Payment and Guaranteed Minimum Payment described on **Exhibit C** as they relate to periods after January 1, 2003. In connection with the performance of such agreed upon procedures, Alcoa will, and will cause Alcoa Alumínio and ALAHC to, provide such firm with: (a) access to all relevant financial statements of Alcoa Alumínio and ALAHC; (b) responses to such firm's reasonable requests for clarification or additional relevant information; and (c) the opportunity to meet with Alcoa Alumínio's chief financial officer. The costs of these annual reviews will be shared equally by Alcoa and Camargo Corrêa. All information disclosed as part of such reviews shall be subject to the confidentiality provisions of Section 7.9.

- 7.9 **Confidentiality**. Each member of the Camargo Group will hold, and will ensure that its officers, directors, employees, accountants, counsel, consultants, advisors and agents hold, in confidence and not disclose to any third party, unless compelled to disclose by judicial or administrative process or by other requirements of law, all financial, technical or other confidential information furnished to the Camargo Group by the Alcoa Group in connection with the transactions contemplated by this Agreement, except to the extent that such information can be shown to have been (i) previously known on a nonconfidential basis by a member of the Camargo Group, (ii) in the public domain through no fault of any member of the Camargo Group or (iii) later lawfully acquired by a member of the Camargo Group from sources other than the Alcoa Group who had no duty to hold it as confidential. The Camargo Group may disclose such information to its officers, directors, employees, accountants, counsel, consultants, advisors and agents who have a need to know such information in connection with the transactions contemplated by this Agreement so long as such persons are informed by the Camargo Group of the confidential nature of such information and are obligated by the Camargo Group to treat such information confidentially and in any arbitral or judicial proceeding pursuant to Section 7.17. Notwithstanding the foregoing, the Camargo Group may disclose the tax treatment and tax structure of the transactions contemplated by this Agreement (including any materials, opinions or analyses relating to such tax treatment or tax structure, but without disclosure of identifying information or, except to the extent relating to such tax structure or tax treatment, any nonpublic commercial or financial information). Moreover, notwithstanding any other provision of this Agreement, there shall be no limitation on the Camargo Group's ability to consult any tax adviser, whether or not independent from Camargo Group or its Affiliates, regarding the tax treatment or tax structure of the transactions contemplated by this Agreement. The Camargo Group shall be responsible for any failure to treat such information confidentially by such Persons and their respective officers, directors, employees, accountants, counsel, consultants, advisors and agents.
- 7.10 <u>Assignment</u>. Neither this Agreement nor any of the rights or obligations hereunder may be assigned by the Alcoa Group or the Camargo Group without the prior written consent of the other, which shall not be unreasonably withheld; provided, however, that the Camargo Group may assign the right to receive any Earn-Out Payment(s) and any Guaranteed Minimum Payment to its Affiliates provided the Camargo Group remains liable for its obligations hereunder. Subject to the foregoing, this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective legal representatives, successors, heirs and assigns, and no other person has any right, benefit or obligation hereunder.
- 7.11 **Waiver**. The waiver by any party of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any subsequent breach.
- 7.12 <u>Headings</u>. The headings of the Articles and Sections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.
- 7.13 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts in English, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 7.14 <u>Severability</u>. If any term, covenant, condition or provision of this Agreement or the application thereof to any circumstance shall be invalid or unenforceable to any extent, the

remaining terms, covenants, conditions and provisions of this Agreement shall not be affected thereby and each remaining term, covenant, condition or provision of this Agreement shall be enforceable to the fullest extent permitted by law.

- 7.15 **No Drafting Presumption**. The parties acknowledge that their respective legal counsel have reviewed and participated in settling the terms of this Agreement, and the parties agree that any rule of construction to the effect that any ambiguity is to be resolved against the drafting party shall not be applicable in the interpretation of this Agreement.
- 7.16 <u>Choice of Law</u>. This Agreement shall be construed, interpreted and the rights of the parties interpreted in accordance with the laws of the State of New York, excluding those related to choice or conflicts of law.
- 7.17 **Dispute Resolution**. In an effort to resolve informally and amicably any dispute or controversy which might arise between the parties with respect to this Agreement, without first resorting to any third party or tribunal for resolution, the parties shall first endeavor to resolve the matter through direct discussions and negotiations. If the matter cannot be settled, it shall be referred by each party to a responsible officer of each Alcoa and Camargo Corrêa. If the two officers are unable to resolve the matter within 30 days of its referral, the matter will be referred to the Chief Executive Officers of Alcoa and Camargo Corrêa. If settlement is not thereafter reached through their efforts within an additional 30 days, or such longer time period as they may agree, then either party may initiate arbitration proceedings to resolve the matter. Any such dispute, controversy or claim in connection with this Agreement, including its validity, interpretation, application, scope, enforceability, performance, breach and termination, shall be resolved exclusively and finally by arbitration, to the exclusion of the courts. Notice of arbitration shall be deemed proper if made in accordance with Section 7.6 of this Agreement. If the parties fail to agree in writing on the place where the arbitration is to be conducted, such arbitration shall be held in New York, New York, United States of America. Arbitration shall be conducted in English, pursuant to International Chamber of Commerce ("ICC") Arbitration Rules in force at the time of the arbitration, by a panel of three arbitrators who are fluent in the English language and who are skilled in the legal and business aspects of the subject matter of this Agreement. The arbitrators shall be appointed in accordance with ICC Rules. Any monetary award made pursuant to such arbitration shall be calculated and paid exclusively in Dollars. Judgment upon the award rendered may be entered in any court having jurisdiction, or an application may be made to any court for a judicial acceptance of the award and an order of enforcement, as the case may be. Each of the parties hereby irrevocably and unconditionally submits, for itself and its property, to the nonexclusive jurisdiction of any Federal court of the United States of America sitting in New York City and any appellate court from thereof for recognition or enforcement of any such judgment. Each party hereby irrevocably and unconditionally waives any objection (including any defense of inconvenient forum) which it may now or hereafter have to the laying of such venue, and each party irrevocably consents to service of process in the manner provided for notices in Section 7.6.
- 7.18 Entire Agreement and Modifications. This Agreement, together with the Exhibits hereto (including the Registration Rights Agreement), constitute the entire agreement between the parties with respect to the subject matter hereof and supersede all prior agreements (including the Agreement in Principle), understandings, negotiations and discussions, whether oral or written, of the parties. No supplement, modification or waiver of this Agreement shall be binding unless executed in writing by the party to be bound thereby.

7.19 <u>Nature of Obligations</u>. The covenants, agreements, undertakings and obligations herein of Alcoa and AL are joint and several. The covenants, agreements, undertakings and obligations herein of Camargo Corrêa, Allpar and Trelawney are joint and several.

[The remainder of this page is intentionally left blank.]

IN WITNESS WHEREOF , the parties hereto have exby their duly authorized representative, as of the day and year	xecuted this Agreement, or have caused this Agreement to be duly executed on their respective behalf ir first above written.
CAMARGO CORRÊA S.A.	ALCOA INC.
Ву	By
ALLPAR LIMITED	ALCOA LUXEMBOURG S.À.R.L.
Ву	Ву
TRELAWNEY INC.	

Name of holder of Alcoa Alumínio Stock	Number of shares	Approximate % of total
ALLPAR Limited	1,706,022	28.38598%
Trelawney Inc.	750,988	12.49543%
Carlos Pires Oliveira Dias	28	0.00047%
Fernando de Arruda Botelho	28	0.00047%
Luiz Roberto Ortiz Nascimento	28	0.00047%
Fernando Tigre de Barros Rodrigues	1	0.00002%
Daniel Antonio Biondo Bastos	1	0.00002%
Edy Luiz Kogut	1	0.00002%
Name of holder of ALAHC Stock	Number of shares	Approximate % of total
Trelawney Inc.	363,992	40.93%

Form of Transfer Document

DEED OF TRANSFER OF STOCK

This **DEED OF TRANSFER OF SHARES** (this "Deed") is made as of this __ day of _____, 2003, by **TRELAWNEY INC.**, a corporation duly organized and existing under the laws of The Bahamas ("Transferor"), in favor of **ALCOA INC.**, a corporation duly organized and existing under the laws of the Commonwealth of Pennsylvania, U.S.A. ("Transferee").

WHEREAS, Transferor is the owner of 363,992 common shares (the "ALAHC Stock") of ALCOA LATIN AMERICAN HOLDINGS CORPORATION, a corporation duly organized and existing under the laws of the British Virgin Islands; and

WHEREAS, Transferor, in return for good and valuable consideration, the receipt of which it hereby acknowledges, desires to transfer the ALAHC Stock to Transferee.

NOW THEREFORE, intending to be legally bound, the parties agree as follows:

Transferor hereby transfers to Transferee and Transferee hereby accepts ownership of the ALAHC Stock effective as of the date first stated above.

IN WITNESS WHEREOF, each of the parties hereto has duly executed this Deed of Transfer of Shares as of the date and year first above written.

TRELAWNEY INC.	ALCOA INC.
Ву	Ву

Form of Power of Attorney

POWER OF ATTORNEY

[Name of Grantor], a [Description of Grantor] ("Grantor"), [by its undersigned legal representative,] hereby	appoints and constitutes, as its sufficient and
legitimate attorney-in-fact,[FULL NAME AND QUALIFICATION, INCLUDING ADDF	RESS, CITY, COUNTRY AND TAX ROLL NO.
(CPF)] (the "Grantee"), to whom it grants special powers in order to, [jointly or separately,] on behalf and ir	n the name of Grantor, (i) transfer to Alcoa
Luxembourg, S.à.r.l., a private limited liability company organized and existing under the laws of Luxembourg	urg, enrolled with the Brazilian tax authorities
(CNPJ/MF) under number 05.731.008/0001-41 ("AL"), nominative common shares, with all the	neir attendant rights, which Grantor holds in the
corporate capital of the Brazilian corporation Alcoa Alumínio S.A. ("Alcoa Alumínio") and which are part c	of the group of shares of common stock of Alcoa
Alumínio, defined as the "Alcoa Alumínio Stock" in the Principal Agreement by and among Alcoa Inc., Alc Limited, and Trelawney Inc. dated July, 2003 (the "Agreement"), to be exchanged in accordance with the statement of transfer of the shares listed in item (i) above in the Alcoa Alumínio's Book of Transfer of Nom AL as the assignee of such shares thereby transferred, in compliance with Article 31, § 1, of the Brazilian Lacestand and the statement of transfer of the shares thereby transferred, in compliance with Article 31, § 1, of the Brazilian Lacestand and the statement of transfer of the shares thereby transferred, in compliance with Article 31, § 1, of the Brazilian Lacestand and the statement of transfer of the shares thereby transferred in compliance with Article 31, § 1, of the Brazilian Lacestand and the statement of transfer of the shares the statement of transfer of the shares the statement of transfer of the shares listed in item (i) above in the Alcoa Alumínio's Book of Transfer of Nom AL as the assignee of such shares thereby transferred, in compliance with Article 31, § 1, of the Brazilian Lacestand and the shares the statement of transfer of the shares the shar	e terms and conditions of the Agreement; (ii) sign the inative Shares, showing Grantor as the assignor and
This mandate shall be valid and effective until December 31, 2003.	
[PLACE AND DATE	[2]
Grantor:	
By:	
	[FULL NAME AND TITLE]

EARN-OUT PAYMENTS AND GUARANTEED MINIMUM PAYMENT

A. EARN-OUT PAYMENTS

This is Exhibit C to the Principal Agreement dated July 25, 2003 (the "Agreement") among Alcoa Inc., Alcoa Luxembourg S.à.r.l., Camargo Corrêa S.A., ALLPAR Limited and Trelawney Inc. Capitalized terms used and not defined herein have the respective meanings assigned to them in the Agreement.

The Camargo Group will be entitled to receive Earn-Out Payments (defined below), if the conditions set forth herein are met. Earn-Out Payments will be calculated on the basis of growth in Alcoa Alumínio's and ALAHC's (the "Subsidiaries") five-year average annual EBITDA (defined below) from an agreed base EBITDA of U.S.\$182 million.

The right to receive Earn-Out Payments and the Guaranteed Minimum Payment is not conditioned or contingent on the Camargo Group retaining an interest in the Shares.

Each Earn-Out Payment will be made in cash by Alcoa (as a U.S. Person) or by its designated Affiliate (provided that such Affiliate is a U.S. Person) to Trelawney and Allpar, in the same proportion as they receive Shares of Alcoa Common Stock on the Closing Date (i.e. 42% to Trelawney and 58% to Allpar). However, notwithstanding that an Earn-Out Payment, if the conditions set forth herein are met, is made in cash, any subsequent calculation of Capital Gain Offset (defined below) or notional dividends per share (see the Guaranteed Minimum Payment section below) will be made as if such Earn-Out Payment had been made in additional shares of Alcoa Common Stock (see below for calculations). The Earn-Out Payment period (the "Earn-Out Period") will be five fiscal years beginning with the fiscal year ended December 31, 2003 and will be based on the calculation periods described herein. Each Earn-Out Payment will be made, if the conditions set forth herein are met, annually by June 30 of the year subsequent to the year in respect of which the payment is made, beginning June 30, 2004.

Based upon the parties' understanding of current U.S. tax law, each Earn-Out Payment made by Alcoa to Trelawney and Allpar is not subject to deduction for any tax or withholding. In the event that U.S. tax law changes, the parties agree that taxes or withholdings will be properly withheld from future payments if required by applicable U.S. tax law. Additionally, for any Earn-Out Payment made by a U.S. Person, the parties agree to treat the amount of interest imputed with respect to such payment pursuant to Section 483 of the Internal Revenue Code of 1986, as amended, ("IRC") as "portfolio interest" within the meaning of Section 871(h) of the IRC. In this regard, it is agreed that in no event will the right to receive Earn-Out Payment(s) be assigned or transferred to a U.S. Person. It is further agreed that all payments which may contain imputed interest are payable outside the U.S. Any U.S. Person who holds this obligation for Earn-Out Payment(s) will be subject to limitations provided in Section 165(j) and 1287(a) of the Internal Revenue Code of 1986. In the event that the Alcoa Group or its Affiliates are assessed for the withholding taxes by the tax authorities, and all administrative appeals are exhausted in good faith (provided that the Alcoa Group shall not be obligated to pursue litigation), the Camargo Group agrees to reimburse the Alcoa Group and its Affiliates for any past withholding taxes paid to the U.S. tax authorities by the Alcoa Group or its Affiliates.

In establishing the value of U.S. \$410 million as forth in Section 2.1 of the Agreement, the parties have agreed that the Subsidiaries' five-year average annual EBITDA for the period January 1, 1998 through December 31, 2002 is U.S.\$182 million (the "Base EBITDA") and is based on the following annual EBITDA figures, which shall be used as necessary in the subsequent annual five-year average EBITDA calculations described below: U.S. \$134 million for the period January 1, 1998 through December 31, 1998; U.S.\$155 million for the period January 1, 1999 through December 31, 1999; U.S.\$233 million for the period January 1, 2000 through December 31, 2000; U.S.\$206 million for the period January 1, 2001 through December 31, 2001; and U.S.\$182 million for the period January 1, 2002 through December 31, 2002. The same methodology and definition (as set forth below) used in determining EBITDA for periods after January 1, 2003, provided that:

- 1. The results of any new business that is acquired or established by the Subsidiaries on or after January 1, 2003 and the results of any major investment made by the Subsidiaries on or after January 1, 2003 will be disregarded in determining EBITDA. For example, if either of the Subsidiaries makes any material acquisition or investment after January 1, 2003 and before the end of the Earn-Out Period, neither the financial results (including, without limitation, the revenues, liabilities and expenses of the acquired business) nor any debt or other liability incurred to finance such acquisition or investment shall be included in any annual EBITDA calculation. Examples of "new businesses" or "major investments" for purposes of this paragraph include: (a) entering a new line of business; and (b) material expansions of capacity (e.g., an additional line at a smelter, investing in cold-finish rod and bar equipment and facilities, significant investments in hydroelectric facilities, etc.). For clarification, the 27.23% investment in the hydroelectric facility at Machadinho and the 31.57% investment in the hydroelectric facility at Barra Grande will be included in the calculations of EBITDA.
- 2. If any business that had been part of the Base EBITDA calculation is sold on or after January 1, 2003, the gain or loss related to the sale of such business will not be taken into consideration in the annual EBITDA calculation, but the Alcoa Group will pay to Trelawney and Allpar (42% to Trelawney and 58% to Allpar), in cash in the currency received by the Subsidiaries in such sale, an amount equal to 40.9% (representing the Camargo Group's ownership interest in the Subsidiaries prior to the Closing) of the excess by which the sales price received for such business exceeded 6.5 times (the agreed earnings multiple) the EBITDA of such business for the average of the last five fiscal years prior to such sale, payable within 30 days after the later of the Closing Date and the date of such sale.

As used herein, EBITDA is defined and calculated as follows:

		1998	1999	2000	2001	2002
	Total revenue	1077.8	891.1	1057.5	929.8	825.1
Minus	Total cost of goods sold	972.5	744.5	825.6	726.9	635.8
Minus	General administrative and selling expense	68.7	64.5	66.3	55.2	51.7
Minus	Research and development expense	1.2	1.9	1.9	3.9	3.3
Resulting in the subtotal:	Net margin	35.4	80.3	163.7	143.9	134.3
Dlue (for Equity						

Plus (for Equity income) or

Minus (for Equity loss)	Equity income/(Equity loss)	6.3	(4.8)	3.1	2.0	1.3
Plus	Depreciation (including amortization)	92.3	79.8	66.6	60.0	46.7
Resulting in total:	EBITDA	134.0	155.3	233.4	205.9	182.4

The EBITDA calculation amounts will be derived from U.S. Dollar financial information prepared from the books and records of Alcoa Alumínio and ALAHC. Such information will be prepared in accordance with Alcoa's accounting policies and procedures applied on a consistent basis, which policies and procedures shall comply in all material respects with the accounting policies and procedures used in determining Base EBITDA. For clarification, results of operations held for sale but not yet sold (discontinued operations) are part of the EBITDA calculations, but items classified in the non-operating income/expense categories of Alcoa's chart of accounts (e.g., other income/expense, foreign exchange and derivative gains/losses, special items and gains/losses on investments) other than equity income (as specified in the chart above) are not part of the EBITDA calculations. All calculations shall be in U.S. Dollars. Annual reviews of calculations as they relate to periods after January 1, 2003 shall be as provided in Section 7.8 of the Agreement.

To determine the notional number of shares that represent an Earn-Out Payment for these calculations (the Notional Earn-Out Shares, as defined below), the amount of the Earn-Out Payment will be divided by the average closing price of Alcoa Common Stock on the New York Stock Exchange (as reported by Bloomberg Financial Markets, or, if not reported thereby, by another authoritative source as may be agreed by the parties in writing) for the three calendar month period from March 1 through May 31 immediately preceding such Earn-Out Payment. Specifically, as used herein:

Year 1 Average Closing Price	means the average closing price of Alcoa common stock on the New York Stock Exchange for the three calendar month period from March 1, 2004 through May 31, 2004.
Year 2 Average Closing Price	means the average closing price of Alcoa common stock on the New York Stock Exchange for the three calendar month period from March 1, 2005 through May 31, 2005.
Year 3 Average Closing Price	means the average closing price of Alcoa common stock on the New York Stock Exchange for the three calendar month period from March 1, 2006 through May 31, 2006.
Year 4 Average Closing Price	means the average closing price of Alcoa common stock on the New York Stock Exchange for the three calendar month period from March 1, 2007 through May 31, 2007.
Year 5 Average Closing Price	means the average closing price of Alcoa common stock on the New York Stock Exchange for the three calendar month period from March 1, 2008 through May 31, 2008.
Year 1 Notional Earn-Out Shares	means the number of shares calculated by dividing the Year 1 Earn-Out Payment (defined below) by the greater of: the Year 1 Average Closing Price; or \$19.00.
Year 2 Notional Earn-Out Shares	means the number of shares calculated by dividing the Year 2 Earn-Out Payment (defined below) by the greater of: the Year 2 Average Closing Price; or \$19.00.
Year 3 Notional Earn-Out Shares	means the number of shares calculated by dividing the Year 3 Earn-Out Payment (defined below) by the greater of : the Year 3 Average Closing Price; or \$19.00.

Year 4 Notional Earn-Out Shares means the number of shares calculated by dividing the Year 4 Earn-Out Payment (defined below) by the greater of: the Year 4 Average Closing Price; or \$19.00.

1. Annual Five-Year Average EBITDA Calculation.

At the end of each fiscal year, the Alcoa Group will calculate the Subsidiaries' average annual EBITDA for the five years then ended. This average will be compared to the Base EBITDA of U.S.\$182 million.

2. Earn-Out Opportunity

If the difference between the latest average annual five-year EBITDA amount and the Base EBITDA of U.S.\$182 million is a positive amount, such difference will be multiplied by 6.5 (the agreed earnings multiple) and the resulting product will be further multiplied by 40.9% (representing the Camargo Group's ownership interest in the Subsidiaries prior to the Closing) to obtain the Earn-Out Opportunity. The Earn-Out Opportunity for each year is capped at U.S.\$235 million. Any amount over U.S.\$235 million is disregarded for purposes of any further calculation. As used herein:

Year 1 Earn-Out Opportunity	means the calculation described above based on the average annual five-year EBITDA amount for the period January 1, 1999 through December 31, 2003.
Year 2 Earn-Out Opportunity	means the calculation described above based on the average annual five-year EBITDA amount for the period January 1, 2000 through December 31, 2004.
Year 3 Earn-Out Opportunity	means the calculation described above based on the average annual five-year EBITDA amount for the period January 1, 2001 through December 31, 2005.
Year 4 Earn-Out Opportunity	means the calculation described above based on the average annual five-year EBITDA amount for the period January 1, 2002 through December 31, 2006.
Year 5 Earn-Out Opportunity	means the calculation described above based on the average annual five-year EBITDA amount for the period January 1, 2003 through December 31, 2007.

3. Capital Gain Offset

As used herein, the "Closing Date Shares" means the 17,773,540 Shares of Alcoa Common Stock exchanged on the Closing Date.

3(a) Capital Gain Offset for Closing Date Shares

Capital Gain Offset for Closing Date Shares is a notional calculation of the following factors: (a) the positive difference (if any) by which the average closing price of Alcoa Common Stock on the New York Stock Exchange (as reported by Bloomberg Financial Markets, or, if not reported thereby, by another authoritative

source as may be agreed by the parties in writing) for the three calendar month period from March 1 through May 31 of each year beginning March 1, 2004, exceeds \$23.068 (i.e. the average price per share used to calculate the number of Shares exchanged on the Closing Date); (b) multiplied by 3,554,708 (i.e. 20% of the Closing Date Shares). As used herein:

means (a) the positive difference (if any) by which the Year 5 Average Closing Price exceeds

\$23.068, (b) multiplied by 3,554,708 (i.e. 20% of the Closing Date Shares).

Year 1 Capital Gain Offset for means (a) the positive difference (if any) by which the Year 1 Average Closing Price exceeds \$23.068, (b) multiplied by 3,554,708 (i.e. 20% of the Closing Date Shares). Closing Date Shares Year 2 Capital Gain Offset for means (a) the positive difference (if any) by which the Year 2 Average Closing Price exceeds Closing Date Shares \$23.068, (b) multiplied by 3,554,708 (i.e. 20% of the Closing Date Shares). means (a) the positive difference (if any) by which the Year 3 Average Closing Price exceeds Year 3 Capital Gain Offset for Closing Date Shares \$23.068, (b) multiplied by 3,554,708 (i.e. 20% of the Closing Date Shares). Year 4 Capital Gain Offset for means (a) the positive difference (if any) by which the Year 4 Average Closing Price exceeds Closing Date Shares \$23.068, (b) multiplied by 3,554,708 (i.e. 20% of the Closing Date Shares).

3(b) Capital Gain Offset for Notional Earn-Out Shares

Year 5 Capital Gain Offset for

Closing Date Shares

A similar calculation applies to notional shares representing Earn-Out Payments. As used herein:

Year 1 Capital Gain Offset for is 0. Notional Earn-Out Shares Year 2 Capital Gain Offset for means (a) the positive difference (if any) by which the Year 2 Average Closing Price exceeds the Notional Earn-Out Shares Year 1 Average Closing Price, (b) multiplied by 20% of the Year 1 Notional Earn-Out Shares. Year 3 Capital Gain Offset for means the sum of: Notional Earn-Out Shares (i) (a) the positive difference (if any) by which the Year 3 Average Closing Price exceeds the Year 2 Average Closing Price, (b) multiplied by 20% of the Year 2 Notional Earn-Out Shares; plus (ii) (a) the positive difference (if any) by which the Year 3 Average Closing Price exceeds the Year 1 Average Closing Price, (b) multiplied by 20% of the Year 1 Notional Earn-Out Shares. Year 4 Capital Gain Offset for means the sum of: Notional Earn-Out Shares (i) (a) the positive difference (if any) by which the Year 4 Average Closing Price exceeds the Year 3 Average Closing Price, (b) multiplied by 20% of the Year 3 Notional Earn-Out Shares; plus (ii) (a) the positive difference (if any) by which the Year 4 Average Closing Price exceeds the Year 2 Average Closing Price, (b) multiplied by 20% of the Year 2 Notional Earn-Out Shares; plus (iii) (a) the positive difference (if any) by which the Year 4 Average Closing Price exceeds the Year 1 Average Closing Price, (b) multiplied by 20% of the Year 1 Notional Earn-Out Shares. Year 5 Capital Gain Offset for means the sum of: Notional Earn-Out Shares (i) (a) the positive difference (if any) by which the Year 5 Average Closing Price exceeds the Year

4 Average Closing

Price, (b) multiplied by 20% of the Year 4 Notional Earn-Out Shares; plus

(ii) (a) the positive difference (if any) by which the Year 5 Average Closing Price exceeds the Year 3 Average Closing Price, (b) multiplied by 20% of the Year 3 Notional Earn-Out Shares; plus (iii) (a) the positive difference (if any) by which the Year 5 Average Closing Price exceeds the Year 2 Average Closing Price, (b) multiplied by 20% of the Year 2 Notional Earn-Out Shares; plus (iv) (a) the positive difference (if any) by which the Year 5 Average Closing Price exceeds the Year 1 Average Closing Price, (b) multiplied by 20% of the Year 1 Notional Earn-Out Shares.

3(c) Total Capital Gain Offset

Thus, for each year, the Total Capital Gain Offset is the sum of that year's Capital Gain Offset for Closing Date Shares and that year's Capital Gain Offset for Notional Earn-Out Shares. As used herein:

Year 1 Total Capital Gain Offset means the sum of:

the Year 1 Capital Gain Offset for Closing Date Shares and the Year 1 Capital Gain Offset for Notional Earn-Out Shares.

Year 2 Total Capital Gain Offset means the sum of:

the Year 2 Capital Gain Offset for Closing Date Shares and the Year 2 Capital Gain Offset for Notional Earn-Out Shares.

Year 3 Total Capital Gain Offset means the sum of:

the Year 3 Capital Gain Offset for Closing Date Shares and the Year 3 Capital Gain Offset for Notional Earn-Out Shares.

Year 4 Total Capital Gain Offset means the sum of:

the Year 4 Capital Gain Offset for Closing Date Shares and the Year 4 Capital Gain Offset for Notional Earn-Out Shares.

Year 5 Total Capital Gain Offset means the sum of:

the Year 5 Capital Gain Offset for Closing Date Shares and the Year 5 Capital Gain Offset for Notional Earn-Out Shares.

4. Earn-Out Payments

Earn-Out Payments will be calculated and defined as follows:

Year 1 Earn-Out Payment means the positive difference (if any) between

(i) 20% of the Year 1 Earn-Out Opportunity, minus

(ii) the Year 1 Total Capital Gain Offset.

Year 2 Earn-Out Payment means the positive difference (if any) between

(i) 40% of the Year 2 Earn-Out Opportunity, minus

(ii) the sum of (a) the Year 2 Total Capital Gain Offset, (b) the Year 1 Total Capital Gain Offset,

and (c) any Year 1 Earn-Out Payment.

Year 3 Earn-Out Payment means the positive difference (if any) between

(i) 60% of the Year 3 Earn-Out Opportunity, minus

(ii) the sum of (a) the Year 3 Total Capital Gain Offset, (b) the Year 2 Total Capital Gain Offset, (c) the Year 1 Capital Gain Offset, (d) any Year 2 Earn-Out Payment, and (e) any Year 1 Earn-Out

Payment.

Year 4 Earn-Out Payment means the positive difference (if any) between

(i) 80% of the Year 4 Earn-Out Opportunity, minus

(ii) the sum of (a) the Year 4 Total Capital Gain Offset, (b) the Year 3 Total Capital Gain Offset, (c) the Year 2 Total Capital Gain Offset, (d) the Year 1 Capital Gain Offset, (e) any Year 3 Earn-Out Payment,

(f) any Year 2 Earn-Out Payment, and (g) any Year 1 Earn-Out Payment.

Year 5 Earn-Out Payment means the

means the positive difference (if any) between (i) 100% of the Year 5 Earn-Out Opportunity, minus

(ii) the sum of (a) the Year 5 Total Capital Gain Offset, (b) the Year 4 Total Capital Gain Offset, (c) the Year 3 Total Capital Gain Offset, (d) the Year 2 Total Capital Gain Offset, (e) the Year 1 Capital Gain Offset, (f) any Year 4 Earn-Out Payment, (g) any Year 3 Earn-Out Payment, (h) any Year 2 Earn-Out

Payment, and (i) any Year 1 Earn-Out Payment.

The cumulative amount of the Year 1, Year 2, Year 3, Year 4 and Year 5 Earn-Out Payments may not exceed U.S.\$235 million.

B. GUARANTEED MINIMUM PAYMENT

Except in the case of the occurrence of a Share Price Event (as hereafter defined), the Alcoa Group will make a payment to Trelawney and Allpar (in the proportion 42% and 58%, respectively), in cash, promptly after the end of the five-year Earn-Out Payment period (i.e. on or prior to June 30, 2008) of the positive difference (if any) between (i) U.S.\$550 million, minus (ii) the sum of the following:

- U.S.\$410 million;
- The aggregate amount of the Year 1 Total Capital Gain Offset, the Year 2 Total Capital Gain Offset, the Year 3 Total Capital Gain Offset, the Year 4 Total Capital Gain Offset and the Year 5 Total Capital Gain Offset;
- The aggregate amount of any Year 1 Earn-Out Payment, any Year 2 Earn-Out Payment, any Year 3 Earn-Out Payment, any Year 4 Earn-Out Payment and any Year 5 Earn-Out Payment;
- The Total Notional Dividends on the Closing Date Shares (defined below); and
- The Total Notional Dividends on Notional Earn-Out Shares (defined below).

As used herein, the following terms shall have the following meanings:

Year 1 Alcoa Dividends Per Share	means the amount of dividends declared per share of Alcoa Common Stock for the period from May 5, 2003 through May 4, 2004.
Year 2 Alcoa Dividends Per Share	means the amount of dividends declared per share of Alcoa Common Stock for the period from May 5, 2004 through May 4, 2005.
Year 3 Alcoa Dividends Per Share	means the amount of dividends declared per share of Alcoa Common Stock for the period from May 5, 2005 through May 4, 2006.
Year 4 Alcoa Dividends Per Share	means the amount of dividends declared per share of Alcoa Common Stock for the period from May 5, 2006 through May 4, 2007.
Year 5 Alcoa Dividends Per Share	means the amount of dividends declared per share of Alcoa Common Stock for the period from May 5, 2007 through May 4, 2008.
Total Notional Dividends on the Closing Date Shares means	the sum of: (i) the Year 1 Alcoa Dividends Per Share, multiplied by 17,773,540 (i.e. 100% of the Closing Date Shares); plus (ii) the Year 2 Alcoa Dividends Per Share, multiplied by 14,218,832 (i.e. 80% of the Closing Date Shares); plus

(iii) the Year 3 Alcoa Dividends Per Share, multiplied by 10,664,124 (i.e. 60% of the Closing Date Shares); plus

(iv) the Year 4 Alcoa Dividends Per Share, multiplied by 7,109,416 (i.e. 40% of the Closing Date Shares); plus

(v) the Year 5 Alcoa Dividends Per Share, multiplied by 3,554,708 (i.e. 20% of the Closing Date Shares).

Total Notional Dividends on Notional Earn-Out Shares means the sum of:

(i) the Year 2 Alcoa Dividends Per Share, multiplied by 100% of the Year 1 Notional Earn-Out Shares; plus

(ii) the Year 3 Alcoa Dividends Per Share, multiplied by the sum of (a) 80% of the Year 1 Notional Earn-Out Shares and (b) 100% of the Year 2 Notional Earn-Out Shares; plus

(iii) the Year 4 Alcoa Dividends Per Share, multiplied by the sum of (a) 60% of the Year 1 Notional Earn-Out Shares, (b) 80% of the Year 2 Notional Earn-Out Shares and (c) 100% of the Year 3 Notional Earn-Out Shares; plus

(iv) the Year 5 Alcoa Dividends Per Share, multiplied by the sum of (a) 40% of the Year 1 Notional Earn-Out Shares, (b) 60% of the Year 2 Notional Earn-Out Shares, (c) 80% of the Year 3 Notional Earn-Out Shares and (d) 100% of the Year 4 Notional Earn-Out Shares.

A Share Price Event is deemed to have occurred whenever the simple arithmetic average of each day's closing price per share of Alcoa Common Stock on the New York Stock Exchange (as reported by Bloomberg Financial Markets, or, if not reported thereby, by another authoritative source as may be agreed by the parties in writing) determined for any period of 30 consecutive Trading Days during the five-year period from May 5, 2003 through May 4, 2008 exceeds U.S.\$32 per share (the Threshold Share Price). For purposes of the calculation in the previous sentence, from the Closing Date until the second anniversary thereof, if: (a) Camargo Corrêa delivers a Proposed Sale Notice to Alcoa pursuant to Section 2.4.2 of the Agreement; (b) the Camargo Group and its Affiliates are prevented from selling the Shares described in such Proposed Sale Notice under the Shelf Registration Statement because the Shelf Registration Statement is not effective or a Delay Period is in effect; (c) such Shares cannot, based upon written advice of counsel to Camargo Corrêa reasonably acceptable to Alcoa, be sold by the Camargo Group and its Affiliates under Rule 144 without violating the confidentiality agreement set forth in Section 7.9 of the Agreement or in Section 4.2(g)(3) of the Registration Rights Agreement; (d) Camargo Corrêa does not rescind such Proposed Sale Notice pursuant to Section 2.7 of the Agreement; and (e) no sale of such Shares to Alcoa or its Affiliates (or to a third party arranged by Alcoa after public disclosure of Confidential Information) is consummated under Section 2.4.2 or 2.7 of the Agreement, then the Trading Day next succeeding the Trading Day on which such Proposed Sale Notice was effective shall be excluded from the period of 30 consecutive Trading Days described in the preceding sentence. For the avoidance of doubt, the period of 30 consecutive Trading Days for the purposes of such calculation shall include Trading Days (other than excluded Trading Days) before and after the excluded Trading Day. Alcoa will gi

If applicable, the number of shares and per share prices used in the above calculations (e.g., the Threshold Share Price) will be adjusted in accordance with Section 2.6 of the Agreement. For illustrative purposes only, examples of the above calculations are attached hereto as Exhibit C-1.

CALCULATION EXAMPLES OF EARN-OUT PAYMENTS AND GUARANTEED MINIMUM PAYMENT

Example 1 Hypothetical Scenario:

- Earn-Out Payments are made.
- The Earn-Out Opportunity cap is not reached.
- No Guaranteed Minimum Payment is made.

			1998	1999	2000	2001	2002	
	Historical EBITDA		134	155	233	206	182	
		Base	Year 1	Year 2	Year 3	Year 4	Year 5	Total
1	EBITDA							
	Annual EBITDA		202	218	236	254	275	
	Annual five-year average EBITDA	182	196	208	209	218	237	
	EBITDA difference for Earn-Out Opportunity calculation		14	26	27	36	55	
2	Earn-Out Opportunity calculation							
	For 100% share		88.4	170.5	173.9	236.9	357.6	
	For Camargo Group's 40.9% share		36.2	69.7	71.1	96.9	146.2	
	For Camargo Group's share after US\$235m cap		36.2	69.7	71.1	96.9	146.2	
	Portion of Earn-Out Opportunity used in calculations (%)		20%	40%	60%	80%	100%	
4	Earn-Out Payment							
	Earn-Out Opportunity used in calculation		7.2	27.9	42.7	77.5	146.2	
	Total Capital Gain Offset		4.1	8.4	13.1	18.0	23.4	
	Cumulative prior-year Total Capital Gain Offsets		_	4.1	12.5	25.6	43.7	
	Cumulative prior-year Earn-Out Payments		_	3.1	15.4	17.0	33.8	
	Earn-Out Payment		3.1	12.2	1.7	16.8	45.4	
	Value Received by Camargo Group							
	Value of Closing Date Shares	410.0						410.0
	Earn-Out Payments		3.1	12.2	1.7	16.8	45.4	79.2
	Total Capital Gain Offsets		4.1	8.4	13.1	18.0	23.4	67.1
	Total Notional Dividends		10.7	8.6	6.7	4.6	2.7	33.3
	Guaranteed Minimum Payment							_
	Cumulative Total	410.0	427.9	457.2	478.7	518.1	589.6	589.6

Note: All amounts in US\$ million unless otherwise stated.

Numbers on the left are references to paragraphs in Exhibit C.

Exam	ple 1 (continued):						
		Base	Year 1	Year 2	Year 3	Year 4	Year 5
3	Capital Gain Offset Calculation						
	Price of Alcoa Common Stock (US\$/share)						
	Price used to calculate Closing Date Shares	\$23.068					
	Average Closing Price		\$24.22	\$25.43	\$ 26.70	\$28.04	\$ 29.44
	Number of shares used for Capital Gain Offset calculations (million shares)						
3(a)	Closing Date Shares	17.77					
3(b)	Notional Earn-Out Shares		0.13	0.48	0.06	0.60	1.54
	Cumulative Total	17.77	17.90	18.38	18.45	19.05	20.59
	Imputed sale of shares (million shares)						
3(a)	Closing Date Shares		3.55	3.55	3.55	3.55	3.55
3(b)	Year 1 Notional Earn-Out Shares			0.03	0.03	0.03	0.03
3(b)	Year 2 Notional Earn-Out Shares				0.10	0.10	0.10
3(b)	Year 3 Notional Earn-Out Shares					0.01	0.01
3(b)	Year 4 Notional Earn-Out Shares						0.12
	Capital Gain Offset						
3(a)	Closing Date Shares		4.10	8.41	12.93	17.67	22.66
3(b)	Year 1 Notional Earn-Out Shares			0.03	0.06	0.10	0.13
3(b)	Year 2 Notional Earn-Out Shares				0.12	0.25	0.39
3(b)	Year 3 Notional Earn-Out Shares					0.02	0.03
3(b)	Year 4 Notional Earn-Out Shares						0.17
3(c)	Total Capital Gain Offset		4.10	8.44	13.11	18.04	23.38
	Notional Dividends Calculation						
	Alcoa Dividends Per Share (US\$/share)		\$ 0.60	\$ 0.60	\$ 0.60	\$ 0.60	\$ 0.60
	Imputed shareholding for Notional Dividend calculations (million shares)						
	Closing Date Shares		17.77	14.22	10.66	7.11	3.55
	Year 1 Notional Earn-Out Shares			0.13	0.10	0.08	0.05
	Year 2 Notional Earn-Out Shares				0.48	0.38	0.29
	Year 3 Notional Earn-Out Shares					0.06	0.05
	Year 4 Notional Earn-Out Shares						0.60
	Total		17.77	14.35	11.25	7.63	4.54
	Notional Dividends		10.7	8.6	6.7	4.6	2.7

Note: All amounts in US\$ million unless otherwise stated.

Numbers on the left are references to paragraphs in Exhibit C.

Example 2 Hypothetical Scenario:

- No Earn-Out Payments are made.
- No Guaranteed Minimum Payment is made.

			1998	1999	2000	2001	2002	
	Historical EBITDA		134	155	233	206	182	
		Base	Year 1	Year 2	Year 3	Year 4	Year 5	Total
1	EBITDA							
	Annual EBITDA		202	208	214	221	227	
	Annual five-year average EBITDA	182	196	206	202	205	214	
	EBITDA difference for Earn-Out Opportunity calculation		14	24	20	23	32	
2	Earn-Out Opportunity calculation							
	For 100% share		88.4	157.4	133.1	152.2	211.2	
	For Camargo Group's 40.9% share		36.2	64.4	54.4	62.3	86.4	
	For Camargo Group's share after US\$235m cap		36.2	64.4	54.4	62.3	86.4	
	Portion of Earn-Out Opportunity used in calculations (%)		20%	40%	60%	80%	100%	
4	Earn-Out Payment							
	Earn-Out Opportunity used in calculation		7.2	25.7	32.7	49.8	86.4	
	Total Capital Gain Offset		9.8	20.9	33.2	47.0	62.5	
	Cumulative prior-year Total Capital Gain Offsets		_	9.8	30.7	63.9	110.9	
	Cumulative prior-year Earn-Out Payments							
	Earn-Out Payment		_	_	_	_	_	
	Value Received by Camargo Group							
	Value of Closing Date Shares	410.0						410.0
	Earn-Out Payments		_	_	_	_	_	_
	Total Capital Gain Offsets		9.8	20.9	33.2	47.0	62.5	173.4
	Total Notional Dividends		10.7	8.5	6.4	4.3	2.1	32.0
	Guaranteed Minimum Payment							_
	Cumulative Total	410.0	430.5	459.9	499.5	550.8	615.4	615.4

Note: All amounts in US\$ million unless otherwise stated.

Numbers on the left are references to paragraphs in Exhibit C.

Exam	ple 2 Hypothetical Scenario (continued):						
		Base	Year 1	Year 2	Year 3	Year 4	Year 5
3	Capital Gain Offset Calculation						
	Price of Alcoa Common Stock (US\$/share)						
	Price used to calculate Closing Date Shares	\$23.068					
	Average Closing Price		\$ 25.84	\$ 28.94	\$32.41	\$36.30	\$ 40.65
	Number of shares used for Capital Gain Offset calculations (million shares)						
3(a)	Closing Date Shares	17.77					
3(b)	Notional Earn-Out Shares		_	_	_	_	_
	Cumulative Total	17.77	17.77	17.77	17.77	17.77	17.77
	Imputed sale of shares (million shares)						
3(a)	Closing Date Shares		3.55	3.55	3.55	3.55	3.55
3(b)	Year 1 Notional Earn-Out Shares			_	_	_	_
3(b)	Year 2 Notional Earn-Out Shares				_	_	_
3(b)	Year 3 Notional Earn-Out Shares					_	_
3(b)	Year 4 Notional Earn-Out Shares						_
	Capital Gain Offset						
3(a)	Closing Date Shares		9.84	20.86	33.20	47.03	62.51
3(b)	Year 1 Notional Earn-Out Shares			_	_	_	_
3(b)	Year 2 Notional Earn-Out Shares				_	_	_
3(b)	Year 3 Notional Earn-Out Shares					_	_
3(b)	Year 4 Notional Earn-Out Shares						_
3(c)	Total Capital Gain Offset		9.84	20.86	33.20	47.03	62.51
	Notional Dividends Calculation						
	Alcoa Dividends Per Share (US\$/share)		\$ 0.60	\$ 0.60	\$ 0.60	\$ 0.60	\$ 0.60
	Imputed shareholding for Notional Dividend calculations (million shares)						
	Closing Date Shares		17.77	14.22	10.66	7.11	3.55
	Year 1 Notional Earn-Out Shares			_	_	_	_
	Year 2 Notional Earn-Out Shares				_	_	_
	Year 3 Notional Earn-Out Shares					_	_
	Year 4 Notional Earn-Out Shares						_
	Total		17.77	14.22	10.66	7.11	3.55
	Notional Dividends		10.7	8.5	6.4	4.3	2.1

Note: All amounts in US\$ million unless otherwise stated.

Numbers on the left are references to paragraphs in Exhibit C.

Example 3 Hypothetical Scenario:

- Earn-Out Payments are made.
- A Guaranteed Minimum Payment is made.

			1998	1999	2000	2001	2002	
	Historical EBITDA		134	155	233	206	182	
		Base	Year 1	Year 2	Year 3	Year 4	Year 5	Total
1	EBITDA							
	Annual EBITDA		202	208	214	221	227	
	Annual five-year average EBITDA	182	196	206	202	205	214	
	EBITDA difference for Earn-Out Opportunity calculation		14	24	20	23	32	
2	Earn-Out Opportunity calculation							
	For 100% share		88.4	157.4	133.1	152.2	211.2	
	For Camargo Group's 40.9% share		36.2	64.4	54.4	62.3	86.4	
	For Camargo Group's share after US\$235m cap		36.2	64.4	54.4	62.3	86.4	
	Portion of Earn-Out Opportunity used in calculations (%)		20%	40%	60%	80%	100%	
4	Earn-Out Payment							
	Earn-Out Opportunity used in calculation		7.2	25.7	32.7	49.8	86.4	
	Total Capital Gain Offset		8.0	1.7	2.5	3.4	4.4	
	Cumulative prior-year Total Capital Gain Offsets		_	8.0	2.5	5.0	8.5	
	Cumulative prior-year Earn-Out Payments		_	6.4	23.3	27.6	41.3	
	Earn-Out Payment		6.4	16.9	4.4	13.7	32.2	
	Value Received by Camargo Group							
	Value of Closing Date Shares	410.0						410.0
	Earn-Out Payments		6.4	16.9	4.4	13.7	32.2	73.5
	Total Capital Gain Offsets		8.0	1.7	2.5	3.4	4.4	12.9
	Total Notional Dividends		10.7	8.7	7.0	4.8	2.9	34.0
	Guaranteed Minimum Payment							19.6
	Cumulative Total	410.0	427.9	455.1	469.0	490.9	530.4	550.0

Note: All amounts in US\$ million unless otherwise stated.

Numbers on the left are references to paragraphs in Exhibit C.

Exam	ple 3 Hypothetical Scenario (continued):						
		Base	Year 1	Year 2	Year 3	Year 4	Year 5
3	Capital Gain Offset Calculation						
	Price of Alcoa Common Stock (US\$/share)						
	Price used to calculate Closing Date Shares	\$23.068					
	Average Closing Price		\$23.30	\$23.53	\$23.77	\$24.00	\$ 24.24
	Number of shares used for Capital Gain Offset calculations (million shares)						
3(a)	Closing Date Shares	17.77					
3(b)	Notional Earn-Out Shares		0.28	0.72	0.18	0.57	1.33
	Cumulative Total	17.77	18.05	18.76	18.95	19.52	20.85
	Imputed sale of shares (million shares)						
3(a)	Closing Date Shares		3.55	3.55	3.55	3.55	3.55
3(b)	Year 1 Notional Earn-Out Shares			0.06	0.06	0.06	0.06
3(b)	Year 2 Notional Earn-Out Shares				0.14	0.14	0.14
3(b)	Year 3 Notional Earn-Out Shares					0.04	0.04
3(b)	Year 4 Notional Earn-Out Shares						0.11
	Capital Gain Offset						
3(a)	Closing Date Shares		0.82	1.65	2.48	3.33	4.18
3(b)	Year 1 Notional Earn-Out Shares			0.01	0.03	0.04	0.05
3(b)	Year 2 Notional Earn-Out Shares				0.03	0.07	0.10
3(b)	Year 3 Notional Earn-Out Shares					0.01	0.02
3(b)	Year 4 Notional Earn-Out Shares						0.03
3(c)	Total Capital Gain Offset		0.82	1.66	2.54	3.44	4.38
	Notional Dividends Calculation						
	Alcoa Dividends Per Share (US\$/share)		\$ 0.60	\$ 0.60	\$ 0.60	\$ 0.60	\$ 0.60
	Imputed shareholding for Notional Dividend calculations (million shares)						
	Closing Date Shares		17.77	14.22	10.66	7.11	3.55
	Year 1 Notional Earn-Out Shares			0.28	0.22	0.17	0.11
	Year 2 Notional Earn-Out Shares				0.72	0.57	0.43
	Year 3 Notional Earn-Out Shares					0.18	0.15
	Year 4 Notional Earn-Out Shares						0.57
	Total		17.77	14.49	11.60	8.03	4.81
	Notional Dividends		10.7	8.7	7.0	4.8	2.9

Note: All amounts in US\$ million unless otherwise stated.

Numbers on the left are references to paragraphs in Exhibit C.

Example 4 Hypothetical Scenario:

- Earn-Out Payments are made.
- The Earn-Out Opportunity cap is reached.
- No Guaranteed Minimum Payment is made.

			1998	1999	2000	2001	2002	
	Historical EBITDA		134	155	233	206	182	
		Base	Year 1	Year 2	Year 3	Year 4	Year 5	Total
1	EBITDA							
	Annual EBITDA		202	236	277	324	379	
	Annual five-year average EBITDA	182	196	212	221	244	283	
	EBITDA difference for Earn-Out Opportunity calculation		14	30	39	62	101	
2	Earn-Out Opportunity calculation							
	For 100% share		88.4	194.1	250.7	403.5	659.0	
	For Camargo Group's 40.9% share		36.2	79.4	102.5	165.0	269.5	
	For Camargo Group's share after US\$235m cap		36.2	79.4	102.5	165.0	235.0	
	Portion of Earn-Out Opportunity used in calculations (%)		20%	40%	60%	80%	100%	
4	Earn-Out Payment							
	Earn-Out Opportunity used in calculation		7.2	31.8	61.5	132.0	235.0	
	Total Capital Gain Offset		8.0	1.7	2.6	3.5	4.6	
	Cumulative prior-year Total Capital Gain Offsets		_	0.8	2.5	5.0	8.6	
	Cumulative prior-year Earn-Out Payments		_	6.4	29.3	56.5	123.5	
	Earn-Out Payment		6.4	22.9	27.2	67.0	98.4	
	Value Received by Camargo Group							
	Value of Closing Date Shares	410.0						410.0
	Earn-Out Payments		6.4	22.9	27.2	67.0	98.4	221.8
	Total Capital Gain Offsets		8.0	1.7	2.6	3.5	4.6	13.2
	Total Notional Dividends		10.7	8.7	7.1	5.5	4.8	36.8
	Guaranteed Minimum Payment							_
	Commission Theal	410.0	427.0	401.1	400.0		C01.0	C01.0
	Cumulative Total	410.0	427.9	461.1	498.0	574.0	681.8	681.8

Note: All amounts in US\$ million unless otherwise stated.

Numbers on the left are references to paragraphs in Exhibit C.

Exam	ple 4 Hypothetical Scenario (continued):						
		Base	Year 1	Year 2	Year 3	Year 4	Year 5
3	Capital Gain Offset Calculation						
	Price of Alcoa Common Stock (US\$/share)						
	Price used to calculate Closing Date Shares	\$23.068					
	Average Closing Price		\$23.30	\$23.53	\$23.77	\$24.00	\$ 24.24
	Number of shares used for Capital Gain Offset calculations (million shares)						
3(a)	Closing Date Shares	17.77					
3(b)	Notional Earn-Out Shares		0.28	0.97	1.14	2.79	4.06
	Cumulative Total	17.77	18.05	19.02	20.17	22.96	27.01
	Imputed sale of shares (million shares)						
3(a)	Closing Date Shares		3.55	3.55	3.55	3.55	3.55
3(b)	Year 1 Notional Earn-Out Shares			0.06	0.06	0.06	0.06
3(b)	Year 2 Notional Earn-Out Shares				0.19	0.19	0.19
3(b)	Year 3 Notional Earn-Out Shares					0.23	0.23
3(b)	Year 4 Notional Earn-Out Shares						0.56
	Capital Gain Offset						
3(a)	Closing Date Shares		0.82	1.65	2.48	3.33	4.18
3(b)	Year 1 Notional Earn-Out Shares			0.01	0.03	0.04	0.05
3(b)	Year 2 Notional Earn-Out Shares				0.05	0.09	0.14
3(b)	Year 3 Notional Earn-Out Shares					0.05	0.11
3(b)	Year 4 Notional Earn-Out Shares						0.13
3(c)	Total Capital Gain Offset		0.82	1.66	2.56	3.51	4.62
	Notional Dividends Calculation						
	Alcoa Dividends Per Share (US\$/share)		\$ 0.60	\$ 0.60	\$ 0.60	\$ 0.60	\$ 0.60
	Imputed shareholding for Notional Dividend calculations (million shares)						
	Closing Date Shares		17.77	14.22	10.66	7.11	3.55
	Year 1 Notional Earn-Out Shares			0.28	0.22	0.17	0.11
	Year 2 Notional Earn-Out Shares				0.97	0.78	0.58
	Year 3 Notional Earn-Out Shares					1.14	0.92
	Year 4 Notional Earn-Out Shares						2.79
	Total		17.77	14.49	11.86	9.20	7.95
	Notional Dividends		10.7	8.7	7.1	5.5	4.8

Note: All amounts in US\$ million unless otherwise stated.

Numbers on the left are references to paragraphs in Exhibit C.

TERMINATION OF SHAREHOLDERS AGREEMENT

WHEREAS, Alcoa Brazil Holdings Company, Construções e Comércio Camargo Corrêa, Trelawney Inc. and Allpar Limited are parties to a Shareholders Agreement dated March 28, 1995, as amended, regarding Alcoa Alumínio S.A. (the "Shareholders Agreement");

WHEREAS, the parties desire to terminate the Shareholders Agreement effective August 1, 2003.

NOW, THEREFORE, intending to be legally bound hereby, the parties hereby agree as follows:

Termination. The Shareholders Agreement is hereby terminated without further liability, effective August 1, 2003.

IN WITNESS WHEREOF, the parties have executed this Termination of Shareholders Agreement as of August 1, 2003.

ALCOA BRAZIL HOLDINGS COMPANY	CONSTRUÇÕES E COMÉRCIO CAMARGO CORRÊA S.A.
By:	By:
Witnessed:	Witnessed:
By:	By:
By:	By:
TRELAWNEY INC.	ALLPAR LIMITED
By:	By:
Witnessed:	Witnessed:
By:	By:
By:	By:

TERMINATION OF ALAHC SHAREHOLDERS AGREEMENT

WHEREAS, Alcoa Inc., Construções e Comércio Camargo Corrêa, Trelawney Inc. and Alcoa Latin American Holdings Corporation are parties to an ALAHC Shareholders Agreement dated March 15, 1995, as amended, regarding Alcoa Latin American Holdings Corporation (the "ALAHC Shareholders Agreement");

WHEREAS, the parties desire to terminate the ALAHC Shareholders Agreement effective August 1, 2003.

NOW, THEREFORE, intending to be legally bound hereby, the parties hereby agree as follows:

Termination. The ALAHC Shareholders Agreement is hereby terminated without further liability, effective August 1, 2003.

IN WITNESS WHEREOF, the parties have executed this Termination of ALAHC Shareholders Agreement as of August 1, 2003.

ALCOA INC.	CONSTRUÇÕES E COMÉRCIO CAMARGO CORRÊA S.A.				
By:	By:				
TRELAWNEY INC.	ALCOA LATIN AMERICAN HOLDINGS CORPORATION				
By:	By:				

EXHIBIT E

REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT (this "Agreement") is entered into as of July 25, 2003 by and among ALCOA INC., a corporation organized and existing under the laws of the Commonwealth of Pennsylvania, United States of America ("Alcoa"), CAMARGO CORRÊA S.A., a corporation organized and existing under the laws of the Federative Republic of Brazil ("Camargo Corrêa"), ALLPAR LIMITED, a corporation organized and existing under the laws of the Cayman Islands ("Allpar") and TRELAWNEY INC., a corporation organized and existing under the laws of The Bahamas ("Trelawney"). The entities listed above are collectively referred to herein as the "parties" and are referred to individually as a "party". Allpar and Trelawney are referred to individually herein as an "Acquirer" and collectively herein as the "Acquirers".

RECITALS

The parties have entered into an agreement of even date herewith (the "Principal Agreement") pursuant to which, among other things, the parties agreed that the Camargo Group (as defined herein) will exchange all of their shares in the capital stock of Alcoa Alumínio S.A. ("Alcoa Alumínio"), a Brazilian corporation, and Alcoa Latin American Holdings Corporation ("ALAHC"), a British Virgin Islands company, with Alcoa, Alcoa Luxembourg S.à.r.l., a Luxembourg company, and/or their designated U.S. Person Affiliate(s) for shares of common stock of Alcoa and certain other consideration in accordance with the terms and conditions of the Principal Agreement.

The parties desire to set forth in this Agreement certain additional terms and conditions applicable to the holding and transfer of the shares of common stock of Alcoa to be delivered to Allpar and Trelawney in such exchange.

In consideration of the mutual promises, covenants and conditions hereinafter set forth, the parties mutually agree as follows:

ARTICLE I DEFINITIONS

As used herein, the terms below shall have the following meanings:

"Affiliate" means any person or entity which directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such person or entity. The term "control" (including the terms "controlled by" and "under common control with") as used with respect to any person or entity means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such person or entity, whether through the ownership of voting securities, by contract or otherwise.

- "Camargo Group" means the following members: Camargo Corrêa, Allpar and Trelawney.
- "Common Stock" means the common stock of Alcoa, par value \$1.00 per share.
- "Closing" means the satisfactory completion of the actions described in Article III of the Principal Agreement, including, without limitation, satisfaction or waiver of the conditions set forth in Sections 3.4 and 5.2.5 of the Principal Agreement.
 - "Closing Date" means the date on which the Closing occurs pursuant to the Principal Agreement.
- "Exchange" means the exchange of all of the shares of common stock of Alcoa Alumínio held by Allpar, Trelawney and the individuals set forth on Exhibit A to the Principal Agreement, as set forth on such Exhibit A and all of the shares of common stock of ALAHC held by Trelawney, as set forth on Exhibit A to the Principal Agreement for the Shares in accordance with the Principal Agreement.
 - "Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended.
- "Permitted Seller" means the Camargo Group and its Affiliates and, subject to Section 4.2(g)(4) of this Agreement, their Transferees to the extent of Shares Transferred pursuant to Section 2.4.3(d), 2.4.3(e), 2.4.3(f), 2.4.3(g) or 2.4.3(h) of the Principal Agreement.
 - "Person" means any individual, corporation, limited liability company, partnership or any other private, public or government entity.
 - "SEC" means the U.S. Securities and Exchange Commission.
 - "Securities Act" means the U.S. Securities Act of 1933, as amended.
- "Shares" means the 17,773,540 shares of Common Stock to be delivered in the Exchange in accordance with the Principal Agreement as such number may be adjusted pursuant to the Principal Agreement.
- "Shelf Registration Beneficiary" means each Permitted Seller, if any, that has delivered a Shelf Notice of Sale to Alcoa pursuant to Section 4.2(g)(2) of this Agreement.
 - "Shelf Registration Statement" has the meaning set forth in Section 4.1 of this Agreement.
 - "Trading Day" means a day on which the New York Stock Exchange and commercial banks in New York City are open for the transaction of business.
 - "Transfer" has the meaning set forth in Section 2.4.2 of the Principal Agreement.

"Transferee" has the meaning set forth in Section 2.4.5 of the Principal Agreement.

"U.S." means United States of America.

Unless otherwise specifically defined herein, each term used herein that is defined in the Principal Agreement has the meaning assigned to such term in the Principal Agreement.

ARTICLE II EXCHANGE AND OTHER MATTERS

- 2.1 **Exchange**. In accordance with Article II of the Principal Agreement, in the Exchange, Trelawney and Allpar are acquiring the Shares in a private placement under Section 4(2) of the Securities Act.
- 2.2 <u>Legends</u>. The certificates representing and/or the book-entry accounts evidencing ownership of the Shares by the Camargo Group shall be endorsed with a legend(s) substantially to the following effect:

"The Shares represented by [this certificate] [this book-entry account] have not been registered under the Securities Act of 1933 and may not be transferred in the absence of such registration or an exemption therefrom under such Act. In addition, the Shares are subject to certain rights of first refusal and restrictions on transfer as provided in the Principal Agreement, dated as of July 25, 2003 among Alcoa Inc. and certain other signatories thereto and the Registration Rights Agreement, dated as of July 25, 2003 among Alcoa Inc. and certain other signatories of which are on file at the principal office of Alcoa Inc."

2.3 Notice of Proposed Transfer; Opinions of Counsel. Until the second anniversary of the Closing Date, prior to making any Transfer of any Shares to any Person, other than a Transfer registered under an effective registration statement under the Securities Act, the holder of such Shares (a "Holder") (other than a member or an Affiliate of the Camargo Group, each of whom will be subject to Section 2.8 of the Principal Agreement in lieu of this Section 2.3) will give written notice to Alcoa of such Holder's intention to effect such Transfer. Each such notice shall describe the manner and circumstances of the proposed Transfer. If within 5 Trading Days after receipt by Alcoa of such notice, Alcoa requests an opinion of counsel for such Holder that the proposed Transfer may be effected without registration of the Transfer under the Securities Act (such counsel opinion to be reasonably satisfactory to legal counsel for Alcoa), then Alcoa shall not be required to cause its share transfer agent to record such Transfer, and the Holder shall not be entitled to effect such Transfer, unless and until Alcoa receives such counsel opinion reasonably satisfactory to Alcoa. Such Holder shall thereupon be entitled to Transfer such Shares in accordance with the terms of the notice delivered by such Holder to Alcoa. Each certificate or book-entry account representing such Shares issued upon or in connection with such Transfer shall bear the restrictive legends required by Section 2.2.

ARTICLE III REMOVAL OF LEGEND AND TRANSFER RESTRICTIONS

Any legend relating to the Securities Act or any U.S. state securities laws endorsed on a stock certificate and/or the book-entry accounts representing any Shares delivered pursuant to the Principal Agreement and this Agreement, and the stop transfer instructions with respect to such Shares shall be removed and Alcoa shall issue a certificate without any legend to the Holder of such Shares or create a book-entry account without any legend, as the case may be, if (i) the offer and sale of such Shares are registered under the Securities Act, a prospectus meeting the requirements of the Securities Act is available with respect to such securities, such Shares are sold in accordance with the plan of distribution contained in the prospectus, and the Holder agrees to deliver such prospectus to the extent required by law, (ii) such Shares are sold in accordance with Rule 144 under the Securities Act and applicable U.S. state securities laws and the Holder provides Alcoa with an opinion of the Holder's counsel, reasonably satisfactory to legal counsel for Alcoa, to the effect that such Shares have been sold in accordance with Rule 144, or (iii) all of such Shares may be sold under Rule 144(k) and the Holder will provide Alcoa with an opinion of counsel reasonably satisfactory to legal counsel to Alcoa to the effect that Rule 144(k) is so available and such legend is not required in order to ensure compliance with the provisions of the Securities Act. Any legend relating to the rights of first refusal and restrictions on transfer set forth in the Principal Agreement shall be removed and Alcoa shall issue a certificate without such legend to the Holder of such Shares or create a book-entry account without such legend, as the case may be, as and when required pursuant to the terms of the Principal Agreement.

ARTICLE IV REGISTRATION RIGHTS

4.1 Shelf Registration. As set forth in Section 2.3 of the Principal Agreement, Alcoa shall (a) no later than 30 days after the Closing Date, prepare and file with the SEC a registration statement for the resale of the Shares in an offering to be made on a delayed or continuous basis pursuant to Rule 415 under the Securities Act (or any similar rule that may be adopted by the SEC) (the "Shelf Registration Statement"), (b) use its reasonable best efforts to cause such Shelf Registration Statement to become effective as soon as practicable after such filing, but in all events within 120 days after the Closing Date, (c) use its reasonable best efforts to cause the Shelf Registration Statement to remain effective at all times thereafter until the second anniversary of the Closing Date or such shorter period as will terminate when all of the Shares have been disposed of by the Permitted Sellers or may be disposed of without regard to the volume limitations imposed under Rule 144 (but in any event not before the expiration of any longer period required under the Securities Act) (the "Effectiveness Period"), (d) prepare and file with the SEC such amendments and supplements to the Shelf Registration Statement and the prospectus used in connection therewith as may be necessary to keep such registration statement effective for the period specified in this sentence above, and (e) when requested by a Shelf Registration Beneficiary, keep such Shelf Registration Beneficiary reasonably apprised of the status of the Shelf Registration Statement prior to the expiration of the Effectiveness Period. The

registration of the Shares so effected by Alcoa pursuant to this Section is referred to herein as the "Shelf Registration".

- 4.2 Registration Procedures. With respect to the Shelf Registration under this Article IV, Alcoa will:
- (a) before filing the Shelf Registration Statement or prospectus or any amendment or supplement, furnish to and afford the Camargo Group and any Permitted Seller and the managing underwriters, if any have been identified, a reasonable opportunity to review copies of all such documents (including copies of any documents to be incorporated by reference therein and all exhibits thereto) proposed to be filed (in each case at least three business days prior to such filing or such later date as is reasonable under the circumstances), and Alcoa shall not file the Shelf Registration Statement or prospectus or any amendments or supplements thereto if the Permitted Seller, its counsel, or the managing underwriters, shall reasonably object on a timely basis to such filing by reason of such filing containing incorrect or misleading information pertaining to the Permitted Seller or the managing underwriters; and furnish to Camargo Corrêa (on behalf of each member or Affiliate of the Camargo Group that is a Holder of Shares) and each Shelf Registration Beneficiary such number of copies of the Shelf Registration Statement, each amendment and supplement thereto, the prospectus included in such Shelf Registration Statement (including each preliminary prospectus, all exhibits thereto and documents incorporated by reference therein) and such other documents as such Shelf Registration Beneficiary may reasonably request during the Effectiveness Period in order to facilitate the disposition of the Shares owned by the Shelf Registration Beneficiary;
- (b) use its best efforts during the Effectiveness Period to register or qualify all Shares included in the Shelf Registration under such other securities or blue sky laws of such jurisdictions in the U.S. as any selling Shelf Registration Beneficiary reasonably requests and do any and all other acts and things which may be reasonably necessary or advisable to enable such Shelf Registration Beneficiary to consummate the disposition in such jurisdictions of such Shares to be sold by such Shelf Registration Beneficiary (provided that Alcoa will not be required to (i) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this subparagraph, (ii) subject itself to taxation in any such jurisdiction or (iii) consent to general service of process in any such jurisdiction);
- (c) notify each Shelf Registration Beneficiary selling Shares in such offering, at any time during the Effectiveness Period when a prospectus relating thereto is required to be delivered under the Securities Act, (i) upon discovery that, or upon the discovery of the happening of any event as a result of which, the prospectus included in the Shelf Registration contains an untrue statement of a material fact or omits any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made, and subject to Section 4.2(g) below, at the request of such Shelf Registration Beneficiary, Alcoa will as promptly as practically possible prepare and file with the SEC and, at the request of any such Shelf Registration Beneficiary, furnish to such Shelf Registration Beneficiary a reasonable number of copies of, a supplement or amendment to such prospectus (if such a supplement or amendment is required under the Securities Act) so that, as thereafter delivered to the purchasers of such Shares, such prospectus will not contain an untrue statement of a material fact or omit to

state any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made, (ii) when the prospectus or any supplement or post-effective amendment has been filed, and, with respect to the Shelf Registration Statement or any post-effective amendment, when the same has become effective under the Securities Act (including in such notice a written statement that any Shelf Registration Beneficiary may, upon request, obtain, at the sole expense of Alcoa, one conformed copy of the Shelf Registration Statement or post-effective amendment including financial statements and schedules, documents incorporated or deemed to be incorporated by reference and exhibits), (iii) if at any time when a prospectus is required by the Securities Act to be delivered in connection with sales of the Shares the representations and warranties of Alcoa contained in any agreement entered into pursuant to (f) below (including any underwriting agreement) cease to be true and correct in all material respects, (iv) of the receipt by Alcoa of any notification with respect to the suspension of the qualification or exemption from qualification of the Shelf Registration Statement or the Shares for offer or sale in any jurisdiction, or the initiation or threatening of any proceeding for such purpose, and (v) of Alcoa's determination that a post-effective amendment to the Shelf Registration Statement would be appropriate.

- (d) otherwise use its best efforts to comply with all applicable rules and regulations of the SEC, and make available to its security holders, as soon as reasonably practicable, an earnings statement (which need not be audited) covering the period of at least twelve months beginning with the first day of the first full calendar quarter of Alcoa after the effective date of the Shelf Registration Statement, which earnings statement shall satisfy the provisions of section 11(a) of the Securities Act and Rule 158 thereunder;
- (e) in the event of the issuance of any stop order suspending the effectiveness of the Shelf Registration Statement, or of any order suspending or preventing the use of any related prospectus or suspending the qualification of any securities included in the Shelf Registration Statement for sale in any jurisdiction, Alcoa will (i) promptly notify each Shelf Registration Beneficiary of such stop order, and (ii) use its best efforts promptly to obtain the withdrawal of such order;
- (f) in the case of an underwritten offering during the Effectiveness Period, enter into such customary agreements (including underwriting agreements in customary form) and take all such other actions as the Shelf Registration Beneficiary or the underwriters, if any, reasonably request in order to expedite or facilitate the disposition of such Shares (but not including any "road-show" participation by management of Alcoa) and cause to be delivered to the underwriters and the Shelf Registration Beneficiary, if any, at the expense of Alcoa, opinions of counsel to Alcoa and "comfort letters" from the auditors to Alcoa in customary form, covering such matters as are customarily covered by opinions and "comfort letters" for an underwritten public offering as the underwriters may reasonably request and addressed to the underwriters and the Shelf Registration Beneficiary;
 - (g) Notwithstanding anything to the contrary contained in this Agreement,
 - (1) Right to Impose Delay Periods. Alcoa shall have the right to delay the effectiveness of the Shelf Registration Statement or to suspend the right of a

Permitted Seller to sell any Shares under the effective Shelf Registration Statement for a reasonable length of time not to exceed 45 calendar days (a "Delay Period") if:

- (A) (i) Alcoa would, in accordance with the reasonable advice of its legal counsel, be required to disclose in the prospectus non-public information not otherwise required by law to be publicly disclosed and (ii) in the reasonable judgment of the Chief Executive Officer, Chief Financial Officer or General Counsel of Alcoa, (x) there is a reasonable likelihood that such disclosure, or any other action to be taken in connection with the sale of any Shares under the Shelf Registration Statement, would materially and adversely affect or interfere with a financing, acquisition, merger, joint venture, disposition of assets (not in the ordinary course of business), corporate reorganization, recapitalization or other similar transaction involving Alcoa, or (y) there is a bona fide material commercial transaction or event and in the reasonable judgment of the Chief Executive Officer, Chief Financial Officer or General Counsel of Alcoa, there is a bona fide business purpose that relates to such transaction or event that is served by a delay in disclosing information related to such transaction or event; or
- (B) an amendment to the Shelf Registration is required so that, as of such date, the Shelf Registration and prospectus do not contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading;

<u>provided</u>, that the aggregate number of days in all Delay Periods occurring in any period of twelve consecutive months shall not exceed 90 and a minimum of 5 Trading Days shall elapse between successive Delay Periods.

(2) Delivery of Shelf Notices of Sale. Prior to the Camargo Group or its Affiliates or any other Permitted Seller offering to sell or selling any Shares under the effective Shelf Registration Statement, Camargo Corrêa (with respect to any member or Affiliate of the Camargo Group) or such other Permitted Seller (with respect to itself), as applicable, shall give written notice to Alcoa of its intention to sell

Shares under the effective Shelf Registration Statement (a "Shelf Notice of Sale") specifying the following information to the extent not contained in the effective Shelf Registration Statement and related prospectus, as then amended or supplemented: (i) the name of each Holder intending to sell Shares, (ii) the nature of any position, office or other material relationship which such Holder has had within the past three years with Alcoa or any of its predecessors or Affiliates, (iii) the number of shares of Common Stock owned by such Holder prior to the offering, (iv) the number of Shares to be offered for sale for such Holder's account, (v) the amount and (if one percent or more) the percentage of the class to be owned by such Holder after completion of the offering, (vi) the plan of distribution of the Shares that are to be offered for sale under the Shelf Registration Statement, and (vii) the proposed date(s) of first and last use of the prospectus for offer or sale of the Shares. The Shelf Notice of Sale shall be delivered to Alcoa not later than: (i) in the case of a proposed sale of Shares under the effective Shelf Registration Statement by the Camargo Group or any Affiliate that is named in the Shelf Registration Statement and related prospectus, as then amended or supplemented, concurrently with the corresponding Proposed Sale Notice delivered to Alcoa under Section 2.4.2 of the Principal Agreement, or (ii) in the case of a proposed sale of Shares under the effective Shelf Registration Statement by any other Affiliate of the Camargo Group or other Permitted Seller, not later than the fifth Trading Day before the proposed date of first use of the prospectus for such offer or sale of the Shares. The Shelf Notice of Sale shall be effective in accordance with the provisions of Section 2.4.2 of the Principal Agreement relating to effectiveness of a Proposed Sale Notice. If requested in writing by Alcoa, Camargo Corrêa (with respect to any member or Affiliate of the Camargo Group) or such other Permitted Sell

- (3) Delivery and Effect of Delay Period Notices. Alcoa shall, not later than 9:00 a.m., New York City Time, on the Trading Day next succeeding the effective date of the Shelf Notice of Sale (or, if earlier, concurrently with the Alcoa Exercise Notice delivered under Section 2.4.2 of the Principal Agreement, in the case of a Shelf Notice of Sale given by Camargo Corrêa) (the "Delay Period Notice Deadline"), give Camargo Corrêa or the Permitted Seller, as applicable, written notice if a Delay Period is in effect for the sale indicated in the Shelf Notice of Sale and the expected duration thereof (a "Delay Period Notice"). If a Delay Period is in effect, until the end of the Delay Period, the Camargo Group and its Affiliates and the Permitted Sellers shall cease all efforts to offer or sell any of the Shares under the effective Shelf Registration Statement and shall not disclose to any third party the fact that a Delay Period is in effect.
- (4) Use Of Shelf Registration Statement by Transferees; Persons Eligible to be Permitted Sellers. Other than assignments to Affiliates of the Camargo Group, the Camargo Group and its Affiliates and other Persons who are subject to the restrictions and obligations of this Agreement agree that the rights and obligations

under this Agreement may only be assigned to a Transferee acquiring Shares in a Transfer if such Transfer cannot reasonably be accomplished under the Shelf Registration Statement or under Rule 144 of the Securities Act by the proposed closing date of the proposed Transfer. Furthermore, Camargo Group and its Affiliates and such other Persons agree that the number of Transferees who may be assignees of this Agreement and thus eligible to be Permitted Sellers (as defined herein) is subject to the following limitations:

- (A) no more than two (2) Transferees pursuant to each of Sections 2.4.3(d), 2.4.3(e), 2.4.3(f), 2.4.3(g) and 2.4.3(h) of the Principal Agreement; and
- (B) no more than eight (8) Transferees (other than the Camargo Group and its Affiliates) in the aggregate at any time. Furthermore, each such Transferee (other than the Camargo Group and its Affiliates) shall be eligible to reassign in whole its rights and obligations under this Agreement only once to one subsequent Transferee of the Shares (who shall have no further assignment rights) if such subsequent Transfer cannot reasonably be accomplished under the Shelf Registration Statement or under Rule 144 of the Securities Act by the proposed closing date of the proposed Transfer and provided that under all circumstances the total number of Transferees (other than the Camargo Group and its Affiliates) eligible at any given time to be Permitted Sellers under this Agreement is subject to the limitation set forth in this clause (B). The Camargo Group and its Affiliates shall: (i) give prior written notice to Alcoa in accordance with Section 7.6 of the Principal Agreement of any assignment by it of the rights and obligations of this Agreement hereunder (which notice may be included as part of the Proposed Sale Notice); (ii) ensure that the assignee agrees to all restrictions and obligations imposed under this Agreement; (iii) ensure that the assignee agrees to the provisions of this Section 4.2(g)(4), including those relating to any reassignment by it, and agrees to give prior written notice to Alcoa in accordance with the provisions of Section 5.4 of any such reassignment. Any such assignment shall not be effective unless the proposed Transferee executes and delivers a written instrument reasonably satisfactory to Alcoa whereby such proposed Transferee agrees to be bound by the restrictions and obligations under this Agreement and the limitations on assignment of this Agreement contained in this Section 4.2(g)(4).

(h) cooperate with the Shelf Registration Beneficiaries and the managing underwriter, if any, to facilitate the timely preparation and delivery of certificates representing the Shares to be sold, which certificates shall not bear any restrictive legends and shall be in a form eligible for

deposit with The Depository Trust Company and enable such Shares to be in such denominations and registered in such names as the managing underwriter, if any, or Shelf Registration Beneficiaries may request at least three Trading Days prior to any sale of such Shares;

- (i) make available for inspection by any Shelf Registration Beneficiary selling Shares, any underwriter participating in any such disposition of Shares, if any, and any attorney, accountant or other agent retained by any such Shelf Registration Beneficiary, or underwriter (collectively, the "Inspectors"), at the offices where normally kept, during reasonable business hours, all financial and other records, pertinent corporate documents and instruments of Alcoa and its subsidiaries (collectively, the "Records") as shall be reasonably necessary to enable them to exercise any applicable due diligence responsibilities, and cause the officers, directors and employees of Alcoa and its subsidiaries to supply all information reasonably requested by any such Inspector in connection with the Shelf Registration Statement and prospectus, and each Inspector shall agree in writing pursuant to confidentiality agreements on customary terms to keep the Records confidential; and
- (j) cooperate with each Shelf Registration Beneficiary selling Shares and each underwriter, if any, participating in the disposition of such Shares and their respective counsel in connection with any filings required to be made with the National Association of Securities Dealers, Inc. (the "NASD").
- 4.3 <u>Shelf Registration Beneficiary Information</u>. Each Shelf Registration Beneficiary shall furnish to Alcoa such information regarding such Shelf Registration Beneficiary and the distribution of Shares proposed by such Shelf Registration Beneficiary as Alcoa may reasonably request in writing and as shall be reasonably required in connection with any registration, qualification, or compliance referred to in this Article IV.

4.4 Registration Expenses.

- (a) Except as expressly provided in this Article IV, Alcoa shall pay all Registration Expenses relating to the Shelf Registration. "Registration Expenses" shall mean any and all fees and expenses incident to Alcoa's performance of or compliance with this Article IV, including: (i) SEC, stock exchange or NASD registration and filling fees, (ii) fees and expenses of compliance with U.S. state securities or "blue sky" laws and in connection with the preparation of a "blue sky" survey, including reasonable fees and expenses of blue sky counsel, (iii) printing expenses, (iv) messenger and delivery expenses related to distributing prospectuses in preliminary and final form as well as any supplements thereto, (v) fees and disbursements of counsel for Alcoa, (vi) fees and expenses of Alcoa's auditors in connection with the preparation and delivery of any "comfort letters" reasonably required by the underwriter(s), and (vii) fees and disbursements of all independent public accountants and fees and expenses of other Persons, including special experts, retained by Alcoa.
- (b) Notwithstanding the foregoing, (i) the provisions of this Article IV shall be deemed amended to the extent necessary to cause these expense provisions to comply with "blue sky" laws of each state within the U.S. in which the offering is made and (ii) in connection with the Shelf Registration, each Shelf Registration Beneficiary shall pay all underwriting discounts

and commissions and transfer taxes, if any, attributable to the Shares included in the offering by the Shelf Registration Beneficiary and all fees and expenses of such Shelf Registration Beneficiary's counsel.

4.5 Indemnification.

- (a) Alcoa agrees to indemnify and hold harmless, to the extent permitted by law, each Shelf Registration Beneficiary and underwriter of any underwritten transaction, their respective officers, directors and Affiliates against any losses, claims, damages or liabilities to which such Shelf Registration Beneficiary and underwriter or any such director, officer, or Affiliate may become subject under the Securities Act or otherwise, insofar as such losses, claims damages or liabilities (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon (i) any untrue or alleged untrue statement of a material fact contained (A) in the Shelf Registration Statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto used in connection with this Article IV or (B) in any application or other document or communication (in this Section 4.5 collectively called an "application") executed by or on behalf of Alcoa or based upon written information furnished by or on behalf of Alcoa filed in any jurisdiction in order to qualify any securities covered by the Shelf Registration Statement under the "blue sky" or securities laws thereof, or (ii) any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading contained (A) in the Shelf Registration Statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto used in this Article IV or (B) in any application executed by or on behalf of Alcoa or based upon written information furnished by or on behalf of Alcoa filed in any jurisdiction in order to qualify any securities covered by the Shelf Registration Statement under the "blue sky" or securities laws thereof, and Alcoa will reimburse the Shelf Registration Beneficiary and underwriter and each such director, officer and Affiliate for any legal or any other expenses incurred by them, as such expenses are incurred, in connection with investigating or defending any such loss, claim, liability, action or proceeding; provided that Alcoa shall not be liable in any such case to the extent that any such loss, claim, damage, liability (or action or proceeding in respect thereof) or expense arises out of or is based upon an untrue statement or omission made in any such prospectus or preliminary prospectus or any amendment or supplement thereto, or in any application, in reliance upon and in conformity with written information prepared and furnished to Alcoa by any Shelf Registration Beneficiary or underwriter expressly for use therein or by any Shelf Registration Beneficiary's or underwriter 's failure to deliver a copy of the prospectus or any amendments or supplements thereto after Alcoa has furnished any Shelf Registration Beneficiary or underwriter with a sufficient number of copies of the same.
- (b) In connection with the Shelf Registration, such Shelf Registration Beneficiary and underwriter will furnish to Alcoa in writing such information and documents concerning such Shelf Registration Beneficiary or underwriter as Alcoa reasonably requests for use in connection with the Shelf Registration Statement or any prospectus and, to the extent permitted by law, will indemnify and hold harmless Alcoa and its directors, officers and Affiliates against any losses, claims, damages or liabilities to which Alcoa or any such director, officer or Affiliate may become subject under the Securities Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions or proceedings, whether commenced or threatened, in respect thereof) arise

out of or are based upon (i) any untrue or alleged untrue statement of a material fact contained in the Shelf Registration Statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto or in any application or (ii) any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue or alleged untrue statement or omission or alleged omission is made in the Shelf Registration Statement, any such prospectus or preliminary prospectus or any amendment thereof or supplement thereto, or in any application, in reliance upon and in conformity with written information concerning such Shelf Registration Beneficiary or underwriter prepared and furnished to Alcoa by or on behalf of such Shelf Registration Beneficiary or underwriter in writing expressly for use therein, and each Shelf Registration Beneficiary will reimburse Alcoa and each such director, officer and Affiliate for any legal or any other expenses incurred by them, as such expenses are incurred, in connection with investigating or defending any such loss, claim, liability, action or proceeding; provided that the obligation to indemnify will be individual to each Shelf Registration Beneficiary and underwriter and will be limited to the net amount of proceeds received by such Shelf Registration Beneficiary and underwriter from the sale of Shares by such Shelf Registration Beneficiary or underwriter pursuant to the Shelf Registration Statement.

- (c) Any Person entitled to indemnification hereunder will (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified parties and the indemnifying party may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party; provided, however, that the failure of any indemnified party to give notice as provided herein shall not relieve the indemnifying party of its obligations hereunder unless the failure to give such notice is materially prejudicial to an indemnifying party's ability to defend such action. If such defense is assumed, the indemnified party will not be subject to any liability for any settlement made by the indemnifying party without its consent (but such consent will not be unreasonably withheld). Anything to the contrary appearing in this Agreement notwithstanding, an indemnifying party will not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified hereunder with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. If an indemnifying party assumes the defense, the indemnified party may engage its own counsel at its own sole cost and expense. All fees and expenses of counsel to any indemnified party required to be paid by the indemnifying party shall be paid by the indemnifying party as incurred by such indemnified party.
- (d) The indemnification provided for under this Agreement will remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director, Affiliate or controlling or controlled Person of such indemnified party and will survive the transfer of Shares by any Shelf Registration Beneficiary or underwriter. If the indemnification provided for herein is unavailable to an indemnified party or insufficient in respect of any losses, claims, damages or liabilities referred to herein, then the indemnifying party, in lieu of indemnifying such indemnified party hereunder, shall contribute to the amount paid or payable by such indemnified party as a result of such losses, claims, damages or

liabilities in such proportion as is appropriate to reflect the relative fault of the indemnifying party, on the one hand, and the indemnified party or parties, on the other hand, in connection with the statements or omissions that resulted in such losses, claims, damages or liabilities, as well as any other relevant equitable considerations; the relative fault of the indemnifying party on the one hand and of the indemnified party on the other shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party, and by such party's relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission; provided, however, that in no event shall any contribution by any Shelf Registration Beneficiary or underwriter or any director, officer, Affiliate or controlling or controlled Person thereof exceed the amount of the net proceeds received by such Shelf Registration Beneficiary or underwriter from the sale of Shares pursuant to such registration statement.

4.6 <u>Current Public Information</u>. At all times prior to the earlier of the second anniversary of the Closing Date and the date on which each Permitted Seller has disposed of all of such Permitted Seller's Shares, Alcoa will file all reports required to be filed by it under the Securities Act and the Exchange Act and the rules and regulations adopted by the SEC thereunder to the extent required to enable the Permitted Seller to sell Shares pursuant to Rule 144.

ARTICLE V MISCELLANEOUS

- 5.1 **Third Parties.** This Agreement is made for the purpose of defining and setting forth certain obligations, rights and duties of the parties and their respective successors and permitted assigns. No Person other than the parties or the Permitted Sellers and any underwriters shall have any rights or obligations of any nature hereunder or by reason hereof.
- 5.2 <u>Publicity</u>. Unless required by law (including a requirement that, in the reasonable judgment of Alcoa's counsel, disclosure should be included in any filing with, or report required by, the SEC), no party and no Permitted Seller shall issue any press release or make any public statement regarding the transactions contemplated hereby, without the prior approval of the other parties hereto.
- 5.3 Expenses. Subject to Article IV hereof, each party and each Permitted Seller will bear its own costs and expenses incurred in connection with the negotiation, preparation, execution and performance of this Agreement and the transactions contemplated herein.
- 5.4 <u>Notices</u>. All notices required to be given hereunder shall be in writing and shall be served personally, sent by tested facsimile (with transmission confirmation) or delivered by an international courier to the addresses set forth in Section 7.6 of the Principal Agreement (and set forth at the end of this Section 5.4). Each such notice shall be effective (a) if delivered by international courier, when delivered to the recipient's address provided above, or, if such recipient shall have given effective notice hereunder of a new address, when delivered at such

new address, (b) if sent by facsimile, when received by the recipient at his address provided above, or, if such recipient shall have given effective notice hereunder of a new address, when received at such new address, or (c) if delivered in person by hand, upon the delivery to the intended recipient.

To the Alcoa Group: Alcoa Inc.

390 Park Avenue

New York, New York, U.S.A. 10022

Attention: Group President - Latin America

Facsimile: (212) 836-2816

Attention: Treasurer Facsimile: (212) 836-2823 Attention: General Counsel Facsimile: (212) 836-2844

To the Camargo Group: Camargo Corrêa S.A.

Rua Funchal, 160 Sao Paulo, SP Brasil

Attention: President

Facsimile: 55-11-3849-7837

Attention: Treasurer

Facsimile: 55-11-3841-5658

To Permitted Sellers: Address delivered in writing to Alcoa and the

04551-903

Camargo Group

5.5 <u>Assurances and Cooperation</u>. On and after the Closing Date, the parties and the Permitted Sellers will take all appropriate action and execute all documents, instruments or conveyances that may reasonably be necessary or advisable to carry out any of the provisions hereof. Each party and the Permitted Sellers shall coordinate and cooperate in supplying such reasonable assistance as may be reasonably requested by any other party or Permitted Seller in connection with the actions contemplated by this Agreement.

5.6 <u>Confidentiality</u>. Each Permitted Seller (other than the Camargo Group and its Affiliates, who will be bound by the provisions of Section 7.9 of the Principal Agreement in lieu of this Section 5.6) will hold, and will ensure that their respective officers, directors, employees, accountants, counsel, consultants, advisors and agents hold, in confidence and not disclose to any third party, unless compelled to disclose by judicial or administrative process or by other requirements of law, all financial, technical or other confidential information furnished to the

Permitted Seller by Alcoa or its Affiliates in connection with the transactions contemplated by this Agreement, except to the extent that such information can be shown to have been (i) previously known on a nonconfidential basis by the Permitted Seller, (ii) in the public domain through no fault of the Permitted Seller or (iii) later lawfully acquired by the Permitted Seller from sources other than Alcoa or its Affiliates who had no duty to hold it as confidential. The Permitted Seller may disclose such information to its officers, directors, employees, accountants, counsel, consultants, advisors and agents who have a need to know such information in connection with the transactions contemplated by this Agreement so long as such persons are informed by the Permitted Seller of the confidential nature of such information and are obligated by the Permitted Seller to treat such information confidentially. Notwithstanding the foregoing, the Permitted Seller may disclose the tax treatment and tax structure of the transactions contemplated by this Agreement (including any materials, opinions or analyses relating to such tax treatment or tax structure, but without disclosure of identifying information or, except to the extent relating to such tax structure or tax treatment, any nonpublic commercial or financial information). Moreover, notwithstanding any other provision of this Agreement, there shall be no limitation on the Permitted Seller's ability to consult any tax adviser, whether or not independent from the Permitted Seller, regarding the tax treatment or tax structure of the transactions contemplated by this Agreement. The Permitted Seller shall be responsible for any failure to treat such information confidentially by such persons and their respective officers, directors, employees, accountants, counsel, consultants, advisors and agents.

- 5.7 <u>Assignment</u>. Except as expressly set forth in or contemplated by this Agreement, neither this Agreement nor any of the rights or obligations hereunder may be assigned by Alcoa or the Camargo Group without the prior written consent of the other, which shall not be unreasonably withheld (it being understood, however, that Alcoa may withhold its consent to any such assignment not expressly permitted under Section 4.2(g)(4) of this Agreement). Subject to the foregoing, this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective legal representatives, successors, heirs and permitted assigns (including the Permitted Sellers), and no other Person has any right, benefit or obligation hereunder.
- 5.8 **Waiver**. The waiver by any party or any Permitted Seller of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any subsequent breach.
- 5.9 <u>Headings</u>. The headings of the Articles and Sections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.
- 5.10 **Counterparts**. This Agreement may be executed in two or more counterparts in English, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 5.11 <u>Severability</u>. If any term, covenant, condition or provision of this Agreement or the application thereof to any circumstance shall be invalid or unenforceable to any extent, the remaining terms, covenants, conditions and provisions of this Agreement shall not be affected

thereby and each remaining term, covenant, condition or provision of this Agreement shall be enforceable to the fullest extent permitted by law.

- 5.12 **No Drafting Presumption**. The parties acknowledge that their respective legal counsel have reviewed and participated in settling the terms of this Agreement, and the parties agree that any rule of construction to the effect that any ambiguity is to be resolved against the drafting party shall not be applicable in the interpretation of this Agreement.
- 5.13 <u>Choice of Law</u>. This Agreement shall be construed, interpreted and the rights of the parties interpreted in accordance with the laws of the State of New York, excluding those related to choice or conflicts of law.
- 5.14 **Dispute Resolution**. In an effort to resolve informally and amicably any dispute or controversy which might arise between the parties or between Alcoa and a Permitted Seller (a "party" for purposes of the remainder of this provision) with respect to this Agreement, without first resorting to any third party or tribunal for resolution, the parties shall first endeavor to resolve the matter through direct discussions and negotiations. If the matter cannot be settled, it shall be referred by each party to a responsible officer of each of Alcoa and Camargo Corrêa. If the two officers are unable to resolve the matter within 30 days of its referral, the matter will be referred to the Chief Executive Officers of Alcoa and Camargo Corrêa. If settlement is not thereafter reached through their efforts within an additional 30 days, or such longer time period as they may agree, then either party may initiate arbitration proceedings to resolve the matter. Any such dispute, controversy or claim in connection with this Agreement, including its validity, interpretation, application, scope, enforceability, performance, breach and termination, shall be resolved exclusively and finally by arbitration, to the exclusion of the courts. Notice of arbitration shall be deemed proper if made in accordance with Section 5.4 of this Agreement. If the parties fail to agree in writing on the place where the arbitration is to be conducted, such arbitration shall be held in New York, New York, United States of America. Arbitration shall be conducted in English, pursuant to International Chamber of Commerce ("ICC") Arbitration Rules in force at the time of the arbitration, by a panel of three arbitrators who are fluent in the English language and who are skilled in the legal and business aspects of the subject matter of this Agreement. The arbitrators shall be appointed in accordance with ICC Rules. Any monetary award made pursuant to such arbitration shall be calculated and paid exclusively in Dollars. Judgment upon the award rendered may be entered in any court having jurisdiction, or an application may be made to any court for a judicial acceptance of the award and an order of enforcement, as the case may be. Each of the parties hereby irrevocably and unconditionally submits, for itself and its property, to the nonexclusive jurisdiction of any Federal court of the United States of America sitting in New York City and any appellate court from any thereof for recognition or enforcement of any such judgment. Each party hereby irrevocably and unconditionally waives any objection (including any defense of inconvenient forum) which it may now or hereafter have to the laying of such venue, and each party irrevocably consents to service of process in the manner provided for notices in Section 5.4.
- 5.15 Entire Agreement and Modifications. This Agreement and the Principal Agreement, together with the exhibits thereto, constitute the entire agreement between the parties with respect to the subject matter hereof and supersede all prior agreements,

understandings, negotiations and discussions, whether oral or written, of the parties. No supplement, modification or waiver of this Agreement shall be binding unless executed in writing by the party to be bound thereby.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement, or have caused this Agreement to be duly executed on their respective behalf by their duly authorized representative, as of the day and year first above written.

CAMARGO CORRÊA S.A.	ALCOA INC.
Ву	Ву
ALLPAR LIMITED	TRELAWNEY INC.
By	By

August 13, 2003

Alcoa Inc. 201 Isabella Street Pittsburgh, Pennsylvania 15212-5858

Ladies and Gentlemen:

This opinion is furnished in connection with the registration by Alcoa Inc., a Pennsylvania corporation (the "Company"), pursuant to a Registration Statement on Form S-3 (the "Registration Statement") under the Securities Act of 1933, as amended (the "Securities Act"), of an aggregate of 17,773,540 shares of common stock, \$1.00 par value, of the Company (the "Shares"), for resale by the selling shareholders named in the Registration Statement.

As Assistant General Counsel of the Company, I am of the opinion, based upon my familiarity with the affairs of the Company and upon my examination of the law and pertinent documents, that the Shares have been duly authorized and are legally issued, fully paid and non-assessable.

I am a member of the bar of the Commonwealth of Pennsylvania and my opinion is limited to the laws of the Commonwealth of Pennsylvania and the laws of the United States of America.

I hereby consent to the filing of this opinion as an exhibit to the Registration Statement and to the reference to me under the heading "Legal Matters" in the prospectus. In giving my consent, I do not admit that I am within the category of persons whose consent is required under Section 7 of the Securities Act or the rules and regulations of the Securities and Exchange Commission issued thereunder.

Very truly yours,

/s/ DENIS A. DEMBLOWSKI Denis A. Demblowski Assistant General Counsel August 13, 2003

Securities and Exchange Commission 450 Fifth Street, N.W. Washington, D.C. 20549

Commissioners:

We are aware that our report dated July 8, 2003, except for Note Q, as to which the date is July 18, 2003, on our review of interim financial information of Alcoa Inc. and subsidiaries as of and for the three and six month periods ended June 30, 2003 and included in the Company's quarterly report on Form 10-Q for the quarter then ended is incorporated by reference in its Registration Statement dated August 13, 2003.

Very truly yours,

/S/ PRICEWATERHOUSECOOPERS LLP

PricewaterhouseCoopers LLP

CONSENT OF INDEPENDENT ACCOUNTANTS

We hereby consent to the incorporation by reference in this Registration Statement on Form S-3 of Alcoa Inc. of our report dated January 8, 2003 relating to the financial statements, which appears in the 2002 Annual Report to Shareholders of Alcoa Inc., which is incorporated by reference in Alcoa Inc.'s Annual Report on Form 10-K for the year ended December 31, 2002. We also consent to the incorporation by reference of our report dated January 8, 2003 relating to the financial statement schedules, which appears in such Annual Report on Form 10-K. We also consent to the reference to us as "experts" under the heading "Independent Accountants" in such Registration Statement.

/S/ PRICEWATERHOUSECOOPERS LLP

Pittsburgh, Pennsylvania August 13, 2003

POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENTS, that each of the undersigned Directors of Alcoa Inc. (the "Company") hereby constitutes and appoints RICHARD B. KELSON, WILLIAM B. PLUMMER, CHARLES D. MCLANE, JR. and DONNA C. DABNEY, or any of them, his or her true and lawful attorneys-in-fact and agents to do any and all acts and things and to execute any and all instruments which said attorneys-in-fact and agents, or any of them, may deem necessary or advisable or may be required to enable the Company to comply with the Securities Act of 1933, as amended (the "Securities Act"), and any rules, regulations and requirements of the Securities and Exchange Commission (the "Commission") in respect thereof, in connection with the registration under the Securities Act of shares of common stock, par value \$1.00 per share, of the Company (the "Shares") and/or their resale by ALLPAR Inc. and/or Trelawney Inc. and/or other affiliates of Construções e Comércio Camargo Corrêa S.A., as selling shareholders, including specifically, but without limiting the generality of the foregoing, power and authority to sign the name of each of the undersigned in the capacity of Director of the Company to any registration statement to be filed with the Commission in respect of the Shares and/or the resale thereof, to any and all pre-effective amendments, post-effective amendments and supplements to any such registration statement, and to any instruments or documents filed as part of or in connection with any such registration statement or pre-effective amendments, post-effective amendments or supplements thereto, and to file or cause to be filed the same with the Commission; and to effect any and all applications and other instruments in the name and on behalf of each of the undersigned which said attorneys-in-fact and agents, or any of them, shall do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, each of the undersigned has subscribed these presents this 13th day of June, 2003.

/3/ KATIKIN S. I OLLEK
Kathryn S. Fuller
/s/ Carlos Ghosn
Carlos Ghosn
/s/ Joseph T. Gorman
Joseph T. Gorman
/s/ Judith M. Gueron
Judith M. Gueron
/S/ SIR RONALD HAMPEL
Sir Ronald Hampel
/S/ JOHN P. MULRONEY
John P. Mulroney
/S/ HENRY B. SCHACHT
Henry B. Schacht
/S/ FRANKLIN A. THOMAS
Franklin A. Thomas
/s/ Ernesto Zedillo

/S/ KATHRYN S. FULLER

Ernesto Zedillo

POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENTS, that the undersigned Executive Vice President and Chief Financial Officer of Alcoa Inc. (the "Company") hereby constitutes and appoints WILLIAM B. PLUMMER, CHARLES D. MCLANE, JR. AND DONNA C. DABNEY, or any of them, his true and lawful attorneys-in-fact and agents to do any and all acts and things and to execute any and all instruments which said attorneys-in-fact and agents, or any of them, may deem necessary or advisable or may be required to enable the Company to comply with the Securities Act of 1933, as amended (the "Securities Act"), and any rules, regulations and requirements of the Securities and Exchange Commission (the "Commission") in respect thereof, in connection with the registration under the Securities Act of shares of common stock, par value \$1.00 per share, of the Company (the "Shares") for resale by ALLPAR Inc. and/or Trelawney Inc. and/or other affiliates of Construções e Comércio Camargo Corrêa S.A., as selling shareholders, including specifically, but without limiting the generality of the foregoing, power and authority to sign the name of the undersigned in the capacity of Principal Financial Officer of the Company to any registration statement to be filed with the Commission in respect of the Shares and/or the resale thereof, to any and all pre-effective amendments, post-effective amendments and supplements to any such registration statement, and to any instruments or documents filed as part of or in connection with any such registration statement or pre-effective amendments, post-effective amendments or supplements thereto, and to file or cause to be filed the same with the Commission; and to effect any and all applications and other instruments in the name and on behalf of the undersigned which said attorneys-in-fact and agents, or any of them, deem advisable in order to qualify or register the Shares and/or the resale thereof under the securities laws of any of the several States; and the undersigned hereby ratifies and confirms all that said attorn

IN WITNESS WHEREOF, the undersigned has subscribed these presents this 13^{th} day of June, 2003.

/S/ RICHARD B. KELSON

Richard B. Kelson Executive Vice President and Chief Financial Officer